

National Telecommunications and

Information Administration

Broadband Technology Opportunities Program

Evaluation Study

Order Number D10PD18645

Study Design

Deliverables A-B

Revision Date: January 30, 2012

Originally Submitted 03/25/2011 by:

ASR Analytics, LLC 1389 Canterbury Way Potomac, MD 20854 Federal TIN: 20-1204680 DUNS: 15-108-3305 GSA Schedule #: GS-10F-0062R

Submitted to:

Tamia True, Contracting Officer Brian Baker, Contract Specialist Acquisition Services Directorate National Business Center Department of the Interior





Table of Contents

Section 1.	Introduction	6
1.1	1 Project Timeline	6
	1.1.1 Ongoing Review	7
	1.1.2 Engaging Internal and External Stakeholders	7
1.2	2 Organization of this Document	8
Section 2.	Methodological Overview	10
2.1	1 Introduction	10
2.2	2 Data Collection and Integration	11
2.3	3 Statistical Modeling	12
2.4	4 Case Studies	12
Section 3.	Data	13
3.1	1 Introduction	13
	3.1.1 Geographical Unit of Analysis	13
3.2	2 Data Sources	15
	3.2.1 NTIA Performance Progress Reports	15
	3.2.2 American Community Survey Data	16
	3.2.3 U.S. Census Zip Code Business Patterns Data	17
	3.2.4 USDA Economic Research Service Data	17
	3.2.5 National Broadband Mapping Data	17
	3.2.6 FCC Form 477 Data	17
	3.2.7 Recovery Act Reports	18
	3.2.8 Community Anchor Institutions	18
	3.2.9 Data Obtained from Case Study Participants	18
Section 4.	Sampling and Measurement	19
4.1	1 Introduction	19



4.2 0	Gran	t Selection	19
4.	.2.1	Process	19
4.	.2.2	Selected Grants	20
4.3 \$	Selec	cting the Sample for the Study	30
4.4 N	Neas	suring Impacts Some Complexities	32
4.5 N	Matc	hed Pairs Selection	33
Section 5. S	Statis	stical Estimation Methodology	34
5.1 li	ntroc	duction	34
5.2 F	Purpo	ose of the Statistical Estimation Methodology	34
5.3 E	Estim	nating Short-Term Economic Impacts	35
5.4 E	Estim	nating Intermediate-Term Impacts	36
5.	.4.1	Characteristics of BTOP Infrastructure Projects	38
5.5 F	Prop	osed Methodology	39
5.	.5.1	Strengths and Weaknesses of this Approach	41
5.6 E	Estim	nating Longer-Term Impacts	41
5.7 S	Sens	sitivity Analysis	42
Section 6. C	Case	e Studies	43
6.1 li	ntroc	duction	43
6.2 F	Purpo	ose of Case Studies	43
6.3 N	Neth	odology	43
6.	.3.1	Grantee Notification and Communication	43
6.	.3.2	Data Collection	44
6.	.3.3	Case Study Report	47
Appendix A.E	BTO	P Overview	50
Appendix B.L	itera	ature Review	52
B.1 li Broac	nforr dban	mation Technology, Information and Communication Technology, and nd Impacts on the Economy	52
В	.1.1	Community Impacts	52
В	.1.2	Employment Impacts	53



B.1.3 Business and Productivity Impacts	55
B.2 Broadband Access and Adoption	56
B.2.1 Factors Affecting Broadband Availability	57
B.2.2 Factors Affecting Adoption	58
Appendix C.Social Impacts	61
C.1 Introduction	61
C.2 Data Availability	61
C.3 Measuring Impacts on Education	62
C.3.1 Online Learning	64
C.3.2 K-12 Institutions	64
C.3.3 Community College & Four Year Institutions	66
C.4 Measuring Impacts on Healthcare	67
C.5 Measuring Impacts on Public Safety	75
C.6 Measuring Impacts Quality of Life/Standard of Living and Civic Engagement	79
Appendix D.Glossary	84
Appendix E.Bibliography	86

Figures

Figure 1: Final Report Deliverable Timeline	7
Figure 2: Methodology – Functional View	10
Figure 3: OneCommunity Ohio infrastructure Project	39

Tables

Table 1: Selected Grants	20
Table 2: Attributes of Selected Grants	26
Table 3: BTOP Grant Rural / Urban Designations	29
Table 4: Mapping Cost Classifications to Three-Digit NAICS to IMPLAN 440	36



Table 5. Potential Data Sources and Variables	62
Table 6: Adoption Rates	73



Section 1. Introduction

In September 2010, the National Telecommunications and Information Administration (NTIA) contracted with ASR Analytics, LLC (ASR) to conduct an evaluation of the economic and social impacts of its Broadband Technology Opportunities Program (BTOP) project grants. Over the past several months, a team consisting of ASR and Grant Thornton LLP developed a methodology that combines econometric analysis and case studies to estimate impacts of these grants, both quantitatively and qualitatively.

The culmination of this study will be the production of a Final Report that quantitatively and qualitatively measures the economic and social impact of BTOP grants (including CCI, SBA, and PCC). This report is intended to assess whether NTIA's implementation of BTOP has encouraged the fulfillment of the American Recovery and Reinvestment Act of 2009's (Recovery Act) goals. It will have sufficient methodological, theoretical, and pragmatic rigor to withstand critical scrutiny by technically proficient practitioners. The analytical work in the report will be of similar quality as that published in a peer-reviewed academic journal.

The centerpiece of the Final Report will be an assessment of the benefits that BTOP grant awards have on availability and adoption and in achieving economic and social benefits in areas served by the grantees. The economic impact analysis will examine the distribution of resources and value across groups and individuals as a result of BTOP grants. Social impacts will be measured, including incremental impact on social value and social welfare as a result of BTOP grant awards. The analysis of social impacts will include relative efficiency and cost effectiveness of policies and actions associated with the program.

The purpose of this Study Design deliverable is to present the research design that will be used to develop the Final Report. It provides a justification for the research design chosen to perform the economic and social impact assessments, including a detailed explanation of the study criteria, approaches, methodologies, data, and models to be used in the analyses.

1.1 Project Timeline

As shown in Figure 1 below, the results of these activities will be provided in two interim reports and a final report to be produced at the end of the study. We expect that after the production of each interim deliverable there will be an assessment of progress towards the goals of the project and an evaluation of actions to be taken to maintain or improve the quality of the final report. These interim reports (due in the second and third years of the project), will also provide the opportunity to discuss interim results with our Academic Advisory Committee and, to the extent desired by NTIA, with other stakeholders in the broadband community.





Figure 1: Final Report Deliverable Timeline

1.1.1 Ongoing Review

As we developed the research criteria, methodology, data, and sampling designs, we subjected our work to review by several highly regarded academics with experience in the areas of economic development and econometric estimation techniques. This Academic Advisory Committee operates on an independent basis. Although they worked on a subcontract to ASR, they were asked to provide impartial guidance on our study design. The evaluation study team briefed the Academic Advisory Committee throughout the iterative process of data discovery and our initial methodology, interpreting the significance of a variety of methodological factors, and revising the methodology. The evaluation study teams' Academic Advisory Committee includes Dr. Marvin Sirbu, Dr. Kingsley Haynes, and Dr. Roger Stough. We will continue to work with the Academic Advisory Committee throughout the project timeline.

1.1.2 Engaging Internal and External Stakeholders

A key component of our study design is the proactive engagement of stakeholders. Engaging internal and external stakeholders during the process of developing our study design will help to communicate the following to a wide audience:

- The objectives of the study
- The scope of the study
- The availability and comprehensiveness of existing data, including BTOP administrative data, public use data, and third party sources
- The technological and data limitations encountered to date



To that end, we conducted interviews with stakeholders in order to better understand NTIA and BTOP programmatic goals, as well as the underlying objectives for the evaluation study. During these meetings we obtained important feedback that we incorporated in the study design presented here.

Our team also invited several well-known academic researchers specializing in the areas of quantitative and qualitative assessment of broadband impacts to participate in a Methodology Conference that was held on April 1, 2011. Convening this Methodology Conference provided insight into the questions that the research community is likely to have regarding the evaluation study. It also provided input on the strengths and limitations of various data sources and quantitative and qualitative methods for measuring broadband impacts. This feedback was incorporated into this Study Design document.

1.2 Organization of this Document

This subsection presents the structure of the remainder of the document.

- Section 1 is the introduction to this document.
- Section 2 presents a brief overview intended to present the highlights of our methodology in a compact format. We do not discuss methodological details in this section and the reader is referred to Sections 4 and 5 for additional information.
- Section 3 presents a discussion of the data sources we will use for this project. During the first six months of this project, the evaluation study team, in consultation with the Academic Advisory Committee, telecommunications industry experts, and several BTOP grantees, gathered and assessed NTIA Program data as well as public and private data sources that were available to measure the economic and social impacts of BTOP. One driver of the evaluation study design was the assessment of data currently available, data that will be available through quarterly and annual updates, and data that can be gathered during our case studies.
- Section 4 presents the sample of PCC, SBA, and CCI grants that were selected as the basis for the study, the rationale for their selection, and descriptive statistics on each grant selected. The list of grants to be used as the basis for the study has not been finalized and may be changed in the future. This section discusses some areas of measurement complexity and the selection of geographic areas matching those selected for the purpose of comparisons over time.
- Section 5 presents our statistical methodology. This methodology was developed based on our findings from current scholarship on estimating the impacts of broadband presented in the literature review, as well as guidance from our Academic Advisory Committee. This section also describes our statistical methodology for estimating short-, intermediate-, and long-term impacts of BTOP grants. This analysis will produce interim results during the course of the study as well as material to be incorporated into the Final Report.
- Section 6 presents our case study methodology. We detail our methodology for how case studies will be conducted, data we will gather to support case study site visits, and the outputs of our analysis the case study reports.
- Appendix A presents a short overview of the NTIA BTOP program. It is intended as a brief orientation and reference. This information informs the methodological choices included in the study design.
- Appendix B presents a review of the current literature. We have reviewed current state of the literature surrounding the economic and social impacts of broadband technologies. We have reviewed over 500 articles in academic literature, technical



publications, and other sources. The bibliography for this literature review concludes this document.

- Appendix C presents an analysis of the current research on data, measures, and findings studied for social impacts within the context of broadband adoption.
- Appendix D presents a glossary of commonly used terms and abbreviations.
- Appendix E presents our bibliography, which contains the sources reviewed in the creation of this study design.



Section 2. Methodological Overview

2.1 Introduction

This section presents a brief overview intended to present the highlights of our methodology in a compact format. We have not discussed methodological details in this section and the reader is referred to Sections 4 and 5 for additional information.

As shown in Figure 2, our study methodology will include short-, intermediate-, and long-term impact assessments. In addition, the study will develop a source of public use data that may be used by researchers in the future who are seeking to better understand the impact of broadband on economic and social outcomes.

BTOF	Project – Functional View		
	Short Term Impacts	Intermediate Term Impacts	Long Term Impacts
Statistical Estimation	Determine direct, indirect and induced employment effects of BTOP spending using Input-Output modeling	Identify capacity provided and used Estimate uses of provided capacity Matched-pairs analysis	Calibrate forecasting model using results of intermediate term model Create impact forecasts
Case Studies	Describe effect of BTOP grants as catalyst to organization around broadband projects	Demonstrate community level industry specific impacts	Develop before and after comparisons Describe long term goals and expectations
Data Management		Report on changes in speed data developed by NTIA. Correlate speed availability to potential uses by CAIs.	Create public use data source for use by researchers over a longer term.

Figure 2: Methodology – Functional View

Our methodology includes three major task areas:

- Data Management
- Case Studies
- Statistical Estimation

Each of these will be described in more detail in the subsections below.

Based on previously reviewed research and the goals of this evaluation, we have defined the following outcome areas under which to categorize our potential impact measures or criteria:

• **Broadband Access in Unserved or Underserved Areas:** These are measures related to new broadband service provided, the types of broadband technologies implemented, the improvements to broadband speeds provided, and the types of outcomes that can be achieved with these various speeds and technologies.



- **Economic Impacts:** This area is primarily focused on jobs, but also includes other economic impacts such as wages, property values, and the number of firms in a region.
- Educational Impacts (including focus on impacts to identified vulnerable populations): These are measures, broken out by different levels of education, in changes to elements of educational content distribution and instruction.
- Healthcare Impacts (including focus on impacts to identified vulnerable populations): These are measures of changes in various elements of the provision and administration of the healthcare services including Health Information Technology (IT), e-Care, Electronic Health Record (EHR), Telehealth, and Mobile Health.
- **Public Safety Impacts:** These are measures related to both the provision and administration of public safety activities including those associated with law enforcement agencies, fire safety agencies, and emergency medical services.
- Quality of Life Impacts: These are measures related to various indicators of changes to standard of living associated with the uptake of broadband technologies such as community and civic participation (e.g., voter registration, volunteerism, and membership in civic associations), establishment of home businesses, time spent online, internet purchases, telecommuting rates, home schooling rates, increases in communications via email and other online channels.

2.2 Data Collection and Integration

The foundation of any empirical study is a collection of high-quality data. Our evaluation study will include both quantitative and qualitative data, gathered through a variety of modalities. Our data sources will include:

- NTIA Performance Progress Reports (PPR)
- American Community Survey (ACS) Data
- U.S. Census Zip Code Business Patterns (ZBP) Data
- U.S. Department of Agriculture (USDA) Economic Research Service (ERS) Data
- NTIA Broadband Mapping data
- Federal Communication Commission (FCC) Form 477 data
- Recovery Act reports
- Community Anchor Institution data
- Data obtained from case study participants

As part of our ongoing project activities will establish data quality routines and validate data we have received to ensure we are using accurate data for our analysis. Once data has been transformed into analysis data sets, we will perform preliminary analyses on those data to identify patterns, outliers, and generate summary statistics. While data gathering will only be complete at the end of the project, we will create ongoing documentation of database tables, database design, and data formats as data availability permits. We expect a substantial portion of the data gathering activities to be scheduled for Year 1. At the end of the project the data we have used, together with all relevant documentation will be packaged for use by future researchers.



2.3 Statistical Modeling

We will use a statistical estimation methodology to quantitatively assess the economic and social impacts of BTOP grants, especially infrastructure programs. The impacts of infrastructure programs are most amenable to quantitative analysis, and the methodology we present here is targeted toward the evaluation of these programs. Infrastructure projects have also received the majority of BTOP funding. In some cases PCC and SBA grants may have quantitatively measurable quantitative impacts, especially in the short-term. To the extent possible we will attempt to characterize these effects using quantitative methods.

2.4 Case Studies

We will perform case studies of selected CCI, PCC, and SBA grants to provide a window into the initial impacts of BTOP awards and provide ongoing snapshots of the Program's economic and social impacts, allowing for review and possible model adjustments for the longer-term, in-depth longitudinal study that will be a centerpiece of the Final Report.

The case studies will identify how the grantee maximized the impact of the BTOP investment; identify successful techniques, tools, materials and strategies used to implement the project; highlight best practices; and gather evidence from grantees, anchor institutions, and various publicly available data as to the impacts of the project in the community. The case studies will provide qualitative information; and, depending on the individual evaluation efforts of selected grantees, perhaps some quantitative data related to economic and social impacts of the activities resulting from the BTOP funded grant.



Section 3. Data

3.1 Introduction

During the first six months of this project, the evaluation study team, in consultation with the Academic Advisory Committee, telecommunications industry experts, and several BTOP grantees, gathered and assessed NTIA Program data as well as public and private data sources that were available to measure the economic and social impacts of BTOP. One driver of the evaluation study design is the assessment of data currently available, data that will be available through quarterly and annual updates, and data that can be gathered during our case studies site visits.

Given the scope, complexity, and implementation timelines of many of these grants, most of the projects are not expected to be complete for months or even years. Therefore, data on project outcomes, affected populations, connected anchor institutions, miles of fiber, number of towers, etc., will be evolving over the period of performance. It is also possible that grant projects could make changes to the approach described in their initial applications. Because the implementation of these projects is ongoing during the evaluation, the evaluation study team will evaluate grants as they evolve over their life cycle.

We will use a SharePoint portal as the technology for project metadata and document management. This secure environment is the central control point for all project materials and artifacts. The portal provides access to project artifacts that are planned, in process, and completed. This directly supports the corporate level and project level quality plan by retaining research data, case study workspaces, contact lists, literature review articles, background information, and controlled versions of all deliverables. This system provides the evaluation study team with the ability to automate data quality routines, as well as data integration tasks, so we can quickly reproduce statistical findings as the NTIA program data and public use data are being updated through quarterly and annual data calls. Data received during case studies can be easily added to our data sets and checked against any data previously submitted.

3.1.1 Geographical Unit of Analysis

NTIA awarded grants to applicants representing all states and territories of varying sizes, ranging from a single county to entire states, or even multiple states. Overall, grantees have reported that projects will connect or improve speeds at over 24,000 anchor institutions throughout the nation. Additionally, BTOP will fund over 35,000 new or upgraded workstations and more than 3,500 new or upgraded public computer centers. Grant recipients have also estimated that sustainable broadband adoption and awareness campaigns will reach a total of 40 million people.

The Final Report will examine the economic and social impacts of BTOP funded grants on communities, households, individuals, and anchor institutions. In order to assess impacts at these varied levels, data must be available at a variety of levels, including the county, ZIP Code Tabulation Areas (ZCTA), Census tract and Census block levels. The following describes the data we will seek to obtain at various levels of detail:

• County level data provides background on the areas that are served by BTOP projects. Public use data sources such as the Current Population Survey (CPS) provide a rich set of descriptive statistics on a county-by-county basis. While there is likely to be considerable variation within particular counties regarding social or economic conditions, county level analysis has been used to describe populations served by broadband and those who are excluded.



- A ZCTA comprises a group of census blocks that roughly maps to ZIP Codes. Older FCC 477 data is reported at the ZCTA level, which provides a source of broadband availability data in addition to the NTIA Broadband Mapping Data. In addition, broadband industry pricing and speed information is typically quoted in terms of ZIP Code. While matches between ZIP Codes and ZCTAs are not exact, with due care a ZIP Code may be found that corresponds to a particular ZCTA in most cases. The use of ZCTAs as one of the fundamental geographic units of analysis provides several benefits:
 - ZCTAs are contained in only one county.
 - Each census block is contained in only one ZCTA. This allows aggregation of NTIA Broadband Mapping data to the ZCTA level.
 - ZCTA is the lowest level of geographic aggregation available for some Census data.
 - Matches can be made between ZIP Codes and ZCTAs, although this must be done carefully. This allows for the incorporation of ZIP Code indexed data, such as pricing data into the analysis.
- Census tract data provides additional information on geographically smaller areas, although this information is available less frequently and for fewer measurements of interest when compared to higher levels of geographic aggregation such as county and ZCTA. Census tract data has the following benefits:
 - Each census block is contained in only one census tract. This allows for the aggregation of NTIA Broadband Mapping data to the Census tract level.
 - Detailed demographic data is available from the ACS.
 - More recent FCC 477 data is available at the census tract level, which provides additional information on broadband availability.
- Census block data is the lowest level of geographic aggregation available for use in this study. NTIA Broadband Mapping data is generally available at the census block level, although in some cases it is more detailed than that.
 - In general, census block data will be used as the basis of the measures we will develop to track broadband development over time.
 - Aggregation of census blocks to the census tract, ZCTA or county level will be necessary to interpret the implication of changes to broadband access, availability and adoption.
- Anchor institutions are expected to play a key role in the promotion of broadband use, and their activities and experiences will provide insight into key factors influencing access and adoption.
 - Anchor institutions will be identified by name in the Performance Progress Reports provided by NTIA infrastructure grant recipients. We will track these reports to identify anchor institutions affected by BTOP grants.
 - Anchor institutions are also identified in the NTIA Broadband Mapping data, which tracks information on broadband speed and availability. This information will be used to supplement the data provided in the PPRs.
 - Some anchor institutions may be included in the case study process, for both PCC/SBA case studies and for CCI case studies. Information from these anchor institutions will be collected during interviews and will be qualitative, rather than quantitative in nature.



3.2 Data Sources

Data will be integrated from a variety of data sources to create the analysis datasets. The major data sources we have identified include:

- NTIA PPRs
- American Community Survey Data
- U.S. Census ZBP Data
- USDA ERS Data
- NTIA Broadband Mapping data
- FCC Form 477 data
- Recovery Act reports
- Community Anchor Institution data
- Data obtained from case study participants

Each of these data sources is discussed below in more detail.

3.2.1 NTIA Performance Progress Reports

NTIA program data provides a valuable starting point for our analysis of the effects of BTOP grants. While our study is intended to present the social and economic impacts of the activities actually undertaken by grant recipients, as opposed to those actions described in applications or other plans, grant applications provide information that will assist the evaluation study team in early stages of the project. Our primary source of project information will be data reported by grantees through PPRs.

After discussions with NTIA, we believe we have assembled the most comprehensive set of program data currently possible, taking into account the proprietary nature of some data sources and inherent uncertainties in the implementation of longer-term projects. We plan to reassess these program data sources over time, including the incorporation of PPR information as it becomes available.

Quarterly and Annual PPRs are a requirement from each grantee and will summarize progress to date for each project. We will collect, integrate, and track updates to those data elements quarterly and annually in order to measure progress. These reports provide quantitative and qualitative descriptions of the project's status and specific grant funded activities. PPRs for all project types include an overall assessment for the current and upcoming quarters percent complete within different categories of project activities. Beyond this, however, the PPRs include more detailed information on activities specific to each grant type. This information will be useful in identifying potential project impacts specific to a particular location that can be verified and validated through the case study. This information will also be useful in identifying project specific areas of inquiry to pursue further with individual anchor institution contacts. This will help monitor accuracy and validate data we have at a given point, as well as data received from each grantee. As these updated data are collected, BTOP economic impact forecasts will be calculated to reflect the most up-to-date data.

The PAM database has been provided as one source of BTOP program data, including actual grant awards. This database provides quarterly information about grant location, amount, recipient information, and a set of measures and characteristics specific to the grant. As these data are updated, more information about the specific accomplishments at particular locations will become



available and will allow the evaluation study team to associate the BTOP grant award data with geographic areas and enhanced broadband availability. As such, it forms a core data source.

3.2.2 American Community Survey Data

As part of our analysis we will characterize the economic and demographic characteristics of areas and correlate these to broadband access, availability, and adoption. Most of our baseline sociodemographic data will be generated from the American Community Survey. As of the 2010 United States Census, the long form will not be used to collect detailed socio-demographic data. Instead, the American Community Survey has been focused on this data gathering effort.

Data for the ACS is produced from a random sample of households. Geographic areas are surveyed yearly, and data from successive years is aggregated into 3- or 5-year estimates. In order to preserve confidentiality, there is a built-in tradeoff between the level of geographic granularity at which data is available and the number of years of survey data that are aggregated in order to compute summary statistics. The data for this project will be drawn primarily at the census tract and ZBP level. This data is only available from an average of 5 successive ACS survey years. The data drawn from the ACS will be used to develop control variables that characterize the geographies included in the study. These characteristics are expected to be slowly changing variables, which makes for an acceptable tradeoff between the number of years used and the size of the area examined. Potential descriptive statistics include:

- Age
- Ethnicity
- Place of birth, citizenship, and year of entry
- Language spoken at home
- Educational attainment
- School enrollment
- Residence 1 year ago
- Veteran status and period of military service
- Disability
- Whether grandparents are caregivers
- Income
- Whether food stamp recipient
- Labor force status including industry, occupation, and class of worker
- Place of work and journey to work
- Work status last year
- Vehicles available
- Financial characteristics
- Tenure in current residence
- Value of current residence
- Rental rates
- Health insurance coverage



• Service-connected disability

3.2.3 U.S. Census Zip Code Business Patterns Data

On a yearly basis, the U.S. Census Bureau provides data on ZBPs. These data are currently available for 2009 and will be updated to 2010 by the end of calendar year 2011. This data includes statistics on the number of business establishments, the classification of these establishments by NAICS code, employment, and payroll. All of these statistics are available on a yearly basis at the ZIP Code level. We will use these data to characterize the areas affected by the grants we examine.

3.2.4 USDA Economic Research Service Data

The USDA ERS, in cooperation with the U.S. Census, also categorizes areas as urban or rural, according to commuting and employment patterns. These data provide the evaluation study team with a convenient measure of an area's urban or rural character. The definition of rural depends in part on how the measure designed to capture rural characteristics is used. As part of our methodology we will examine measurements of "rurality" used in previous studies in order to develop a meaningful measure for our study.¹

3.2.5 National Broadband Mapping Data

The centerpiece of our economic criteria are measures obtainable using the National Broadband Mapping Program Data. On February 17, 2011 the NTIA brought online the National Broadband Map (NBM), "the first public, searchable nationwide map of broadband Internet availability."² These data are provided by NTIA in cooperation with the FCC and the 50 states, territories, and District of Columbia. These data are based off of surveys of local broadband providers and has information on availability, technologies, and speed, by census block. The NBM project also makes available information about local community anchor institutions and their access to broadband services. This information can be used to create measures for broadband penetration in these core institutions.

Within the NBM Web site: "users can search by address to find the broadband providers and services available in the corresponding census block or road segment, view data on a map, or use other interactive tools to compare broadband across various geographies, such as states, counties, or congressional districts." In addition to specific searches, the site allows users to download entire data sets with information at the census block level. However, certain challenges have been brought to light surrounding the NBM. Ford states that measurement and sampling error are roots for the immediate issues with these data. It is anticipated that, over time, these "errors" will be resolved.³

3.2.6 FCC Form 477 Data

FCC 477 data refers to information submitted on the FCC's Form 477 by providers of broadband services, and comprising location, service levels, and speed. This allows the evaluation study team to measure broadband penetration on a ZCTA and census block level, forming a key component of our overall data store. Until recently, this was the most detailed data on broadband service provision available.

³ Ford, George S. 2011. "Challenges in Using the National Broadband Map's Data." *Phoenix Center Policy Bulletin*, no. 27 (March). http://www.phoenix-center.org/PolicyBulletin/PCPB27Final.pdf.



¹ "Rural and Suburban America: When One Definition is Not Enough." Last modified August 4, 2011.

http://www.commerce.gov/blog/2011/08/04/rural-and-suburban-america-when-one-definition-not-enough.

² "Commerce's NTIA Unveils National Broadband Map and New Broadband Adoption Survey Results." Last modified February 17, 2011. http://www.ntia.doc.gov/press/2011/NationalBroadbandMap_02172011.html.

A recent change in FCC Form 477 requirements now provides finer tracing on broadband supply penetration. This change should allow the evaluation study team to leverage biannual data around broadband bandwidth footprints and specific technologies at the grain of census tract. These data are available from March 2010 onwards.

3.2.7 Recovery Act Reports

To supplement our input/output analysis, we will make use of data submitted by grant recipients in their Quarterly and Annual PPRs. Since BTOP funding was provided by the Recovery Act, grantees must submit Quarterly PPRs throughout the duration of their projects. These quarterly reports are due no later than 10 days after the close of each quarter and are submitted to FederalReporting.gov. All of the reports are made available publicly after a federal agency performs a quality control and data validation. This process helps ensure transparency and accountability for Recovery Act award management.

The first report provided by each recipient will be used as a baseline report for comparative purposes. Recipients can use the Excel or XML report templates available on FederalReporting.gov for report submissions. The noteworthy variables included in the report templates are total amount of Recovery Act funds received, number of jobs, activity code (NAICS or NTEE-NPC), total federal amount of Recovery Act expenditure, and total federal Recovery Act infrastructure expenditure.

We are most interested in the data related to capital expenditures and job creation estimates and will use this data for a comparison with the results from our input/output analysis.

3.2.8 Community Anchor Institutions

Community anchor institutions play a key role in our analysis, both as sites for PCC or SBA activity and as proxies for CCI investment in particular geographic areas. NTIA estimates that infrastructure projects "propose to connect or improve speeds for approximately 24,000 community anchor institutions."⁴ Understanding the location and type of anchor institutions that are connected by BTOP projects provide insight into geographic areas that should be considered for additional analysis as well as the types and locations of institutions that are likely to be affected by infrastructure projects.

3.2.9 Data Obtained from Case Study Participants

In addition to the data sources described above, we will solicit information from interview participants as part of our case study methodology. We do not anticipate having a set group of questions for each participant. Rather, qualitative information will be gleaned from the conversations we have regarding the projects included in the case studies. Case study interview guides will be targeted to the particular goals, intended outputs/outcomes, types of services provided, community conditions and anchor institution/service location types of each grant and, as such, will be highly customized for each grant location. The guides will be designed to collect information from grantees, anchor institution/service locations, local economic development professionals (where identified by grantees or anchor site contacts), last mile providers (for CCI grants) and individual users (when such access is available and appropriate.) Also, particular facts regarding a project may be confirmed during the interview process and we will make note of these as applicable in our project data.

⁴ The Broadband Technology Opportunities Program. 2010. "Expanding Broadband Access and Adoption in Communities Across America, Overview of Grant Awards." Last modified December 14, 2010. http://www.ntia.doc.gov/reports/2010/NTIA_Report_on_BTOP_12142010.pdf.



Section 4. Sampling and Measurement

4.1 Introduction

This section presents the sample of CCI, PCC, and SBA projects that were selected as the basis for the study, the rationale for their selection, and descriptive statistics on each grant selected. The final list of grants to be used as the basis for the study has not been finalized and may be changed in the future. Finally, this section discusses some areas of measurement complexity and the selection of geographic areas matching those selected for the purpose of comparisons over time.

4.2 Grant Selection

In general, the selection of grants was purposeful and not meant to yield a statistical sample. The total number of selections was defined in our technical proposal and guided by the period of performance for the evaluation study, the logistics involved with conducting case study site visits, NTIA's requirements for the timing of interim reports, and the resources available for staffing the evaluation study team. NTIA and the evaluation study team worked collaboratively to develop a consistent set of criteria for the selection of grants. However, some differences in criteria exist due to the nature of CCI, PCC, and SBA projects. We discuss these differences below.

4.2.1 Process

The primary focus of the final report is to investigate and measure the economic and social impacts of BTOP grants at the community, individual, and anchor institution levels. Therefore, from a grant selection perspective, grants that have more focused outcome goals, a more targeted impact area, and/or identified anchor institutions lend themselves better to this analysis.

The diversity in types of grants, their objectives, target audiences, types of technology, timing of project milestones and completion schedules, and geographic scope are just some of the many factors that require a customized and specific approach for each of the selected grants. In addition, given the number of PCC and SBA selections are limited to 15, and the number of CCI selections are limited to 12, generalizations of findings across the grants requires careful interpretation. A further consideration is that many of these projects will be "in flight" at the same time we are conducting the case studies; therefore, we have suggested that NTIA consider grants that are likely to be closer to completion, especially since we will be conducting anchor visits. We discussed with NTIA several other considerations in the selection process, and they are listed below.

These considerations included:

- Is the grantee (or sub-recipient as the case may be) a willing participant that will be able to
 engage in a meaningful conversation about the impacts both forecasted and actual of
 their project?
- Is this a grant in which there is some high level of confidence that it will be completed without significant technical or financial obstacles? In other words, selected locations can represent grants of varying quality, but extremely troubled or returned grants should not be included.
- Is this a grant for which there is some reason to believe there will be anchor institutions that can be identified and will have been connected (or, in the case of SBA/PCC, are open and operating) by the time the evaluation study team will be conducting case study site visits?



The next level of consideration included various technical and administrative grant attributes including:

- Impact area or project focus with respect to the areas of analysis defined for the study, e.g., education, workforce development, healthcare, or public safety
- Urban versus rural location
- Grant size as measured by federal funding level
- Applicant Type, e.g., State Government, Non-Profit, For Profit, Small Business, Tribal
- Provision of service differentiators such as:
 - Types or style of training, outreach, or media engagement (in the case of PCC/SBAs)
 - Technology deployed, business model of applicant and/or sub-recipients, and/or middle mile versus last mile (in the case of CCIs)
- Type of population served
- Existence of a grant specific evaluation component being conducted by the grantee or a partner organization

Synergies across CCI, PCC, and SBA grant selections in a particular location were considered only if the first two levels of consideration kept the grants in the pool of potential selections. Finally, logistical considerations including budgetary and travel considerations were applied in making the final selection of grants.

4.2.2 Selected Grants

Table 1 presents a list of 15 selected PCC/SBA (7 PCC and 8 SBA) and 12 selected CCI grants along with a brief description of each project.

Recipient	Project Title	Description
		CCI
MCNC	North Carolina Rural Broadband Initiative	To construct over 1,300 more miles of fiber infrastructure to community colleges, libraries, schools, health and public safety facilities, and other community anchor institutions in 69 of the most economically disadvantaged rural counties.
South Dakota Network (SDN), LLC	Project Connect South Dakota	SDN, a partnership of 27 independent telecom providers, will add 140 miles of backbone network and 219 miles of middle mile spurs to SDN's 1,850-mile fiber-optic network.
Zayo Bandwidth, LLC	Indiana Middle Mile Fiber for Schools, Communities, and Anchor Institutions	Directly connect 21 Ivy Tech Community College campuses to the state's existing high-speed network for education and research, known as the I-Light network.

Table 1: Selected Grants



Recipient	Project Title	Description
Merit Network Inc.	REACH Michigan Middle Mile Collaborative	Build a 955-mile advanced fiber-optic network through underserved counties in Michigan's Lower Peninsula to serve institutions, businesses and households, the project will extend Merit's 1600 miles of existing network and intends to directly connect 44 anchor institutions.
Mid-Atlantic Broadband Cooperative	Middle Mile Expansion for Southern Virginia	Add 465 miles of new fiber to an existing 800- mile fiber network, focused on directly connecting 121 K-12 schools.
Executive Office State of West Virginia	West Virginia Statewide Broadband Infrastructure Project	Expand the state's existing microwave public safety network and add about 2,400 miles of fiber, expecting to directly connect more than 1,000 anchor institutions.
Massachusetts Technology Parks	The Massachusetts Broadband Institute: Mass Broadband 123	Link almost 1,400 community anchor institutions through a new middle mile fiber network by constructing over 1,300 miles of new fiber in areas.
OneCommunity	Transforming NE Ohio	Build 900 new miles of fiber, leveraging another 2,000 miles of existing network, across 20 counties.
University of Arkansas	The Arkansas Healthcare, Higher Education, Public Safety & Research Integrated Broadband Initiative	Deploy a new middle-mile fiber network across every county in the state that will extend, integrate, and enhance the capabilities of two major community-serving networks already in existence: the Arkansas Telehealth Oversight & Management (ATOM) Network and the Arkansas Research & Education Optical Network (ARE-ON).
Ocean State Higher Education Economic Development Administrative Network	BEACON 2.0	Construct a new fiber-optic backbone network to connect community anchor institutions and offer interconnection points for local broadband providers in all five counties in the state, as well as Bristol and Plymouth Counties in southeastern Massachusetts.
Delta Communications	Illinois Broadband Opportunities Partnership - Southern	Deploy a high-speed fiber middle mile network across a 23-county region of southern Illinois.
Lane Council of Governments	Oregon South Central Regional Fiber Consortium Lighting the Fiber Middle Mile Project	Enhance an existing fiber-optic backbone and deploy 124 miles of fiber-optic network to deliver broadband capabilities across three large, mostly rural counties Lane, Douglas, and Klamath.



Recipient	Project Title	Description	
PCC			
Florida A&M University	The FAMU Center for Work Force Development	Establish a new Center for Public Computing and Workforce Development on the Florida A&M campus to serve the public of Tallahassee and surrounding Gadsden and Jefferson Counties. The center also plans to serve as a resource to other public computing centers in the region	
Cambridge Housing Authority	Cambridge Housing Authority Community Computer Center	Re-open one computer center and expand the capacity of two public computer centers in public housing developments. The project will purchase 40 computers, replace 24 workstations, and add 16 new ones to serve approximately 384 new users per week. These centers will be located strategically across the city in the largest family public housing developments.	
SC State Board for Technical and Comprehensive Education	SC Reach for Success	Expand the capacity of 51 public computer centers and create 19 new computer labs across the South Carolina Technical College System's 16 member colleges. Plan to conduct surveys and outreach to determine the types of additional continuing education courses students and the community need, including English as a Second Language, Internet and computer skills, and technical and healthcare training.	
WorkForce West Virginia	One-Stop Public Computer Center Modernization	Enhance and expand public-use computers, connection speeds, and wireless capabilities at 20 WorkForce West Virginia One-Stop offices throughout the state. These offices provide referrals, career counseling, and other employment-related services. The AARP Foundation will provide training that focuses on job search assistance, life skills development and work-focused basic education to help individuals overcome barriers to successful employment.	
Michigan State University	Evidence Based Computer Center Deployment II	Partner with 3 community colleges to add workstations at 169 PCCs and create 38 new PCCs throughout the State of Michigan. The PCCs will be located at libraries, community colleges, public housing, and other community support organizations. The State Library of Michigan and Community Outreach Services Corporation, a for-profit entity, will provide assistance at public housing sites and other low- income community centers.	



Recipient	Project Title	Description
Technology For All, Inc	Texas Connects Coalition	Create 8 PCCs and upgrade 62 existing PCCs in libraries, workforce centers, public housing, government-operated community centers, and other community organizations. The PCCs will be located in urban and rural areas of southwest Texas.
Las Vegas-Clark County Urban League	Access to Computer Technology and Instruction in Online Networking (ACTION)	Expand capacity at 14 public computer centers and create 15 new public computer centers in public housing developments as well as community and senior centers throughout Clark County, Nevada. The 29 centers will offer computer classes, job training and certification programs, and community health programs through partner organizations. The project plans to hire at least 10 bilingual instructors capable of teaching courses in Spanish.
		SBA
California Emerging Technology Fund	Broadband Awareness and Adoption	Build awareness and understanding of broadband in vulnerable and low-income communities in Los Angeles, the Central Valley, Orange County, San Diego, and the Inland Empire using a targeted media campaign. Provide digital literacy training for low-income individuals, and partner with statewide organizations to coordinate an intensive media campaign, including multi-lingual outlets, and relevant online software applications targeting hard-to-reach, low-income communities, aided by outreach from trusted ambassadors and grassroots mobilization.
Foundation For California Community Colleges (FCCC)	California Connects	Provide outreach, training, and learning support to increase digital literacy skills and broadband usage in partnership with 33 California community colleges, 136 public libraries, public computing centers, and community-based organizations. Special emphasis placed on 18 Central Valley counties that contain a concentration of the target audience of low- income Hispanic residents.
Future Generations Graduate School	Equipping West Virginia's Fire and Rescue Squads with Technology and Training to Serve Communications	Community-based approach to stimulate broadband adoption, extend computer access, and provide training to low-income and predominantly rural communities across West Virginia by providing these services at 60 volunteer fire and emergency rescue stations. Project will include peer-to-peer outreach, newspaper and radio advertisements, signs to announce services, social networking, and a support Web site.



Recipient	Project Title	Description
Delaware Department of State	Delaware Library Job/Learning Labs	The project plans to serve residents by providing public access computers, assistive technology, videoconferencing capabilities, and wireless networking. The computer centers at the Dover, Georgetown, Seaford, and Wilmington libraries will become Job/Learning Labs focusing specifically on the needs of the unemployed, with specialized training for resume building, job search, and interview skills. This project will also upgrade public computer centers at all 32 public libraries statewide and deploy approximately 50 new computer workstations.
Greater Philadelphia Urban Affairs Coalition (GPUAC)	Freedom Rings: Sustainable Broadband Adoption	GPUAC, in partnership with 10 experienced organizations working with targeted vulnerable populations, proposes to increase broadband adoption by providing strategic outreach, broadband education, awareness, training, access, equipment and support to underserved populations in low-income communities throughout Philadelphia.
Connect Arkansas Inc.	Expanding Broadband Use in Arkansas Through Education	Implement a three-pronged program that will be piloted in the 15 poorest counties in Arkansas. Program focused on basic digital literacy training, entrepreneurship, and healthcare access. Upon successful completion of the training, families will receive a refurbished computer. A technology entrepreneurship program targeting low-income youth will also be developed and implemented. Working with the University of Arkansas Medical Services (UAMS), the project will also oversee a comprehensive broadband-based health awareness campaign.
City of Chicago	SmartChicago Sustainable Broadband Adoption	Spur economic development in five disadvantaged neighborhoods in Chicago by creating public computer centers at six community centers for working families and expand workstation capacity at four Business Resource Centers, as well as provide 1,500 residents and small businesses that complete a multi-session training course with laptops and netbooks. Project will also involve a citywide multilingual broadband awareness campaign including outreach by local community organizations in each neighborhood; and the creation of neighborhood-based Web portals.



Recipient	Project Title	Description
C. K. Blandin Foundation	Minnesota Intelligent Rural Communities	Reach each of Minnesota's 80 rural counties through education, training, technical assistance, and by removing barriers to broadband adoption. Training will be provided in computer literacy, online education, and workforce development, and 1,000 affordable refurbished computers will be distributed to low-income, rural Minnesota residents. Funding will also support the development of institutional broadband applications for schools and healthcare facilities to help increase broadband adoption. The U.S. Department of Commerce's Economic Development Administration (EDA) Center at the University of Minnesota, Crookston will help track the impact of the project, including the number of subscribers generated.

Table 2 presents coverage of the selected grants against the various attributes discussed above. The paragraphs that follow the table provide detail on how each of these categorizations were determined, what they mean for the case studies, and statistics on how the selected grants compare to the universe of grants in each area. Impact area is one of these attributes that benefits from some upfront explanation for context in viewing the table. The impact categories applied and their definitions include:

- Workforce Development Projects that have explicitly linked their location or program to job search by providing training in job skills, or training in how to use the Internet specifically to find a job. Also includes projects that explicitly provide training or other resources geared towards small businesses or to business development skills in general.
- Healthcare Projects that have explicit links to healthcare providers, training on how to use the Internet to find health information, or links to schools of medicine, public health or pharmacy.
- Education Projects that are explicitly linked to GED training or to distance learning. These are projects that are linked to the provision of skills or learning other than how to use the technology itself and, which are tagged "digital literacy" as described below.
- Digital Literacy Projects that include training for users on how broadband technology works and what they can do with it. This category should include most if not all PCCs and all SBAs. Most CCI projects do not have this component.
- Public Safety Projects that have explicit links to public safety outcomes. Though those projects using the 700 MHz band for their deployment are excluded from this study, other projects might have explicit links to public safety outcomes.

Though general digital literacy is not one of the impact areas suggested in the project statement of work (SOW) as a focus for this study, this impact area was created because it was determined that a vast number of PCC and SBA grants in fact focus on general digital literacy, as opposed to other of the impact areas noted in the SOW as key areas of analysis.

For PCC and SBA projects, this determination was made through a review of the grant application and fact sheet and the description provided of the training and other services to be provided. Most CCI grants, on the other hand, do not have a particular impact area focus, but rather include the construction or provision of various types of broadband infrastructure over a specified geographic area. A number of different types of anchor institutions engaging in a variety of impact area



activities may end up being involved in the project. There are some CCI projects, however, that do include a particular impact area focus in that they explicitly intend to connect to an existing statewide health network or educational network or consortium or they are predominantly connecting to one type of anchor institution. These instances were identified through a review of CCI applications and a cross-check with identified anchor institutions in instances when anchor institution data was available. It is important to note that the impact area categorizations, as they are based primarily on a review of project fact sheets developed prior to actual project implementation and including any PPR information to date, could be subject to change as additional project activities are implemented. Reviews of PPR data over the course of the study period may result in updates to these designations. The statistics presented here are a snapshot of how the grants appear to fall out against the impact areas at the time of the development of this study design. No determination was made for those projects that cover numerous counties and/or multiple states.

Selected Grants	Urban/ Rural	Award Amount	Applicant Type	Eval Program (Y/N)	Impact Area(s)		
CCI							
MCNC	Rural	\$75,757,289	Non-Profit		 Education 		
South Dakota Network (SDN), LLC	Urban	\$ 20,572,242	For-Profit				
Zayo Bandwidth, LLC	Urban	\$ 25,140,315	For-Profit		 Education 		
Merit Network Inc.	Urban	\$33,289,221	Non-Profit				
Mid-Atlantic Broadband Cooperative	Rural	\$16,044,290	Non-Profit		Education		
Executive Office State of West Virginia	Urban	\$126,323,296	Government				
Massachusetts Technology Parks	Urban	\$45,445,444	Government				
OneCommunity	Urban	\$44,794,046	Non-Profit				
University of Arkansas	Rural	\$102,131,393	Higher Education		Healthcare		
Ocean State Higher Education Economic Development Administrative Network	Urban	\$21,739,183	Non-Profit				

Table 2: Attributes of Selected Grants



Selected Grants	Urban/ Rural	Award Amount	Applicant Type	Eval Program (Y/N)	Impact Area(s)	
Delta Communications	Rural	\$31,515,253	For-Profit			
Lane Council of Governments	Urban	\$8,325,530	Government			
		PCC				
Florida A&M University	Urban	\$1,477,722	Higher Education	Y	 Digital Literacy Workforce Development Healthcare Education 	
Cambridge Housing Authority	Urban	\$698,924	Government	Ν	 Workforce Development Education 	
SC State Board for Technical and Comprehensive Education	Undetermined	\$5,903,040	Government	N	 Workforce Development Digital Literacy Education 	
WorkForce West Virginia	Undetermined	\$1,901,600	Government	N	Workforce Development	
Michigan State University	Undetermined	\$6,056, 819	Higher Education	Ν	 Workforce Development Education 	
Technology For All, Inc	Undetermined	\$9,588,279	Non-Profit	Ν	 Workforce Development Digital Literacy 	
Las Vegas- Clark County Urban League	Undetermined	\$4,680,963	Non-Profit	Ν	 Workforce Development Digital Literacy Healthcare 	
SBA						
California Emerging Technology Fund	Undetermined	\$7,251,295	Non-Profit	Y	• Digital Literacy	
Foundation For California Community Colleges (FCCC)	Undetermined	\$10,944,843	Non-Profit	Y	• Digital Literacy	



Selected Grants	Urban/ Rural	Award Amount	Applicant Type	Eval Program (Y/N)	Impact Area(s)
Future Generations Graduate School	Rural	\$4,461,874	Non-Profit	Ν	 Workforce Development Digital Literacy Healthcare
Delaware Department of State	Urban	\$1,899,929	Government	Ν	 Workforce Development Digital Literacy
Greater Philadelphia Urban Affairs Coalition (GPUAC)	Urban	\$11,804,015	Non-Profit	Y	DigitalLiteracyHealthcare
Connect Arkansas Inc.	Undetermined	\$3,702,738	Non-Profit	Ν	 Workforce Development Digital Literacy Healthcare
City of Chicago	Urban	\$7,074,369	Government	Y	 Workforce Development Digital Literacy
C. K. Blandin Foundation	Rural	\$4,858,219	Non-Profit	Y	 Workforce Development Healthcare Education Digital Literacy

None of the selected PCC grants are designated rural versus 29 percent urban and 71 percent undetermined; 25 percent of the selected SBA grants are rural versus 38 percent urban and 38 percent undetermined; and 33 percent of the selected CCI grants are rural versus 66 percent urban. Comparable statistics for all CCI, PCC, and SBA grants and the total universe of BTOP grants are presented in Table 3 below.



Grant Type	Rural		Urban		Undetermined	
	#	%	#	%	#	%
CCI	39	34%	19	17%	56	49%
PCC	6	9%	18	28%	41	63%
SBA	9	20%	11	25%	24	55%
All Grants	54	24%	48	22%	121	54%

Table 3: BTOP Grant Rural / Urban Designations

Designating BTOP grants as urban or rural, however, is a difficult and somewhat arbitrary exercise. The exact location(s) of the activities conducted under any given grant can be both numerous and, in some cases, hard to identify. For example, some of the selected grants can have tens or even hundreds of community anchor institutions at which their services are provided or to which their infrastructure are connected. Moreover, these anchor institutions can be located in geographically disparate locations, some of which may be rural and some of which may be urban. For purposes of this report, CCI grants were designated as rural or urban primarily based on the question in their application that asks whether "more than 75 percent of the grant will impact rural areas." If the answer was yes, the grant was classified as rural, if the answer was no, the grant was classified as urban. In instances where this question was not included in the application and/or no application was available, the project's fact sheet was reviewed and a judgment was made based on the description of the project provided. In the case of PCC and SBA grants, this question did not exist in the application. For these grants, review of the application (when available) and fact sheets was used to make a judgment on the urban or rural nature of the project on the whole.

A final consideration for grant selection was whether or not the grantee or an independent third party was conducting a formative or summative evaluation of the grant. Some grantees, in particular SBA grants, included an evaluation program as part of their grant application. Still others have recognized since award the value of collecting certain data on the outcomes and impacts of their projects. Some grantees have formed arrangements with other grantees in their state or other regional boundary in order to collect data on broadband access and adoption outcomes across projects. The existence of an evaluation program of this type was considered in terms of case study grant selection because such efforts will potentially yield data that the evaluation study team can incorporate into our analysis. These actions signal a willingness and understanding on the part of the grantees that certain measures of economic and social outcomes are only possible if they collect both baseline and ongoing data. These efforts by grantees will augment the information collected as part of the case study, and case study site visits.

Examples of evaluation efforts by case study grantees include those by Northern Illinois University (NIU) and the City of Chicago. NIU is attempting to evaluate issues such as:

- How did project impact access to households and businesses in the region?
- How did project impact use of broadband by households and businesses in the region?
- How did project impact employment in the region?
- How did project impact business vitality in the region?
- How are households and businesses using the increased bandwidth capacity?

They are doing this by collecting, amongst other things, baseline data on current use of broadband services, current price of broadband services, potential market for broadband services, regional wages, and regional domestic product through surveys of anchor institutions, households, and



businesses. We have engaged the NIU grant leadership to discuss ways the evaluation study team can assist in the development of data collection instruments, as well as analysis of the results derived from these efforts.

In the City of Chicago, the evaluation effort includes analysis of several elements through use of surveys and interviews including an evaluation of: (1) the implementation process (interviews); (2) programmatic outcomes for participants in various individual and business focused training initiatives (surveys and interviews); (3) organizational changes in capacity, leadership, collaboration, and resources for sustainability (interviews); (4) interpersonal and community-level outcomes including analysis of social networks and activities in such networks for FamilyNet participants (participant surveys); and (5) 2011 and 2013 city-wide survey tracking changes in technology use across community areas (including those communities involved in the BTOP grant). The city-wide survey will assist in the longitudinal assessment of the grant on residents by combining the results with a prior study of technology use in Chicago during the summer of 2008. The city-wide surveys will allow for comparison of changes in the grant recipient communities with other low-income community areas and with city averages.

Though not an explicit consideration for grant selection, the number and type of anchor institutions associated with case study grants will affect the manner in which the case study is conducted (see Subsection 4.3 below). Many of the selected case study grants have a large number, and in some cases wide variety, of anchor institutions at which their various services are targeted or will be located. As discussed in more detail in the Methodology sub-section below, this fact has implications on how the case study will be conducted and documented. The time and budget constraints of this project do not allow, for example, for a case study of the Michigan State University project to include a case study visit to each of the 207 institutions in which the project is placing PCCs, or to include a detailed analysis of the economic and/or social impacts of each of these PCCs on their individual communities. Rather, based on interviews with the grantee and the cost and logistical considerations associated with the exact location of the 207 PCCs, the evaluation study team will visit a small subset of these locations to observe, collect, and measure economic and social impacts. As a result, the case study reports will describe in detail only those specific locations visited, and where possible, generalizations will be made to the other locations. To the extent the grantee itself is collecting project-wide data, analysis of that data will be included in the case study, but the case study will focus on the observable outputs and outcomes of only the visited targeted anchor institutions and their associated communities or other defined areas of impact.

The selection of the locations the evaluation study team will visit will relate to the activities of the grantees at those locations. In situations where a grantee has demonstrated significant progress at several locations and/or types of community anchor institutions, the locations selected for case study site visits will be coordinated with the grantee and individuals in those locations knowledgeable of the association with BTOP.

4.3 Selecting the Sample for the Study

The statistical estimation and case study methodologies presented in this document will be performed on a sample of fifteen (15) PCC and SBA projects and twelve (12) CCI projects. These results will then be extrapolated to the BTOP as a whole to estimate the overall impact of BTOP. The grants selected for inclusion in the study are as follows:



PCC Grants

- Florida A&M University The FAMU Center for Public Computing and Workforce Development
- Cambridge Housing Authority Cambridge Housing Authority Community Computer Center
- SC State Board for Technical and Comprehensive Education SC Reach for Success
- WorkForce West Virginia One-Stop Public Computer Center Modernization
- Michigan State University Evidence Based Computer Center Deployment II
- Technology For All, Inc. Texas Connects Coalition
- Las Vegas-Clark County Urban League Access to Computer Technology and Instruction in Online Networking (ACTION)

SBA Grants

- California Emerging Technology Fund Broadband Awareness and Adoption
- Foundation For California Community Colleges (FCCC) California Connects
- Future Generations Graduate School Equipping West Virginia's Fire and Rescue Squads with Technology and Training to Serve Communications
- Delaware Department of State Delaware Library Job/Learning Labs
- Greater Philadelphia Urban Affairs Coalition (GPUAC) Freedom Rings: Sustainable Broadband Adoption
- Connect Arkansas Inc. Expanding Broadband Use in Arkansas Through Education
- City of Chicago SmartChicago Sustainable Broadband Adoption
- C. K. Blandin Foundation Minnesota Intelligent Rural Communities

CCI Grants

- MCNC North Carolina Rural Broadband Initiative
- South Dakota Network (SDN), LLC Project Connect South Dakota
- Zayo Bandwidth, LLC Indiana Middle Mile Fiber for Schools, Communities, and Anchor Institutions
- Merit Network Inc. REACH Michigan Middle Mile Collaborative
- Mid-Atlantic Broadband Cooperative Middle Mile Expansion for Southern Virginia
- Executive Office State of West Virginia West Virginia Statewide Broadband Infrastructure Project
- Massachusetts Technology Parks The Massachusetts Broadband Institute: Mass Broadband 123
- OneCommunity Transforming NE Ohio
- University of Arkansas The Arkansas Healthcare, Higher Education, Public Safety & Research Integrated Broadband Initiative
- Ocean State Higher Education Economic Development Administrative Network BEACON 2.0



- Delta Communications Illinois Broadband Opportunities Partnership
- Lane Council of Governments Oregon South Central Regional Fiber Consortium Lighting the Fiber Middle Mile Project

Grants were selected by NTIA in order to provide geographic diversity and representation of diverse project types. A statistical sampling method such as simple random sampling was not used to create the selected lists.

Our approach to the selection of the grants was motivated by the following factors:

- 1. Identification of the impacts of the projects will require information that goes beyond grant applications, PPRs or other publicly available sources. Gathering additional information from grant recipients requires in depth interviewing and examination of specific project data. The cost of performing these activities for every grant recipient is prohibitive.
- 2. Participation in the study on the part of the grantees is voluntary. Although grantees are required to provide Recovery Act reports and PPRs as part of their participation in BTOP, participation in this study is not required. This makes assembling the sample of grant recipients an exercise in selection of those who are most likely to benefit from or show interest in the study.
- 3. Grant recipients have different levels of internal measurement and reporting planned as part of their projects. As an example, some grant recipients have included a survey or study as part of their projects, while others have not. To the extent that this type of data gathering activity is planned by the recipient, the inclusion of that recipient in the study improves the quantity of data available.

After developing estimates of project impacts for the selected samples, we will extrapolate the results to BTOP as a whole. To the extent that data sources are available, complete, and accurate we will use those sources. Some key pieces of information may be missing for some projects, either because of the project type or due to limitations in public data sources. In these cases we will impute data as required to develop overall program impact estimates. Variation surrounding these imputations will be included in our sensitivity analysis to determine the extent to which they might affect the overall conclusions of the study. Likewise, those areas selected as "controls" in the matched-pairs analysis (see below) might have more limited data availability versus the sampled projects.

4.4 Measuring Impacts -- Some Complexities

The methodology presented here is designed to address several practical difficulties that arise due to the structure of the BTOP program and the time available for completion of the study. Most of these practical considerations stem from the focus of BTOP investments on middle mile infrastructure and from the goals of the Recovery Act in general, which focus on wide-scale impacts on infrastructure investments that can be expected to provide benefits over an extended period of time.

- The primary hurdle to overcome is the long timescale of many BTOP infrastructure projects. While all projects are expected to expend their BTOP-provided funding during the same time period as this study, it is to be expected that larger infrastructure projects would be completed toward the end of the study period. This will limit the amount of time available during the study period for projects with longer durations.
- Public use data sources are expected to lag by at least a year and perhaps more both for variables that would be used to measure social or economic impacts and variables that might be used to identify co-occurring factors that would be used as control variables or for matching similar areas. While these delays are inevitable due to the necessity of



maintaining high-quality public use data sources, they will serve to push the real time at which a study using them will be feasible.

- There are likely to be delays between the availability of additional infrastructure due to projects funded by BTOP grants and adoption of technologies that will take advantage of that additional capacity. This is due to the remaining technological changes that must occur between the middle mile infrastructure backbone and the ultimate user of the technology. Limiting factors could include technological limitations at the last mile, required upgrades to hardware or software on the part of users, training or education of user communities, organizational planning delays, and contractual restrictions that might preclude the implementation of plans to take advantage of new infrastructure if agreements need to be renegotiated.
- Identification of the locations affected by most BTOP infrastructure projects is made more difficult due to distributed access to that infrastructure caused by the structure of the Internet itself. While projects have generalized areas that are targeted for improvement (generally identifiable at the county level) the effects of a large investment in middle mile infrastructure may not be felt equally throughout that county, and they might occur in other geographic regions depending on activities of last mile providers. Establishing the connection between middle mile infrastructure and users is difficult, as most broadband adopters do not know what middle mile infrastructure they are using, and last mile providers cannot be required to provide this information as part of our study design.

4.5 Matched Pairs Selection

An effective and well-established way to develop estimates of the effects of programs such as BTOP is the use of matched-pairs analysis. Geographies are selected that, to the extent possible, are similar to those where BTOP funding was received. These geographies are used as a "control" group, while those areas receiving BTOP funding are considered to be the "treatment" group. Comparison of the "treatment" and "control" groups will allow the evaluation study team to take factors into account that might be affecting broadband availability, access or adoption. In particular, we would be able to net out effects of national economic conditions, such as recovery from recession, which would presumably affect both "control" and "treatment" areas.

Matched-pairs analysis will be incorporated into our analysis of the intermediate-term impacts of infrastructure projects. Matching will take place at the county level, both because this provides a good set of matching parameters and can easily be incorporated into our analysis methodology. The key to this sort of estimator is the availability of geographic areas that are comparable, or nearly so, to areas receiving broadband investments, but which do not receive such investments themselves. Especially important is the identification of the fundamental economic base of each.

Following Gillett et al. we will use Stata's NNMATCH function to develop the control group for the matched-pairs samples of counties.⁵ This function finds a control area (or groups) for each geographic area receiving BTOP funding. The procedure uses nearest-neighbor matching among the independent variables to identify similar observations. These independent variables could include the level of broadband service available at some time in the recent past. There are various options available using this procedure and an investigation of the data and the quality of the matches will form part of our control group selection methodology.

 $http://web.si.umich.edu/tprc/papers/2005/475/TPRC2005_Gillett\%20Lehr\%20Sirbu\%20Osorio\%20submitted.pdf.$



⁵ Gillett, Sharron, William Lehr, Carlos Osorio, and Marvin Sirbu. 2006. "Measuring Broadband's Economic Impact." United States Department of Commerce, Economic Development Administration. http://wab.ai.umiab.adu/tara/apaper/2005/475/TDBC2005. Cillett// 201 abr// 2020aria% 2020ari

Section 5. Statistical Estimation Methodology

5.1 Introduction

This section describes our statistical methodology for estimating short-, intermediate-, and longterm impacts of BTOP grants. This analysis will produce interim results during the course of the study as well as material to be incorporated into the Final Report. The statistical estimation methodology presented here is primarily designed to characterize the economic impacts of BTOP infrastructure grants, although some of the social impacts will also be considered. After a short discussion of previous statistical work in this area, we present the methodologies we will use for short-, intermediate-, and long-term statistical analysis. The final portion of this section discusses matched pairs and sensitivity analysis.

5.2 Purpose of the Statistical Estimation Methodology

We will use the statistical estimation methodology described in this section to quantitatively estimate the economic and social impacts of BTOP grants, especially infrastructure programs. The impacts of infrastructure programs are most amenable to quantitative analysis, and the methodology provided here is targeted toward the evaluation of these programs. Infrastructure projects have also received the majority of BTOP funding. In some cases PCC and SBA grants may have measurable quantitative impacts, especially in the short-term. We anticipate that the case study methodology discussed in the following section will provide a substantial portion of the social and economic impact of these programs. However, the quantitative analysis presented here might supplement these case studies, especially for short-term effects of investments in PCCs.

Within each outcome area, various study criteria will include measurement of the impact of NTIA's implementation of BTOP on individuals, households, anchor institutions, and/or other community level units depending on the type of grant and specific grant activities undertaken. Again, some criteria will have an impact at multiple levels. Our methodology will be designed to report impacts at the levels deemed to be the most relevant for a given criterion.

Our study will examine both broadband availability and broadband adoption when considering how BTOP grants influence broadband use. While these factors have been conflated in many studies due to data limitations, the causes of each are different. The following definitions will be used in the study:

- Availability: According to the National Broadband Map (NBM), broadband service might be available to an area, but only if construction or other work is completed first. The NBM is not intended to include broadband service that is not available to a customer within a typical service interval (7 to 10 business days) and/or without an extraordinary commitment of resources.
- Access: Dependent on availability, access determines if the household / business have the tools/knowledge required to take advantage of the available technology.
- Adoption: The integration of these technologies into processes / daily life. "How do you behave differently now that you have broadband" / "How do you do business differently now that you have broadband"?

Broadband availability is more strongly driven by engineering limitations inherent in broadband technology and economic limitations on the types of projects that are profitable to undertake. Adoption is determined, in part, by the characteristics of the population that could be served and the area in which service is provided.



5.3 Estimating Short-Term Economic Impacts

When grants are distributed, they will have an immediate effect on economic activity within a particular region. In particular, construction activity, which is most closely related to CCI grants, will yield jobs in the short-term due to completion of the activities required by the project itself. To a certain extent, the PCC and SBA projects might exhibit similar results, but on a smaller scale.

We will define short-term economic impacts as those that occur due to the BTOP grant before other factors in the geography of interest have yet adjusted to the BTOP grant expenditure. Long-term economic impacts are observed when all factors may be adjusted. In the interim, some factors might remain fixed while others adjust, leading to intermediate-term impacts.

There exists a small body of well-developed literature surrounding short-term economic effects of broadband CCI grants. It primarily focuses on the use of input-output modeling to estimate the direct, indirect, and induced effects of broadband spending. Generally, input-output analysis examines how different industry sectors affect national or regional economies – that is, how a change in one industry can induce changes in other industries, as an output for one can become an input for another. Input-output analysis generates multipliers to account for these ripple effects. In order to facilitate comparison and aggregation of results, we will use national averages for input-output multipliers.

We will be using the IMPLAN software package to develop input-output estimates of the likely employment effects of BTOP grants in affected areas. Using data on total expenditures by industry sector, IMPLAN can derive the direct, indirect, and induced effects of work completed for BTOP grants on their respective regional economies. Specifically, IMPLAN can estimate industry-sector activity for final demand, final payments, industry output, and employment for each county in the U.S. along with state and national totals.

The basis for our analysis will be PPR data, which we will map to IMPLAN categories for the purpose of determining how various IMPLAN sectors would be affected by BTOP grant expenditures. Then, there are two possible approaches to mapping PAM expenditure data to their respective industries. First, we will assign each Cost Classification to a three-digit NAICS code. We will then follow one of two routes: 1) import the data to IMPLAN with an aggregation step which makes the three-digit NAICS code data IMPLAN-compatible; 2) map the three-digit NAICS codes to their respective IMPLAN 440 industry codes, which properly formats the data for IMPLAN software without an aggregation step. The mapping of NAICS codes to IMPLAN industry codes is show in Table 4 below.

Since PPR reports are due quarterly, it will be possible to update these analyses on a quarterly basis, allowing the evaluation study team to define short-, intermediate-, and long-term impacts. In addition to the methodologies used in the literature, BTOP also has reporting requirements that will produce more detailed information on the short-term effects of the project than would normally be obtainable from such an effort. In particular, Recovery Act reporting requirements include an estimate of the number of jobs that will be created as a result of grant expenditures. These numbers are publicly available and will be used to estimate the short-term job creation effects of BTOP projects. We will make comparisons between the IMPLAN results and the results from Recovery Act reporting.



Table 4: Mapping Cost Classifications to Three-Digit NAICS to IMPLAN 440

PAM Cost Classification	2007 NAICS	NAICS Title	2007 NAICS3	IMPLAN 440
ArchFee_Amt	5413	Architectural, Engineering, and Related Services	541	369
Benefits_Amt	523	Securities, Commodity Contracts, and Other Financial Investments and Related Activities	523	356
Constructn_Amt	23*	New Construction	23*	36
Contract_Amt	5411	Legal Services	541	367
Demolitn_Amt	238910	Building demolition, Land clearing, Trenching	238	39
Equipmnt_Amt	5324	Commercial and Industrial Machinery and Equipment Rental and Leasing	532	365
IndirCh_Conting_Amt	523	Securities, Commodity Contracts, and Other Financial Investments and Related Activities	523	356
LandApprsl_Amt	531320	Appraisal services, real estate	531	360
LegalExp_Amt	5411	Legal Services	541	367
OtherArchFee_Amt	5413	Architectural, Engineering, and Related Services	541	369
OtherMisc_Amt	523	Securities, Commodity Contracts, and Other Financial Investments and Related Activities	523	356
Personnel_Amt	561110	Administrative Management Services	561	384
ProjInspFee_Amt	541350	Architectural, Engineering, and Related Services - Building Inspection Services	541	369
RelocExp_Amt	484	Truck Transportation	484	335
SiteWork_Amt	23*	New construction, Repair and maintenance	23*	39
Supplies_Amt	334210	Carrier equipment (i.e., analog, digital), telephone, manufacturing	334	237
Travel_Amt	5615	Travel Arrangement and Reservation Services	561	383

5.4 Estimating Intermediate-Term Impacts

By "intermediate-term" impacts, we mean an economic or social impact that will be observed during the study period within the geographic area designated for a particular grant. We distinguish between short-term effects, which are primarily due to expenditures on inputs, and intermediate-term effects, which are caused by the outputs of the projects, in this case increased infrastructure development.


A variety of econometric methodologies have been employed to estimate the effects of broadband availability and adoption on employment and other economic outcomes. For instance, Gillett et al. evaluate the effect of broadband investment on employment, wages, rents (a proxy for property values), and the number of business establishments in a region (a proxy for the number of firms in a region).⁶ These authors perform a ZIP Code level analysis on two matched cross sections, the first as of December 1999 and the second four years later. They track changes in economic indicators over time in order to identify differentiated patterns of change based on broadband availability while controlling for community-level factors such as income, education, and urban/rural status. Gillett et al. find the presence of mass-market broadband is a significant factor in explaining growth in employment, the number of businesses in a ZIP Code and the number of IT businesses in a ZIP Code. Wages were not found to be affected by the presence of mass-market broadband, but property values appeared to be increased by broadband availability.

In a similar study, Crandall et al. estimated the effects of broadband penetration on both output and employment, in aggregate, and by sector, using state-level FCC data.⁷ Economic performance captured by state-by-state estimates of output (GDP) data from the Bureau of Economic Analysis through 2005 and employment data from the Bureau of Labor Statistics through 2006. This study uses FCC data on broadband penetration for the lower 48 states over the 2003-2005 period, controlling for a variety of other factors that also could account for the growth in output and employment during this time. The authors find that nonfarm private employment and employment in several industries is positively associated with broadband use. For every one percentage point increase in broadband penetration in a state, employment is projected to increase by 0.2 to 0.3 percent per year.

Other authors have taken a more synthetic approach to the estimate of the effects of broadband technologies on specific areas. In their paper, "[t]he Economic Impact of Digital Exclusion," the Digital Impact Group and Econsult Corporation develop a taxonomy of potential deficits in social or economic outcomes and use studies of specific effects of broadband to estimate the effect on each component.⁸ The sum total of these effects is then computed to present an assessment of broadband's overall impact. This paper presents an annual estimate of \$55 billion for the cost of exclusion of 100 million individuals and 40 million households from broadband access in the United States.

This type of study has several benefits, including:

- The general-purpose nature of broadband technology may be captured by a sufficiently broad and diverse taxonomy of potential effects.
- Estimates of broadband technology in areas where the technology has not been widely deployed, or where broadband technology has been newly adapted for use may be included in the estimation methodology.

The drawbacks of this approach may include:

• Potential for double-counting or undercounting of the same benefit. As an example, education and economic opportunity are potentially overlapping categories with respect to vocational education. Since statistics on the impact of broadband on each of these components may be computed separately by different authors, some educational activities may be counted in both places.

⁸ Digital Impact Group and Econsult Corporation. 2010. "The Economic Impact of Digital Exclusion." Article published March 5. http://www.econsult.com/articles/030810_costofexclusion.pdf



⁶ Gillett, Sharron, William Lehr, Carlos Osorio, and Marvin Sirbu. 2006. "Measuring Broadband's Economic Impact." U.S. Department of Commerce, Economic Development Administration.

http://web.si.umich.edu/tprc/papers/2005/475/TPRC2005_Gillett%20Lehr%20Sirbu%20Osorio%20submitted.pdf.

⁷ Crandall, Robert, William Lehr, and Robert Litan. 2007. "The Effects of Broadband Deployment on Output and Employment: A Cross-sectional Analysis of U.S. Data." *The Brookings Institution: Issues in Economic Policy*, no. 6 (July). http://www.brookings.edu/views/papers/crandall/200706litan.pdf.

 Lack of clarity on definitions of access, availability and adoption, and the factors that influence each. That is, the character of each particular benefit will likely have an impact on the reason why broadband is not in wider use. For example, personal financial management, which requires lower speed than personal entertainment might be more influenced by access, while personal entertainment would be influenced by availability of higher speed connections.

Notwithstanding these potential drawbacks, we will follow a similar methodology to obtain estimates of the economic and social benefits of broadband use as part of our methodology.

5.4.1 Characteristics of BTOP Infrastructure Projects

The particular characteristics of BTOP Infrastructure projects must be carefully considered in the creation of the statistical estimation methodology. NTIA made the strategic decision to provide grant funding to "middle mile" projects. In contrast to "last mile" infrastructure, which serves households or businesses, middle mile projects provide service to last mile broadband service providers themselves. This has important implications for the measurement of the benefits of BTOP broadband infrastructure projects.

In the case of middle mile projects, the situation is more complex. A single middle mile project may serve multiple last mile providers. The middle mile grant recipient is unlikely to know the full extent of the area served by the last mile providers who use the middle mile network built or activated using BTOP funding. Furthermore, the last mile provider may have no relationship at all with NTIA, other than as a consumer of the services provided by a grant recipient. The delineation of the areas served by the last mile providers is also likely to be sensitive financial information. This raises the possibility that little information would be provided on the locations served by last mile providers using connections to middle mile BTOP projects.

Last mile projects in theory will have a defined service area that may be defined by the locations of connection points, and descriptions of distribution infrastructure, which is known to the last mile providers. For mixed middle-mile/last-mile projects some mixture of these situations is likely to obtain, with some last mile activity being known to the grant recipient, with other last mile activity being potentially unavailable. This will depend on the characteristics of the project and the relationships between the last and middle mile providers.

First, most studies of the impact of broadband availability and adoption have focused on whether broadband was available in a certain area and how economic or social measures changed after broadband came online in areas that had not previously had it. Implicit in this form of analysis is the assumption that the "last mile" is the goal of broadband policy. While this is true for some NTIA infrastructure grants, the majority of BTOP infrastructure projects and funding are addressed toward "middle mile" projects. Middle-mile projects were funded by NTIA in part because they have widely-distributed benefits. An example of this is OneCommunity Ohio, which has provided maps and other documentation of their project indicating the extent of the infrastructure improvements planned for that project. In order to provide more context for the discussion of our intermediate- and long-term impacts, we present some background information on the OneCommunity project below.

The OneCommunity Ohio infrastructure project is slated to build 900 miles of fiber, connect 2,000 miles of existing fiber, connect 20 counties and 800 anchor institution and provide users with broadband with speeds between 10 to 40 Gbps. The geographic area covered by OneCommunity is a 20-county area in NE Ohio plus Franklin County, representing ½ the State's population (5.4 million), over 200,000 businesses and nearly 9000 community anchor institutions. Figure 3 below shows before-and-after middle mile connectivity.



Figure 3: OneCommunity Ohio infrastructure Project



THE OHIO MIDDLE MILE CONSORTIUM

Analysis of infrastructure projects of this size and geographic scope presents several research challenges. First, while NTIA Broadband Mapping data allows for the identification of broadband service parameters down to the Census block level, the determination of whether a middle mile project has contributed to the provision of service to a particular Census block or other location depends on the identification of the last mile connections from the middle mile infrastructure to that location. These connections are most often made by entities other than NTIA and the middle mile grant recipient, which means information on specific connections is not generally available. Second, given that the particular geographic area served by a middle mile project cannot be exactly identified, the development of control variables, such as the level of infrastructure investment activity by non-recipient providers, is difficult or impossible to obtain.

5.5 Proposed Methodology

In order to overcome these hurdles, we have developed a multipart statistical estimation methodology. These steps are:



- 1. Establish the increased level of service provided by middle mile projects receiving BTOP funds. These are the only measures in our statistical estimation methodology that require obtaining additional information from BTOP grant recipients. While there are numerous measures of the capacity of broadband infrastructure, in general we are interested in the number of additional customers that could be served at particular speed levels. This points to the use of a measure like TB/month transmitted as a measure of capacity.
- 2. Identify the counties that are likely to be affected by each BTOP CCI grant. The primary sources for this identification will be publicly-released grant application data, press releases, and studies published by the grant recipients. For those CCI grants selected for analysis as part of the CCI case studies, we will confirm this information as part of the case study process. The goal of this step is not to make a definitive list of the areas affected by the grant, but to set some general areas where results would be expected, acknowledging that the final areas benefitting from the infrastructure are likely to differ from the county-level results.
- 3. Develop a list of census blocks in each identified county, and compare broadband service over time. The goal is to develop a measure of broadband development progress in each census block that incorporates information on available broadband speeds and adoption rates. This information is available from the NTIA National Broadband mapping data and is updated twice yearly.
- 4. Develop a list of anchor institutions in the identified counties and compare broadband service over time. These data should be available as part of the NTIA National Broadband Map. The goal for this step is similar to #3, above. Information on available broadband speeds and adoption rates will be captured in relevant statistics. These data are expected to be updated twice each year.
- 5. Assemble benchmark usage statistics for bandwidth use by households and anchor institutions. We will consult with industry experts and trade organizations as well as perform background research to develop expected up and download volumes for different types of users with different types of service. The goal of this step is to develop a methodology for estimating the amount of middle mile infrastructure required to serve a particular identified county and to obtain a measure of the increase (or decrease) in required bandwidth over time.
- 6. Based on the results of #5, we will estimate the volume of middle mile service required to support broadband use in areas that have been targeted for improvement by specific projects. This amount of transport capacity will be stated in the same terms as the amount of capacity provided by BTOP projects (i.e., TB/month).
- 7. We will develop an estimate of the fraction of additional required capacity provided by BTOP grant recipients. This is a simple ratio of the result obtained in #1 to the result obtained in #6. This ratio will be below to apportion likely benefits to BTOP grant recipients.
- 8. Based on the results observed in #3 and #4, above, we will identify potential uses of broadband technology in the various census blocks and at the observed anchor institutions in the affected counties. For instance, lower broadband speeds might be appropriate for streaming of recorded audio material, while streaming of video content or videoconferencing would require higher speeds. Anchor institutions might use broadband at lower speeds for transmission of documents and records, while higher speed connections might facilitate videoconferencing. The end result of this step will be a table of likely uses of the additional broadband capacity provided by the BTOP recipient.
- 9. Perform a demographic analysis on the areas experiencing increases in availability, adoption or speed, and present a summary of the characteristics of the beneficiaries of increased broadband service. Although it will not be possible to say specifically which households or individuals benefitted from the increased availability of infrastructure, the



characteristics of the population of benefit recipients will be assumed to be the same as those in the areas where broadband access and availability increased.

- 10. Measures of significant social outcome variables will be obtained in the affected areas targeted by infrastructure programs and these will be compared to national averages. The main idea of this step is not to make the claim that these indicators would be immediately affected by the broadband infrastructure project, but that the scope for possible improvements is bounded by the extent of the issues observed in the first place. In other words, areas with high crime rates have a greater capacity for improvement in public safety than areas with very low crime rates. Likewise, areas with poor indicators of public health or civic participation would be expected to improve more overall than areas with better indicators of these things, when the effects of broadband technology are measured by future researchers.
- 11. For those projects for which we are performing a case study, additional steps will be taken to integrate the results of both methodologies. We will first present the results of our study to grant recipient personnel, including the assumptions we have made about the areas served, bandwidth usage, and other factors. We will then explore the implications of the increased bandwidth provided by the project, with special attention to areas of focus specified by each infrastructure grant.

5.5.1 Strengths and Weaknesses of this Approach

As is the case with every methodology for impact estimation, the methodology we propose has relative strengths and weaknesses. One of the strengths of the methodology is that it allows for less-than-perfect identification of the specific geographic areas that are affected by the projects receiving BTOP grants, although some idea of where these projects are having their effects is required. Second, the incorporation of NTIA Broadband Mapping data provides much more specific measurements on the changes to broadband usage in potentially unserved or underserved areas than has been developed to date. Finally, the examination of anchor institutions provides insight into key participants in the community's use of broadband.

Drawbacks to this methodology include a lack of empirically-testable hypotheses, such as the extent to which broadband technology facilitates productivity growth. The absence of a mathematical formulation of the effects of broadband limits the use of the mathematical results for forecasting, although our methodology does produce ratio estimates that could be used for the same purpose. Finally, econometric estimation strategies have well-understood error bounds on the results they produce, while the estimates produced by the methodology we propose do not have the same statistical properties. We will, however, perform sensitivity analysis on key parameters to alleviate some of this concern.

5.6 Estimating Longer-Term Impacts

The effects of BTOP infrastructure investments are likely to be felt years in the future. However, this study will be completed before these effects have had a chance to completely play out. As a result, measuring longer-term impacts of BTOP investments will require forecasting of the likely changes in broadband over an extended period of time. We believe that an examination of the results of earlier phases of the project, case study results, and a critical analysis of other research in this area will yield insights as to prospective outcomes where grants were made. The following presents a description of the steps that we will take to estimate the longer-term impacts of BTOP infrastructure projects. Our longer-term impact analysis will be based on the results we obtain from the analysis of several key questions:



- 1. What proportion of the capacity provided by BTOP infrastructure grant recipients has been utilized and what remaining capacity is available for allocation to additional broadband service provision?
- 2. What is the level of broadband usage observed in the areas targeted by BTOP infrastructure grant recipients, and what is the potential for usage to be increased those areas?
- 3. What are the social and economic conditions in the areas targeted by BTOP infrastructure grant recipients, and what is the potential for improvement in those conditions due to broadband.

As is the case with all forecasting exercises, the results will be determined by the selection of assumptions used to create the forecast scenarios. We will perform sensitivity analysis on the forecast assumptions to illustrate the range of results that could be expected for longer-term impacts.

5.7 Sensitivity Analysis

We will undertake sensitivity analysis of our results in order to better understand how the underlying assumptions of our models affect the conclusions we draw. Our sensitivity analysis will consider the composition of the areas affected by the BTOP grants we intend to study. In the case that some of the areas affected by grants are unclear, we will examine different methodologies for designating affected census blocks or ZIP Codes. We will then perform the analysis using these different geographic designations to determine if uncertainty about the boundaries of project grants causes changes in the estimated results.

The sensitivity analysis will be incorporated into our forecast analysis of the likely future effects of BTOP programs. These forecasts will provide a range around the expected future outcomes, based in part on the results of the sensitivity analysis. Additional factors might be included, such as expected changes to technologies or infrastructure not captured in the statistical models.



Section 6. Case Studies

6.1 Introduction

This section presents our case study methodology. We describe the purpose of case studies, our methodology for how case studies will be conducted, data we will gather to support case study site visits, and the outputs of our analysis – the case study reports.

6.2 Purpose of Case Studies

We will perform case studies of selected CCI, PCC, and SBA grants in order to supplement and, in some areas of inquiry, to support the quantitative analysis. The case studies will provide a window into the initial impacts of BTOP awards and provide ongoing snapshots of the Program's economic and social impacts, allowing for review and possible adjustments for the longer-term, in-depth longitudinal study that will be a centerpiece of the Final Report.

The case studies will identify how the grantee maximized the impact of the BTOP investment; identify successful techniques, tools, materials and strategies used to implement the project; identify any best practices; and gather evidence from grantees, anchor institutions, and various publicly available data as to the impacts of the project in the community. The case studies will provide qualitative information; and, depending on the individual evaluation efforts of selected grantees, perhaps some quantitative data related to economic and social impacts of the activities resulting from the BTOP funded grant.

The case study effort will have three components and result in two sets of interim impact reports. The first set of interim reports will be delivered in April 2012. Due to the nature of BTOP, and the CCI projects in particular, this first set of case studies will focus exclusively on 15 selected PCC and SBA projects because CCI projects will not be far enough along in their development to be able to measure impacts. In September 2013, a second set of interim impact reports will be delivered. These reports will include an update on the same 15 PCC and SBA projects analyzed in the first set of case studies and include a set of case studies on 12 selected CCI grants as well. This methodology allows for both a cross-sectional analysis to measure impacts at a point in time and information to feed the longitudinal analysis of the impacts of the BTOP PCC and SBA grants over time.

6.3 Methodology

The development of the case studies will include a number of steps including various communications with the selected grantees, a number of data collection efforts, and development of case study reports. Our approach to each of these steps is described below.

6.3.1 Grantee Notification and Communication

The point of contact of record for each grant will receive initial notification of their selection as a case study grant through their assigned Federal Program Officer. The evaluation study team will then contact each grantee for an initial introduction and description of the effort.

This will be the first of many interactions between the evaluation study team and the grantee over the course of the case study period. Additional communications will occur to establish case study visit logistics and to gather information related to the project. Specifically, the evaluation team will



rely on the grantee to provide any background material on the grant not available from NTIA. We will also rely on the grantees to provide recommendations on the specific location(s) of their various anchor institutions or service delivery locations that we should visit and to provide contact information for the individuals at those locations. Communication with those individuals will be used to plan and conduct case study visits and to collect any available data specific to those locations.

6.3.2 Data Collection

Various data will support the development of the case studies. This includes community level public use economic, demographic and broadband specific data; information specific to each grant as collected from their grant applications; quarterly and annual performance progress reports; and more detailed grant-specific information as collected through visits to anchor institutions and service delivery locations. This information will provide generally for analysis on potential changes in economic conditions, provision of healthcare and/or public safety services, educational opportunities, quality of life or standard of living, and other social goods resulting from BTOP investments. In addition, to the extent possible and where data are available, the case studies might explore important economic concepts that may otherwise be difficult to quantify, such as innovation and entrepreneurship. Though the thrust of the case studies is not to evaluate implementation of the projects against their originally stated goals (but rather to document impacts), project implementation issues such as the grantee's organizational type, business model, and staffing will be explored to the extent they appear to influence the ability of the grant to have an impact on its community and in order to contributed to the discussion of best practices and lessons learned.

6.3.2.1 Economic, Social, Demographic, and Broadband Specific Statistics

Various economic, social, and demographic statistics will be collected through public use data and grant applications in order to characterize the area impacted by the grant. Close review of the grant application and conversations with the grantee will be utilized in order to determine the impacted area or "the community" to be considered for each grant. Economic/demographic statistics collected and described (when available) will include, but not be limited to:

- Population, by age, gender and race,
- Personal income,
- Housing values,
- Educational attainment and graduation rates,
- Local economic environment characteristics including economic development efforts,
- Employment,
- Voter registration and participation,
- Volunteerism,
- Mortality rates,
- Suicide rates,
- Membership in civic associations,
- Homelessness,
- Telecommuting rates,
- Establishment of home businesses



- Communications via email, chat, webcam, etc.
- Home schooling rates, and
- Crime statistics.

The impacted area will also be characterized in terms of its baseline or existing level and type of broadband service. Broadband related statistics collected, if available, will include, but not be limited to:

- List of existing broadband providers, including advertised speeds and rates,
- Market share of existing broadband providers, and
- Examples of broadband use by local industries.

For PCC and SBA grants, these data will be collected at the time of the first round of case studies (2011) to represent the pre-BTOP situation. The same data will be collected again at the time of the second round of case studies (2013) and represent the post-BTOP situation. For CCI grants, the case study report developed towards the end of 2013 will include data from 2010 or 2011 as appropriate and available to represent the before BTOP picture, and data from the time of the case study to represent the post-BTOP picture.

Under the NTIA's State Broadband Data Development Program (SBDD), most states have formed a broadband authority. These entities have engaged to varying degrees in local adoption and impact studies. On a case by case basis, these existing studies will also be used as a data source for the case studies. To the extent possible, information from last mile providers and local businesses regarding their degree of adoption of services related to BTOP middle miles CCI projects will be collected as well. These data may come from state broadband authority studies. If not, conversations with grantees and/or the NTIA Broadband mapping data will be used to identify last mile providers or businesses with whom to speak.

In addition, there are on occasion outside influences such as changes in the state or local level political, legal, or regulatory climate that can affect the ability of BTOP grants to have an impact in a certain area. Therefore, efforts will also be made to investigate and report on state or local level broadband or other relevant public infrastructure-related conditions, laws, statutes, or other elements of the local political climate that could have an influence on the ability of the selected grant to achieve its stated objectives.

6.3.2.2 Grantee Quarterly Performance Progress Reports

Grantees are required to submit Quarterly PPRs to NTIA describing their grant funded activities. These reports provide quantitative and qualitative descriptions of project status and specific grant funded activities. The portion of the PPRs that includes detailed information on activities specific to each grant type will be useful in identifying potential project impacts specific to a particular location that can be verified and validated through the case study effort. This information will also be useful in identifying project specific areas of inquiry to pursue further with individual anchor institution contacts during the case study site visits.

Specific to SBA projects, the PPR data most relevant to the case study effort includes:

- Name, location, and description of the SBA activity(ies) undertaken in the current quarter,
- Size of target audience,
- Actual number of participants,
- New subscribers (households and businesses),



- Number of households and businesses receiving discounted broadband service as a result of BTOP,
- General description of activities planned for the upcoming quarter, and
- Description of method used to define attributed subscribership (i.e. adoption).

For PCC projects, the PPR data most relevant to the case study effort includes:

- Description of significant project accomplishments,
- New workstations installed and available to the public,
- Average users per week,
- Upgraded broadband connectivity at PCC,
- Established broadband wireless connectivity at PCC,
- Number of additional hours per week existing PCC is open to public as result of BTOP funds, and
- Name, length, number of participants and number of training hours for all training programs.

The data relevant to the case study effort from the CCI PPRs includes:

- New network miles deployed,
- New network miles leased,
- Existing network miles upgraded,
- Existing network miles leased,
- Number of miles of new fiber (aerial or underground),
- Number of new wireless links,
- Number of new towers,
- Number of interconnection points,
- Number of signed agreements with broadband wholesalers or last mile providers,
- Number of agreements currently being negotiated with broadband wholesalers or last mile providers,
- Average term of signed agreements,
- Numbers of businesses, households, anchor institutions, and last mile providers receiving new and/or improved access, and
- Name, service area and type of anchor institutions connected and description of how the anchor is using the BTOP funded infrastructure.

It is these last two pieces of information from the PPRs that will be of particular importance to the CCI case study activities because they will help to define the case study site visit specific locations and areas of inquiry.

6.3.2.3 Service Delivery Locations/Anchor Institution Visits

Each case study will include an in-person visit to targeted anchor institutions (CCI grants) or service delivery locations (PCC/SBA grants.) This visit will include a face-to-face visit with the



grantee as well, if possible, depending on their location relative to the selected anchor/other location(s). Case study visits will normally consist of a two-day trip. An interview guide and data collection checklist targeted to the particular goals, intended outputs/outcomes, types of services provided, community conditions, and anchor institution/service location types of each grant will be developed to provide the structure for each case study visit. These questions and lists will be developed in close coordination with NTIA and will consider data collected from the initial contacts with the grantees, our review of all available grant award and post-award documents and the economic, demographic, and broadband specific statistics compiled for each grant location. The guides and checklists will be broken out into sections to identify appropriate areas of inquiry specific to grantees, anchor institution/service locations, local economic development professionals (where identified by grantees or anchor site contacts), last mile providers (for CCI grants), and individual users.⁹

Generally, the case study visit interview guide and data collection checklist will be designed to gather the following types of information:

- Validation of background information on grant purpose, scope, services, and goals
- Description of anchor institutions, if applicable
- Description of grantee type, business model, and staffing issues
- Validation and clarification of reported project activity outputs and outcomes
- An assessment of the broadband adoption readiness of anchor institutions/service locations and the community
- Current/baseline broadband (and/or Internet) operations/uses at targeted anchor institutions/service locations
- Planned operational changes and uses of broadband (and/or Internet) as result of BTOP grant at targeted anchor institutions/service locations
- Observed changes resulting from BTOP funded activities to-date (if any) at targeted anchor institutions/service locations
- Observed economic and/or social outcomes at the anchor institution/service location, community, and/or individual level (if any), and
- An assessment of the sustainability of project activities beyond the term of the BTOP grant, if applicable

The collection of data on impacts at the individual level will depend largely on the degree to which the individual grantees are collecting data and/or conducting evaluation efforts related to their BTOP funded activities. We will also work with the anchor site/service location contacts to identify, where possible, opportunities to speak to individual users.

A case study visit travel and logistics plan along with the targeted, location specific interview guides and data collection checklists will be provided to NTIA for review prior to the start of the case study period. Given the diversity of projects selected for case studies, these guides will be highly customized and specific to each grant.

6.3.3 Case Study Report

Each case study will result in an individual case study report. A case study report template will be developed, although each report will differ based on details of the subject grant. The first round of PCC and SBA grant case studies will result in a set of 15 interim impact reports. As discussed

⁹ Case study visits will be conducted in compliance with 15 CFR Part 27 Protection of Human Subjects.



above, these reports will provide detailed information on the baselines against which future analysis will be compared. The second round of case studies will result in another set of interim impact reports – updates of the 15 PCC and SBA case studies previously developed and 12 CCI grant case studies. At contract closing, the case study reports, along with all other quantitative and qualitative impact assessment efforts detailed throughout this document, will be combined to develop the final report assessing the economic and social impacts of the BTOP grants.

The unit of analysis for each case study will be unique and will depend on the definition of the impacted area and the granularity of the economic, demographic, and broadband specific data available. Some grants, such as the City of Chicago, include activities that are community specific. Most, however, will likely have to be defined at the county level. Most of the public use data will likely only be available at the county level, but some localities may have data available by community. Further, in terms of the discussion of findings, i.e. outcomes or impacts, the affected community may have to be defined in something other than geographic terms. For example, impacts may be discussed in terms of an industry or other topical area.

Each case study report will generally include the following sections:

- Description of grant. This will include a discussion of the grant's purpose, area of impact, and services offered,
- Characterization of the impacted area/community. This will include presentation and discussion of various economic, demographic, and Internet/broadband specific statistics of the community(ies) affected by the BTOP grant, including local economic development efforts,
- Summary of data provided by grantees through their PPRs, and
- Summary of anchor/service location visits and findings. This will include a description of anchor institutions/service locations and overall findings in terms of:
 - Broadband adoption readiness (primarily for infrastructure projects),
 - Current/baseline broadband (or Internet) operations/uses (primarily for infrastructure projects),
 - Planned operational changes and uses of Internet as result BTOP grant (primarily for infrastructure projects),
 - Observed changes resulting from BTOP funded activities to-date (if any),
 - Observed economic and/or social outcomes at the anchor institution/service location, community, and/or individual level (if any),
 - Sustainability of project activities, as determined through review of grantee business plans,
 - Best practices (if any identified), and
 - Lessons learned.

In addition, in order to make a comparison to other communities that did not receive BTOP funding, each case study report will include, to the extent possible, a review of NTIA's BTOP application database to identify if there were any other applications in a similar geography (based on the economic and demographic statistics collected as explained in section 6.3.2.1). If so, we will inquire if any of those initiatives were undertaken, in whole or in part, without BTOP funding and include a description of that effort in the case study report. As grantees complete and submit additional PPRs, the evaluation study team will validate the geographic locations of impacted communities and anchor institutions providing the evaluation study team the ability to associate grantee reported data with BTOP impact areas. By definition, we will be able to identify the communities and anchor institutions that are outside the BTOP impact areas. Consequently, we can include in the second



set of interim case study reports statistics on the economic conditions of areas that were similar at the inception of BTOP, but not directly associated with BTOP initiatives and funding. This comparison will be included in the second round of PCC/SBA case study reports.



Appendix A. BTOP Overview

The Recovery Act provided NTIA with \$4.7 billion to support the deployment of broadband infrastructure, enhance and expand public computer centers, encourage sustainable adoption of broadband service, and develop and maintain a nationwide public map of broadband service capability and availability. The Recovery Act instructed NTIA to implement BTOP to promote five core purposes:

- To provide access to broadband service to consumers residing in unserved areas of the country;
- To provide improved access to broadband service to consumers residing in underserved areas of the country;
- To provide broadband education, awareness, training, access, equipment, and support to:
 - Schools, libraries, medical and healthcare providers, community colleges and other institutions of higher learning, and other community support organizations;
 - Organizations and agencies that provide outreach, access, equipment, and support services to facilitate greater use of broadband services by vulnerable populations (e.g., low-income, unemployed, aged); or
 - Job-creating strategic facilities located in state- or federally-designated economic development zones.
- To improve access to, and use of, broadband service by public safety agencies; and
- To stimulate the demand for broadband, economic growth, and job creation.

In facilitating the expansion of broadband infrastructure and adoption of broadband services, NTIA aims to advance the objectives of the Recovery Act to spur job creation and stimulate economic growth and opportunity. Additionally, NTIA funded projects with the expectation that they will provide long-term economic and social benefits, including improvements to healthcare delivery, education, innovation, and the nation's global competitive position. NTIA administers BTOP within three project categories:

- Comprehensive Community Infrastructure (CCI): Projects to deploy new or improved broadband Internet facilities (e.g., laying new fiber-optic cables, upgrading wireless towers) to connect households, businesses, and Community Anchor Institutions (CAI) such as schools, libraries, hospitals, and public safety facilities. CCI projects funded by BTOP are predominantly middle mile projects, although a small number of last-mile projects were awarded.
- Public Computer Centers (PCC): Projects to establish new public computer facilities or upgrade existing facilities that provide broadband access to the general public or to specific vulnerable populations, such as low-income individuals, the unemployed, seniors, children, minorities, and people with disabilities.
- Sustainable Broadband Adoption (SBA): Projects to provide a focus on increasing broadband Internet usage and adoption, including projects among vulnerable populations where broadband technology traditionally has been underutilized. Many projects include digital literacy training and outreach campaigns to increase the relevance of broadband in the public's everyday lives.

In essence, the purposes described in the two Notice of Funds Availability (NOFA) indicate that BTOP grants are targeted to:



- Provide new or improved broadband access to unserved or underserved areas;
- Affect positive change in the areas of education, healthcare, employment, and public safety through access to various broadband related services and equipment;
- Increase the usage of broadband related services and equipment by vulnerable populations; and/or
- Stimulate overall demand for broadband and economic growth, in particular job creation.

PCC and SBA grants serve a different purpose than CCI grants, and in some cases intended to be complementary to them. While CCI grants address issues of broadband availability, PCC and SBA grants are more focused on broadband access and adoption. Availability, access, and adoption are required for broadband use.

NTIA awarded BTOP grants in two funding rounds. In both funding rounds, NTIA received more than 2,800 applications requesting in excess of \$36 billion. This BTOP portfolio of projects initially included:

- 123 infrastructure projects totaling \$3.5 billion in Federal grant funds to construct broadband networks;
- 66 Public Computer Center (PCC) projects totaling \$201 million in Federal grant funds to provide access to broadband, computer equipment, computer training, job training, and educational resources to the public and vulnerable populations; and
- 44 Sustainable Broadband Adoption (SBA) projects totaling nearly \$251 million in Federal grant funds to support innovative projects that promote broadband adoption, especially among vulnerable population groups where broadband technology traditionally has been underutilized.¹⁰

¹⁰ National Telecommunications and Information Administration. 2011. "Broadband Technology Opportunities Program Quarterly Status Report." Last modified December 22. http://www.ntia.doc.gov/files/ntia/publications/btop-quarterly-congressional-reportdec-2011.pdf.



Appendix B. Literature Review

This section describes selected articles drawn from the literature, in addition to policy responses that have resulted from them. Given the truly enormous volume of research that has been done in the area of the role of broadband on the economy and society, there is no way to present a concise summary of every paper and conclusion. Instead, we seek to provide the reader with a summary of work in this area that is most significant in relation to this study of the economic and social impacts of BTOP grants.

The literature on government-funded plans to stimulate economic activity is distinguished by both its sheer size and the number of issues that have been raised as part of the decade-long debate into what broadband strategies can be expected to achieve on both a local and an economy-wide level. Much of this research has been focused on the effects broadband has on incentives for firms and individuals to relocate to (or remain in) particular areas. Overall, the literature presents few examples of crystal clear conclusions or hard-and-fast rules that may be used to guide the development of policy.

B.1 Information Technology, Information and Communication Technology, and Broadband Impacts on the Economy

The Bureau of Economic Analysis estimates that for each \$1 invested in broadband, the economy benefits nearly \$3.¹¹ Gillett et al. find that after controlling for community-level factors known to influence broadband availability and economic activity between 1998 and 2002, communities in which mass-market broadband was available by December 1999 experienced more rapid growth in (1) employment, (2) the number of businesses overall, and (3) businesses in IT-intensive sectors. In addition, the effect of broadband availability by 1999 can be observed in higher market rates for rental housing in 2000. These impacts will be revealed after a certain amount of "time lag," as noted by Gillett et al., because broadband has to be 1) available, 2) adopted, and then 3) used in order to enhance productivity and economic activity.¹² Research by Atkinson, Castro, and Ezell tie these ideas together, stating that broadband (a necessary foundation for the adoption of information communication technology (ICT)) "encourages upstream investment in industries creating new and innovative applications and services such as telemedicine, Internet search, ecommerce, online education (distance learning), and social networking."¹³

B.1.1 Community Impacts

The benefits of broadband have been found to differ in urban versus rural communities. Kolko reports that the benefits realized by the introduction of broadband in rural communities will be greater when compared to urban communities.¹⁴ Prior research by the Columbia Telecommunications Corporation has quantified that rural small businesses get less service when compared to metro small businesses, holding prices constant; and rural small businesses paid

¹⁴ Kolko, Jed. 2010. "Does Broadband Boost Local Economic Development?" *Public Policy Institute of California*. Article published January. http://www.ppic.org/content/pubs/report/R_110JKR.pdf.



¹¹ Ford, G.S., and T.M. Koutsky. 2005. "Broadband and Economic Development: a Municipal Case Study from Florida." *Applied Economic Studies* 1. http://www.designnine.com/library/docs/other_papers/BroadbandFactFiction.pdf.

¹² Gillett, Sharron, William Lehr, Carlos Osorio, and Martin A. Sirbu. 2006. "Measuring Broadband's Economic Impact." United States Department of Commerce, Economic Development Administration.

http://web.si.umich.edu/tprc/papers/2005/475/TPRC2005_Gillett%20Lehr%20Sirbu%20Osorio%20submitted.pdf.

¹³ Atkinson, Robert D., Daniel Castro, and Stephen Ezell. 2009. "The Digital Road to Recovery: A Stimulus Plan to Create Jobs, Boost Productivity and Revitalize America." *Information Technology and Innovation Foundation*, January 7. http://www.itif.org/files/roadtorecovery.pdf.

higher prices than metro small businesses, holding services constant.¹⁵ Kolko reports a positive relationship between broadband expansion and economic growth, but that relationship is stronger in areas with lower population densities.¹⁶ This is consistent with the theory that smaller or more isolated areas may benefit more from high-speed connections, giving businesses in these areas access to larger markets. However, even for most high density areas, the relationship between broadband and growth remains positive on balance.¹⁷ The positive relationship is especially large for utilities; information; finance and insurance; professional, scientific, and technical services; management of companies and enterprises; and administrative and business support services. The relationship in these sectors is much larger than the relationship for overall employment.¹⁸

Ford and Koutsky offer a real-world example of this magnified "rural" impact. In 2001, Lake County – a small county in central Florida – began offering private businesses and municipal institutions access to one of Florida's most extensive, municipally-owned broadband networks, including fiber optic connections to hospitals, doctor offices, private businesses, and 44 schools. Their econometric model showed that Lake County has experienced approximately 100 percent greater growth in economic activity relative to comparable Florida counties since making its municipal broadband network generally available to businesses and municipal institutions in the county.¹⁹

B.1.2 Employment Impacts

Most economic impact studies will include an analysis of job creation, and our study will be no different. Employment impacts of various kinds have been discussed in the literature, including job creation, job search, telecommuting, and job displacement.

Broadband expansion efforts create jobs both directly and indirectly. Infrastructure investments such as broadband infrastructure create direct, indirect and induced changes to employment. Direct jobs are those created specifically by spending on the project itself. Indirect jobs are those created by newly employed workers spending their earnings.²⁰ In addition to these employment effects, broadband will also cause changes to the structure of the economy in general. All public infrastructure projects can be expected to induce some structural change, but the introduction of a general-purpose technology such as broadband increases the possibility of new forms of economic activity in a way that typical infrastructure projects do not. A Darby, Fuhr, and Pociask study calculated the sum of all of the effects of broadband, including jobs lost because of capital for labor substitution, productivity improvements, and from possible outsourcing. Even with these negative impacts, the number of jobs created was shown to be substantial.²¹ Darby, Fuhr, and Pociask also estimate that the wide-spread availability of broadband can add over 1 percent to the employment growth rate in a typical community.²² These changes are attributed to network effects.

"A network effect is the effect that one user of a good or service has on the value of that product to other users. The value of the network increases logarithmically with each new user added to the

http://archive.sba.gov/advo/research/rs373tot.pdf.

²² Ibid.



¹⁵ Columbia Telecommunications Corporation. 2010. "The Impact of Broadband Speed and Price on Small Business." Report for SBA under contract number SBAHQ-09-C-0050. Report released November. http://crebing.edu.gov/adu.gov/adu.gov/adu/gov/adu

¹⁶ Kolko, Jed. 2010. "Does Broadband Boost Local Economic Development?" *Public Policy Institute of California*. Article published January. http://www.ppic.org/content/pubs/report/R_110JKR.pdf.

¹⁷ *Ibid.*

¹⁸ *Ibid.*

¹⁹ Ford, G.S., and T.M. Koutsky. 2005. "Broadband and Economic Development: a Municipal Case Study from Florida." *Applied Economic Studies* 1. http://www.designnine.com/library/docs/other_papers/BroadbandFactFiction.pdf.

²⁰ Atkinson, Robert D., Daniel Castro, and Stephen Ezell. 2009. "The Digital Road to Recovery: A Stimulus Plan to Create Jobs, Boost Productivity and Revitalize America." *Information Technology and Innovation Foundation*, January 7. http://www.itif.org/files/roadtorecovery.pdf.

²¹ Darby, Larry F., Jospeh P. Fuhr Jr., and Stephen B. Pociask. 2010. "The Internet Ecosystem: Employment Impacts of National Broadband Policy". *The American Consumer Institute*. http://www.theamericanconsumer.org/wp-content/uploads/2010/01/acijobs-study-final1.pdf.

network. The classic example is the telephone; the more people own telephones, the more valuable the telephone is to each owner."²³

This network effect leads to new consumer and business behaviors, new functionalities, and new downstream industries enabled by the ICT infrastructure. This possibility arises because digital infrastructure acts as a platform that serves as the foundation for a multitude of innovative products and services."²⁴ Note that not all of the changes caused by network effects create jobs. It is possible that outsourcing or increases in productivity will cause decreases in employment. As a result, the sum of the effects of broadband investment on employment can be positive or negative.

Relating this to broadband in particular, the short-term job creation can be as simple as the jobs created to develop and deploy the infrastructure necessary for broadband technology. The labor intensive nature of broadband deployment ensures that the construction jobs created are significant and, despite the highly technological nature of the ultimate product, broadband is to be seen as being similarly economically meaningful as conventional infrastructure investments such as roads and bridges.²⁵

Long-term jobs are created or enhanced as a result of the availability of that broadband technology. Having this leading-edge technology, buyers can help IT companies gain a competitive advantage and boost IT jobs.²⁶ Atkinson, Castro, and Ezell go on to explain that the long-term job growth results in enhancing America's competitiveness and expand to higher value-added U.S. jobs.²⁷ This is quantified by Fuhr and Pociask, who found that for every worker employed in manufacturing and constructing a broadband network, 4.1 additional downstream jobs are created elsewhere in the economy.²⁸

Not only do IT infrastructure projects create more jobs than traditional infrastructure investments, in part because of the network multiplier, they also might also create more high-skilled, high-paying jobs.²⁹ Jobs involved in the building and expansion of broadband networks typically pay about 42 percent more than the average for manufacturing jobs.³⁰ Crandall et al. found that for every one percentage point increase in broadband penetration in a state, employment is projected to increase by 0.2 to 0.3 percent per year.³¹

The development of improved broadband infrastructure might also improve job matching. Atkinson et al. state that Internet-based job sites enable superior matching between employers and employees, making the process cheaper and faster and providing both employers and employees with more information on which to base their decisions.³² Improved job matches are beneficial to the employee and to the employer. The lower cost associated with Internet-based applicant

³² Atkinson, Robert D., Stephen Ezell, Scott M. Andes, Daniel Castro, and Richard Bennett. 2010. "The Internet Economy 25 Years After .com." *Information Technology and Innovation Foundation*. http://www.itif.org/files/2010-25-years.pdf.



²³ Atkinson, Robert D., Daniel Castro, and Stephen Ezell. 2009. "The Digital Road to Recovery: A Stimulus Plan to Create Jobs, Boost Productivity and Revitalize America." *Information Technology and Innovation Foundation*, January 7. http://www.itif.org/files/roadtorecovery.pdf.

²⁴ Ibid.

²⁵ Katz, Raul L., Stephan Vaterlaus, Patrick Zenhäusern, and Stephan Suter. 2010. "The Impact of Broadband on Jobs and the German Economy." *Intereconomics*. http://www.teleadvs.com/images/Katz,_final.pdf

²⁶ Atkinson, Robert D. 2009. "The Role of Competition in a National Broadband Policy." *Journal on Telecommunications and High Technology Law* 7. http://www.itif.org/files/JTHTL.pdf.

²⁷ Atkinson, Robert D., Daniel Castro, and Stephen Ezell. 2009. "The Digital Road to Recovery: A Stimulus Plan to Create Jobs, Boost Productivity and Revitalize America." *Information Technology and Innovation Foundation*, January 7. http://www.itif.org/files/roadtorecovery.pdf.

²⁸ Fuhr, Joseph J. and Stephen B. Pociask. 2007. "Broadband Services: Economic and Environmental Benefits." *American Consumer Institute*. http://www.nederlandbreedbandland.nl/uploaded/FILES/Final_Green_Benefits.pdf.

²⁹ Atkinson, Robert D., Daniel Castro, and Stephen Ezell. 2009. "The Digital Road to Recovery: A Stimulus Plan to Create Jobs, Boost Productivity and Revitalize America." *Information Technology and Innovation Foundation*, January 7. http://www.itif.org/files/roadtorecovery.pdf.

³⁰ Ibid.

³¹ Crandall, Robert, William Lehr and Robert Litan. 2007. "The Effects of Broadband Deployment on Output and Employment: A Cross-sectional Analysis of U.S. Data." *The Brookings Institution: Issues in Economic Policy*, no. 6 (July). http://www.brookings.edu/views/papers/crandall/200706litan.pdf.

searches as a result of broadband means that higher quality matches are possible (which raises labor productivity as well).³³

A third, and more recent, employment-related impact of ICT and broadband deployment and expansion is the practice of telecommuting. Benefits of telecommuting impact both the business and the employee. As a result of telecommuting, firms will need less equipment, office space, parking spaces, office equipment, supplies, and other amenities.³⁴ Atkinson finds that broadband facilitates work from home. Telecommuting frees employees from, on average, an hour of commuting each day.³⁵ This translates into greater output (benefit to the employer) and less time and money spent commuting (benefit to the employee).³⁶ Lehr et al. note that these benefits are difficult to quantify, but do extend employment opportunities to candidates with disabilities.³⁷ This extension can reduce unemployment among disabled individuals.³⁸ Telecommuting allows employers to search in a larger geographic area to find employees with rare skill sets.

Broadband technology accompanied with telecommuting could also discourage some of the migration to offshore jobs and encouraging what is called "homeshoring."³⁹ Katz and Suter clarify that the displacement of employment ("offshoring" or "homeshoring") from one targeted area to another should not be viewed as incremental employment, but rather viewed as mutually beneficial employment, matching the right candidate/employee with the right position unencumbered by geographic boundaries.⁴⁰

B.1.3 Business and Productivity Impacts

An analysis of the relationship between broadband availability and productivity suggests that many commercial subscribers are currently realizing productivity gains because of their broadband access.⁴¹ LECG reports that almost half of the projected increase in U.S. productivity growth between 2001 and 2011 could be accounted for by firms adopting Internet business solutions. The increment to U.S. productivity due to broadband has been about 0.25 percentage points per year between 1999 and 2007. During this period, average annual productivity growth in the U.S. has been about 2.1 percent per year, implying that the "broadband effect" is one-eighth of all productivity growth.⁴² Additional research differentiates companies' adoption rates based on size. Large enterprises are three times more likely to have intranet, extranet, and mobile access to the Internet than small enterprises.⁴³ Jayakar, Schejter, and Taylor go on to point out the impacts associated with small firms' lower adoption rates – "as a consequence they (small firms) miss out

³⁶ Ibid.



³³Atkinson, Robert D., Stephen Ezell, Scott M. Andes, Daniel Castro, and Richard Bennett. 2010. "The Internet Economy 25 Years After .com." *Information Technology and Innovation Foundation*. http://www.itif.org/files/2010-25-years.pdf.

³⁴ Fuhr, Joseph J. and Stephen B. Počiask. 2007. "Broadband Services: Economic and Environmental Benefits." American Consumer Institute. http://www.nederlandbreedbandland.nl/uploaded/FILES/Final_Green_Benefits.pdf.

³⁵ Atkinson, Robert D. 2009. "Policies to Increase Broadband Adoption at Home." *Information Technology and Innovation Foundation*. http://www.itif.org/files/2009-demand-side-policies.pdf.

³⁷ Crandall, Robert W., and Charles L. Jackson. 2003. "The \$500 Billion Opportunity: The Potential Economic Benefit of Widespread Diffusion of Broadband Internet Access." In *Down to the Wire: Studies in the Diffusion and Regulation of Telecommunications Technologies*, edited by Allan L. Shampine. Haupaugge, NY: Nova Science Press.

³⁸ Atkinson, Robert D. and Daniel Castro. 2008. "Digital Quality of Life: Understanding the Personal and Social Benefits of the Information Technology Revolution." *Information Technology and Information Foundation*. http://www.itif.org/files/DQOL.pdf.

³⁹ Fuhr, Joseph J., and Stephen B. Pociask. 2007. "Broadband Services: Economic and Environmental Benefits." *American Consumer Institute*. http://www.nederlandbreedbandland.nl/uploaded/FILES/Final_Green_Benefits.pdf.

⁴⁰ Katz, Raul, and Stephan Suter. 2009. "Estimating the Economic Impact of the Broadband Stimulus Plan." Working paper, Columbia Institute for Tele-Information.

⁴¹ Burton, M.L., and M.J. Hicks. 2005. "The Residential and Commercial Benefits of Rural Broadband: Evidence from Central Appalachia." Center for Business and Economic Research, Marshall University. Article published August 1. www.marshall.edu/cber/research/broadband/Final%20Rural%20Broadband%20July%202005.pdf.

⁴² LECG Ltd. 2009. *Economic Impact of Broadband: An Empirical Study*. London: LECG.

http://www.connectivityscorecard.org/images/uploads/media/Report_BroadbandStudy_LECG_March6.pdf
 ⁴³ Smihly, Maria, and Hendrikus Storm. 2010. "ICT Usage in Enterprises 2009." *Data in Focus, Eurostat.* http://epp.eurostat.ec.europa.eu/cache/ITY_OFFPUB/KS-QA-10-001/EN/KS-QA-10-001-EN.PDF.

on ICT-enabled productivity gains, further widening the economic gap between large and small firms."⁴⁴

Atkinson et al. state that taking a business online gives companies a potential customer base 20 to 30 times larger than those enjoyed by stores in even the largest metropolitan areas.⁴⁵ They also claim that Internet solutions available as a result of broadband help firms boost productivity and cut costs, thereby enabling them to cut prices and expand output. More specifically, Atkinson states that even larger savings accrue to firms from creating Internet-enabled supply chains. The ability to track shipments online allows firms to better time production and to anticipate bottlenecks in supplies, while up-to-the-minute information about inventories tells suppliers when fresh deliveries are needed.⁴⁶ Broadband can reduce transaction costs by making it easier to conduct business and commerce online.⁴⁷

B.2 Broadband Access and Adoption

Our study will examine both broadband availability and broadband adoption when considering how BTOP grants influence broadband use. While these factors have been conflated in many studies due to data limitations, the causes of each are different. The following definitions will be used in the study:

- Availability: According to the National Broadband Map (NBM), broadband service might be available to an area, but only if construction or other work is completed first. The NBM is not intended to include broadband service that is not available to a customer within a typical service interval (7 to 10 business days) and/or without an extraordinary commitment of resources.
- Access: Dependent on availability, access determines if the household / business have the tools/knowledge required to take advantage of the available technology.
- Adoption: The integration of these technologies into processes / daily life. "How do you behave differently now that you have broadband" / "How do you do business differently now that you have broadband"?

Broadband availability is more strongly driven by engineering limitations inherent in broadband technology and economic limitations on the types of projects that are profitable to undertake. Adoption is determined, in part, by the characteristics of the population that could be served and the area in which service is provided.

Evaluating non-Internet users (which as of 2010 was approximately 28.9 percent of adults), 3.1 percent cite lack of availability as their reason for not adopting while 25.3 percent state that it is too expensive.⁴⁸ Citing "not available" as reason for not subscribing to home broadband services is a much more prevalent in rural America than in urban areas. In fact, "not available" accounts for 9.4 percent of the main reasons for non-use in rural areas but accounts for only about a 1 percent factor in urban areas.⁴⁹ Smaller market sizes and low population densities are less attractive to service providers. Additionally, several industries (banking, finance, insurance, real estate, and

⁴⁹ Ibid.



⁴⁴ Jayakar, Krishna, Amit Schejter, and Richard Taylor. 2010. "Small Businesses and Broadband: Key Drivers for Economic Recovery." Working paper, Institute for Information Policy at Pennsylvania State University. http://www.economic.com/institute/for/information/policy/at/Pennsylvania/for/information/f

http://comm.psu.edu/about/centers/institute-for-information-policy/smallbusiness.pdf.

⁴⁵ Atkinson, Robert D., Stephen Ezell, Scott M. Andes, Daniel Castro, and Richard Bennett. 2010. "The Internet Economy 25 Years After .com." *Information Technology and Innovation Foundation*. http://www.itif.org/files/2010-25-years.pdf.

⁴⁶ Ibid.

⁴⁷ Atkinson, Robert D. 2007. "The Case for a National Broadband Policy." *Information Technology and Innovation Foundation.* www.itif.org/files/CaseForNationalBroadbandPolicy.pdf

⁴⁸ National Telecommunications and Information Administration. 2011. "DIGITAL NATION: Expanding Internet Usage." United Sates Department of Commerce. Article published February.

http://www.ntia.doc.gov/files/ntia/publications/ntia_internet_use_report_february_2011.pdf

business services) are more intensive users of advanced IT and digital communication technology. Variation in industry concentration causes the level of information infrastructure to be "unevenly distributed across the country, agglomerating in densely populated urban areas and in regions with large populations of certain types of users."⁵⁰

B.2.1 Factors Affecting Broadband Availability

Service providers ultimately determine the supply of services to underserved and unserved regions. Chaudhuri and Flamm suggest that broadband services may not be available in certain geographical areas "not because the residents are not interested, but because service providers find the economics of small and sparsely populated markets unattractive."⁵¹ Greenstein notes the enhanced profitability of operating in dense regions relative to those sparsely populated due to construction and operating costs.⁵² NTIA further emphasizes the economic reason for slow deployment to rural areas by stating, "the cost to serve a customer increases the greater the distance among customers."

Broadband service over cable and DSL is also limited by technical problems incurred with distance and service to a smaller number of customers."⁵³

From an engineering standpoint, there exists several barriers to broadband deployment and adoption. The key factors in the economics of broadband deployment are the radius of service from a single terminal note and the population density of the area that might be served. For example, DSL technology can only extend at most 18,000 feet from the central office switch.⁵⁴ This shrinks to 12,000 feet for high-quality, low-interruption service. The population density within the served radius limits the number of broadband users, with denser areas being more profitable, at least until the capacity of the central office switch is used up. Likewise, cable systems are more profitable in high-density areas for the same reasons. The cost of deployment for a fixed number of subscribers increases dramatically for providers in sparsely populated areas. This will deter the development of information infrastructure in those areas.

A second engineering barrier is that, with the exception of satellite and some fiber-optic networks, the provision of broadband services is essentially an overlay on top of existing telecommunications networks. These networks evolved to provide voice service, not broadband access. The extent to which the existing telecommunications network is capable of providing broadband service at all depends on the extent to which it has any remaining capacity after providing voice service to existing customers. In rural areas, provision of advanced services is correlated with the population density per route mile. As described by the NTIA and the USDA's Rural Utilities Service (RUS), "low density equates to long loops. When loops are long, they are frequently loaded, which

⁵⁴ Greenstein, Shane. 2005. "The Economic Geography of Internet Infrastructure in the United States." In *The Handbook of Telecommunications Economics Volume II*, edited by Martin Cave, Sumit Majumdar, and Ingo Vogelstang, 289-364. New York: Elsevier. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/geography.pdf.



⁵⁰ Greenstein, Shane, and Mercedes Lizardo. 1999. "Determinants of the Regional Distribution of Information Technology Infrastructure in U.S." In *The Electronic Village: Public Policy Issues of the Information Economy*, edited by Dale Orr and Tom Wilson. Toronto: C.D. Howe Institute.

http://www.kellogg.northwestem.edu/faculty/greenstein/images/htm/Research/chapters/Determinants%20of%20the%20Regio nal%20Distribution%20of%20Information%20Technology%20Infrastructure%20in%20U.S..pdf.

⁵¹ Chaudhuri, Anindya, and Kenneth Flamm. 2005. "An Analysis of the Determinants of Broadband Access." Presented at the Telecommunications Policy Research Conference, Washington, DC, September 24. http://web.si.umich.edu/tprc/papers/2005/485/Broadband.pdf.

⁵² Greenstein, Shane. 2005. "The Economic Geography of Internet Infrastructure in the United States." In *The Handbook of*

Telecommunications Economics Volume II, edited by Martin Cave, Sumit Majumdar, and Ingo Vogelstang, 289-364. New York: Elsevier. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/geography.pdf.

⁵³ National Telecommunications and Information Administration and United States Department of Agriculture's Rural Utilities Service. 2000. "Advanced Telecommunications in Rural America, The Challenge of Bringing Broadband Service to All Americans." Accessed November 15, 2010. http://www.ntia.doc.gov/reports/ruralbb42600.pdf.

prevents DSL operation.^{*55} The result of this can be slow, unreliable, or unavailable broadband service in rural areas with minimally-built telecommunications networks. Low population density can also result in a limited number of providers and limited competition among them, resulting in higher costs for broadband versus more densely populated areas. Some sparsely populated areas of the country have yet to obtain access to broadband services at all.⁵⁶

Americans in rural areas tend to have lower broadband adoption rates than their demographic counterparts in urban areas. This applies across race (African-Americans and Hispanics in rural areas exhibited much lower levels of broadband use than their urban counterparts), and employment levels (employed and unemployed persons in rural America had significantly lower broadband use than those living in urban areas). Although overall adoption rates are lower for rural areas, these rates are increasing. NTIA reports that the gap decreased from 11.8 percentage points in 2009 to 10.1 percentage points in 2010. Three years earlier, in 2007, the gap was 15.0 percentage points (53.8 percent versus 38.8 percent).⁵⁷

Comparing rural areas across the country, the NTIA reports, "the likelihood of offering advanced services in rural service areas is highest in the Southeast and Northeast, lower in the Midwest, and lowest in the Southwest."⁵⁸ Further emphasized in another recent report, they point out that Internet usage is lowest in the South and Midwest.⁵⁹ This discrepancy indicates there are additional factors influencing broadband availability and adoption in underserved and unserved regions, aside from only the urban- rural distinction.

B.2.2 Factors Affecting Adoption

Certain social and demographic factors aid in defining unserved and underserved regions. Noted in many studies, and specifically in reports published by the NTIA, is the correlation between race, age, and levels of both income and education, and broadband access. These factors establish what is widely recognized as the "digital divide."⁶⁰ These factors play a key role in the phenomenon known as the "digital divide."⁶¹

The term "digital divide" initially signified the gap in ownership of computers between demographic groups.⁶² This definition was later extended to include the gap between people with effective access to digital and information technology, including broadband, and those with very limited or no access. This included the imbalance both in physical access to technology and the resources and skills needed to effectively participate as a digital citizen. In 1995 the NTIA published, "Falling through the Net: Toward Digital Inclusion".⁶³ This report outlined the growing disparity in Internet

⁶³ National Telecommunications and Information Administration. 1995. "Falling Through the Net: A Survey of the 'Have Nots' in Rural and Urban America." Accessed January 27, 2012. http://www.ntia.doc.gov/ntiahome/fallingthru.html.



⁵⁵ National Telecommunications and Information Administration and United States Department of Agriculture's Rural Utilities Service. 2000. "Advanced Telecommunications in Rural America, The Challenge of Bringing Broadband Service to All Americans." Accessed November 15, 2010. http://www.ntia.doc.gov/reports/ruralbb42600.pdf

⁵⁶ National Telecommunications and Information Administration. 2010. "Digital Nation: 21st Century America's Progress Towards Universal Broadband Internet Access." United States Department of Commerce.

http://www.ntia.doc.gov/reports/2010/NTIA_internet_use_report_Feb2010.pdf.

⁵⁷ National Telecommunications and Information Administration. 2011. "Digital Nation: Expanding Internet Usage." United States Department of Commerce. Article published February.

http://www.ntia.doc.gov/files/ntia/publications/ntia_internet_use_report_february_2011.pdf

⁵⁸ National Telecommunications and Information Administration and United States Department of Agriculture's Rural Utilities Service. 2000. "Advanced Telecommunications in Rural America, The Challenge of Bringing Broadband Service to All Americans." Accessed November 15, 2010. http://www.ntia.doc.gov/reports/ruralbb42600.pdf

⁵⁹ National Telecommunications and Information Administration. 2010. "Digital Nation: 21st Century America's Progress Towards Universal Broadband Internet Access." United States Department of Commerce. http://www.ntia.doc.gov/reports/2010/NTIA_internet_use_report_Feb2010.pdf.

 ⁶⁰ Chaudhuri, Anindya, and Kenneth Flamm. 2005. "An Analysis of the Determinants of Broadband Access." Presented at the Telecommunications Policy Research Conference, Washington, DC, September 24.

http://web.si.umich.edu/tprc/papers/2005/485/Broadband.pdf.

⁶¹ Ibid.

⁶² Compaine, Benjamin M, ed. 2001. The Digital Divide: Facing a Crisis or Creating a Myth? Massachusetts: MIT Press.

access, broadband in particular, between different demographically defined groups and not just on the availability of the necessary hardware, and more on the disparity of access.

Today the term "digital divide" is also associated with differences by gender, income, race, and location. Results from the FCC Broadband Adoption & Use in America survey indicate that the main dividing factors for broadband adoption are education, income, age, and people with disabilities.⁶⁴ Servon argues that the digital divide is a symptom of a larger and more complex problem – that of persistent poverty and inequality.⁶⁵ Mehra identifies socioeconomic status, income, educational level, and race among other factors associated with technological attainment, or the potential of the Internet to improve everyday life for those on the margins of society and to achieve greater social equity and empowerment.⁶⁶ To the extent that access to broadband technology could help with economic advancement, for instance through improved educational or job search opportunities, the digital divide could represent a self-perpetuating form of social and economic inequality.

Characteristics of broadband adopters give some indications of who is more likely to be in the population of broadband adopters, and who is not. For example, there is a substantial discrepancy in Internet usage comparing those over 25 years old without high school diplomas (28 percent) to college graduates (84 percent).⁶⁷ This suggests that for areas in which low levels of education are characteristic, broadband adoption rates and availability may be lower. It is also possible that those with higher levels of education are more aware of, or more able to afford the benefits of broadband, and therefore settle in areas that have, or will soon have, broadband access. Doms notes, "metropolitan areas with highly educated workforces are those that are likely to become more computer intensive, and these are also areas that enjoy faster real wage growth."⁶⁸ Such areas appear more attractive to service providers. Higher wages suggest greater demand, in addition to broadband adoption being correlated with higher levels of education. Greenstein points out, "the wealth and income of the area's residential population is key to understanding demand."⁶⁹

Similar reasoning can be applied to income differentials across regions of the country. The highest broadband use at home was by those with household incomes greater than \$150,000 annually (89 percent) and the lowest use was by those persons living in households with \$15,000 annual family income or less (28 percent).⁷⁰ An area's per capita income not only influences the population's broadband adoption (based on affordability), but also determines suppliers' entry decisions. As Greenstein and Lizardo note, "real per capita income and the density of large-scale computer users in the region have become important explainers of the distribution of infrastructure."⁷¹

http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/chapters/Determinants%20of%20the%20Regio nal%20Distribution%20of%20Information%20Technology%20Infrastructure%20in%20U.S..pdf.



⁶⁴ Horrigan, John B. 2010. "Broadband Adoption and Use in America." OBI Working Paper Series No. 1. Federal Communications Commission. http://hraunfoss.fcc.gov/edocs_public/attachmatch/DOC-296442A1.pdf.

⁶⁵ Servon, Lisa. 2002. Bridging the Digital Divide: Technology, Community, and Public Policy. Massachusetts: Blackwell Publishers Ltd.

⁶⁶ Mehra, Bharat, Cecelia Merkel, and Ann P. Bishop. 2004. "The Internet for Empowerment of Minority and Marginalized Users", New Media and Society, 781–802. doi:10.1177/146144804047513.

⁶⁷ National Telecommunications and Information Administration. 2010. "Digital Nation: 21st Century America's Progress Towards Universal Broadband Internet Access." United States Department of Commerce.

http://www.ntia.doc.gov/reports/2010/NTIA_internet_use_report_Feb2010.pdf.

⁶⁸ Doms, Mark. 2005. "The Diffusion of Personal Computers across the U.S." The Federal Reserve Bank of San Francisco Economic Letter 2005-37. http://www.frbsf.org/publications/economics/letter/2005/el2005-37.pdf.

⁶⁹ Greenstein, Shane. 2005. "The Economic Geography of Internet Infrastructure in the United States." In *The Handbook of Telecommunications Economics Volume II*, edited by Martin Cave, Sumit Majumdar, and Ingo Vogelstang, 289-364. New York: Elsevier. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/geography.pdf.

⁷⁰ National Telecommunications and Information Administration. 2010. "Digital Nation: 21st Century America's Progress Towards Universal Broadband Internet Access." Washington DC: US Department of Commerce. Last modified February. http://www.ntia.doc.gov/reports/2010/NTIA_internet_use_report_Feb2010.pdf.

⁷¹ Greenstein, Shane, and Mercedes Lizardo. 1999. "Determinants of the Regional Distribution of Information Technology Infrastructure in U.S." In *The Electronic Village: Public Policy Issues of the Information Economy*, edited by Dale Orr and Tom Wilson. Toronto: C.D. Howe Institute.

Age is also negatively correlated with broadband adoption. However, broadband usage among adults ages 65 or older grew from 19 percent in May, 2008 to 30 percent in April, 2009.⁷² The same discrepancies exist, albeit less severe, when comparing the portion of the population using dial-up to broadband users. On average, dial-up users are "older, have lower incomes, have lower levels of educational attainment, are more likely to be African American, and more likely to live in rural areas."⁷³ The supply of broadband service and the willingness of the population in the service area to pay for broadband determine the price at which the service is provided. This cost to consumers can be another reason cited for non-adoption by underserved populations. Broadband service is notably more expensive than dial-up, thus cost can be assumed to be highly influential on low-income populations (which comprise a substantial portion of underserved regions). Chaudhuri and Flamm note that a report from the Office of Technology Policy of the Department of Commerce identifies cost as "the most obvious factor limiting broadband demand."⁷⁴ In his study of broadband adoption policies, Robert Atkinson recognizes the price elasticity of demand in broadband adoption, and suggests, "efforts to reduce costs can play a role in spurring demand, particularly among lower income households."⁷⁵

⁷⁵ Atkinson, Robert D. 2009. "Policies to Increase Broadband Adoption at Home." Information Technology and Innovation Foundation. http://www.itif.org/files/2009-demand-side-policies.pdf.



⁷² Horrigan, John. 2009. "Home Broadband Adoption 2009: Broadband Adoption Increases, But Monthly Prices Do Too." *Pew Research Center*. http://www.pewinternet.org/~/media//Files/Reports/2009/Home-Broadband-Adoption-2009.pdf.

⁷³ Horrigan, John. 2009. "Home Broadband Adoption 2009: Broadband Adoption Increases, But Monthly Prices Do Too." *Pew Research Center*. http://www.pewinternet.org/~/media//Files/Reports/2009/Home-Broadband-Adoption-2009.pdf.

⁷⁴ Chaudhuri, Anindya, and Kenneth Flamm. 2005. "An Analysis of the Determinants of Broadband Access." Presented at the Telecommunications Policy Research Conference, Washington, DC, September 24. http://web.si.umich.edu/tprc/papers/2005/485/Broadband.pdf.

Appendix C. Social Impacts

C.1 Introduction

This section presents background information on data sources that will be used in our study for the purpose of benchmarking projects in the following key impact areas: education, healthcare, public safety, and quality of life. In part due to the construction of the BTOP programs, the impacts of BTOP grants in these areas will affect numerous and diverse issues and types of institutions. For instance, educational impacts could be observed at K-12 institutions, community colleges, 4-year institutions, universities, or other education providers. Likewise, healthcare impacts might be observed at primary care physicians' offices, hospitals, or in areas served by nurse practitioners. Finally, public safety impacts might be observed in emergency response entities, civil defense groups, or police forces. As a result of the multitude of potential impacts and impact areas of a general purpose technology like broadband, it is not feasible to use a standardized instrument to collect data on these topics. Further, there is a quite limited set of public use data available covering indicators of social impacts. We have therefore included these areas in our case study methodology in order to develop in depth data on the particular impacts on education, healthcare, and public safety caused by BTOP grants. To the extent possible, however, we will conduct statistical analyses on the public use data sets identified as sources for the identified social impact criteria discussed in each of the sub-sections below in an attempt to detect correlation and/or causality between BTOP grant funding and changes in these data.

C.2 Data Availability

Various social measures will be collected through public use data and grant applications in order to characterize the area impacted by the grant. Close review of the grant application and conversations with the grantee will be utilized in order to determine the impacted area or "the community" to be considered for each grant.

We assume that data sources outlined below will be available for key variables on an ongoing basis throughout the study, and that the quality of these data is sufficient for use in the study. Our data is assembled from subject-specific sources which cover: Education, Healthcare, Public Safety and Quality of Life / Standard of Living. In the event of discontinuance of or large changes in a data source, data collection may have to be reassessed in order to achieve the study goals. Table 5 below illustrates the types of variables we propose to obtain and data sources from which we will be able to obtain them. The same data may be available from multiple sources; in this case, we list the most reliable or extensive source we have found. Descriptions of these data sets are provided in the following subsections.



Category	Variable(s)	Source	Level of Granularity
Education	Drop-out / Graduation Rates	Local Education Agency Universe Survey Dropout Data	State
	Student Factors, Instructional Content and Practice, Teacher Factors, School Factors, Community Factors, Factors Beyond School, and Government Factors	The National Assessment of Educational Progress (NAEP)	State
	Instituational Characteristics, Institutional Prices, Enrollment, Student Financial Aid, Degrees and Certificates Conferred (Completions), Student Persistence and Success, and Institutional Resources	Integrated Postsecondary Education Data System (IPEDS)	College Level
Healthcare	Variables covering: E-Care, Electronic Health Records, Telehealth, and Mobile Health	American Hospital Associations (AHA) annual survey (electronic health records data)	Hospital Level
Public Safety	Variables covering: Crime, Justice and Socio-demographics	Crime & Justice Electronic Data (Bureau of Justice Statistics)	State Level
	 Number of: computers in the field, vehicle mounted computers and other computers Digital: fingerprints, mugshots and suspect composites used 	Law Enforcement Management and Administrative Statistics (LEMA)	State / Local agency Level
	 The nature and types of specific offenses in the incident, Characteristics of the victim(s) and offender(s), Types and value of property stolen and recovered, and Characteristics of persons arrested in connection with a crime incident. 	National Incident-Based Reporting System (NIBRS)	City Level
Quality of Life / Standard of Living	 Use the Internet at any location? Connect to the Internet from home? Do you currently access the Internet at home using What is the main reason that you do not have high-speed Internet access at home? 	Schooling and Computer / Internet Usage Supplement - Current Population Survey (CPS) - Census	County Level
	- Were you registered, - Did you vote, - Main reason not registered, - Main reason did not vote	Voter registration and participation supplement - Current Population Survey (CPS) - Census	County Level
	 Contacted a public official, Bought or boycotted based on political views, Type of group joined, How often did you communicate via email or internet 	Membership in civic associations supplement - Current Population Survey (CPS) - Census	County Level
	 Volunteer activities for an organizations, including time spent, types of organizations, and types of volunteer activities, and location of volunteer activities Volunteer activities for children's school or youth organizations, including time spent and types of volunteer activities and location of volunteer activities Charitable donations 	Volunteer Workers Supplement - Current Population Survey (CPS) - Census	County Level
	 Type of telecommuting, e.g. employer-sponsored telework vs. home business telework Frequency per week (days and hours) of telework Computer use while working from home 	Telecommuting Supplement - Current Population Survey (CPS) - Census	County Level
	Variables covering: working from home	American Community Survey (ACS) - Census	County Level

Table 5. Potential Data Sources and Variables

C.3 Measuring Impacts on Education

Broadband technologies have been discussed as a key tool in the work that needs to be done to maintain and improve educational outcomes in the United States. Home broadband users are more likely than non-users to plan further education.⁷⁶ The ongoing integration of IT into the activities of workers in most fields is no longer a subject for debate or consideration, but a reality that is forcing

⁷⁶ LaRose, Robert, Jennifer L. Gregg, Sharon Stover, Joseph Straubhaar, and Nobuya Inagaki. 2008. "Closing the Rural Broadband Gap." Final Technical Report, Michigan State University. Report published November 30. https://www.msu.edu/~larose/ruralbb/.



both employers and educators to determine what steps are necessary to ensure a well trained and well educated population. There are two major challenges facing our educational system that must be addressed:

- Not everyone who starts in the educational system today successfully exits. An incomplete education can be a lifelong stumbling block. By providing alternatives to existing forms of education, broadband-based technologies could play a role in reducing dropping out.
- The demography of the United States is shifting so that substantially all of the growth in population is projected to come from minority groups, primarily from Hispanics. The extent to which these populations are unserved or underserved with respect to broadband could have an effect on the availability of educational opportunities and potentially on educational attainment for these groups.

Prior Internet use is positively correlated to learning and adapting new methods of computer and Internet methods.⁷⁷ Also, the use of "distance learning" in the higher education space is more costeffective and breaks down the physical barriers apparent for some students.⁷⁸ The potential benefits of such development are encouraging for the future of our education system. In a study on technology and student achievement, Protheroe reports that: "when properly implemented, computer technology had a significant effect on student achievement, as measured by test scores across subject areas and with students at all levels."⁷⁹ This positive impact transcends many demographic and socio-economic groups. For example, "a study by the American Psychological Association found that low-income children who used the Internet on a regular basis performed better on standardized tests of reading achievement and had higher grade point averages than did children who used it less."⁸⁰ Additionally, "a study of the Computers for Youth model, which provides low-income families with discounted laptops and Internet connections, also found a positive correlation between increased computer and Internet use and improved test scores."

Internet and ICT applications are prevalent in elementary and secondary schools across the country. Cleary noted that by 2003, 95 percent of all public schools with Internet access used broadband.⁸² Survey findings indicate that online learning has been growing in K–12 schools and that this growth will continue for the foreseeable future.⁸³ Research shows that home use of computers and broadband technologies for learning can be a significant factor in boosting math and reading achievement.⁸⁴ Use of computers and broadband at home for educational purposes has also been shown to motivate students and to increase the relevance of content during school hours, ultimately improving student achievement.^{85,86,87,88}

- http://www.uschamber.com/sites/default/files/about/0904Broadband_and_Telemedicine.pdf.
- ⁸¹ *Ibid.*

⁸⁵ Valentine, Gill, Jackie Marsh, Charles Pattie, and BMRB. 2005. "Children and Young People's Home Use of ICT for Educational. Purposes: The Impact on Attainment at Key Stages." http://www.dcsf.gov.uk/research/data/uploadfiles/RR672.pdf.



⁷⁷ Eastin, Matthew S., and Robert LaRose. 2000. "Internet Self-Efficacy and the Psychology of the Digital Divide." *Journal of Computer-Mediated Communication* 6, no. 1. http://jcmc.indiana.edu/vol6/issue1/eastin.html#abstract.

⁷⁸ Hulbert, Linda Ann, and Regina C. McBride. 2004. "Utilizing Videoconferencing in Library Education: A Team Teaching Approach." Working paper, Journal of Education for Library and Information Services, Association for Library and Information Science Education.

⁷⁹ Protheroe, Nancy. 2005. "Technology and Student Achievement." *Principal*, November-December. http://www.naesp.org/resources/2/Principal/2005/N-Dp46.pdf.

⁸⁰ Davidson, Charles, and Michael J. Santorelli. 2010. "The Impact of Broadband on Education." Working paper, Advanced Communications Law & Policy Institute at New York Law School.

⁸² Cleary, Paul. 2009. "The Impact of Broadband on Education in the USA," In *Encyclopedia of Multimedia Technology and Networking Second Edition*, edited by Margherita Pagani. IGI Global. http://www.igi-global.com/Bookstore/Chapter.aspx?Titleld=17461.

⁸³ Picciano, Anthony G., and Jeff Seaman. 2007. "K–12 Online Learning: A Survey of U.S. School District Administrators." Sloan Consortium. http://sloanconsortium.org/sites/default/files/K-12_Online_Learning_1.pdf.

⁸⁴ Texas Center for Education Research. 2009. "Evaluation of the Texas Technology Immersion Pilot: Final Outcomes for a Four-Year Study (2004–05 to 2007–08)." http://www.etxtip.info/y4_etxtip_final.pdf.

C.3.1 Online Learning

In 2007, the Sloan Consortium issued a report on the extent and nature of online learning in K-12 schools. Entitled, K-12 Online Learning: A Survey of U.S. School District Administrators, it was based on a national survey of school district administrators during the 2007-2008 academic year.⁸⁹ It was one of the first studies to collect data on and compare fully online and blended learning (part online and part traditional face-to-face instruction) in K-12 schools. Of particular importance, the study found that three-quarters of the responding public school districts are offering online or blended courses:

- 75 percent had one or more students enrolled in a fully online or blended course.
- 70 percent had one or more students enrolled in a fully online course.
- 41 percent had one or more students enrolled in a blended course.

These percentages represent an increase of approximately 10 percent in school districts offering online or blended courses since 2005-2006. Sixty-six percent of school districts with students enrolled in online or blended courses anticipate their online enrollments will grow. The overall number of K-12 students engaged in online courses in 2007-2008 is estimated at 1,030,000. This represents a 47 percent increase since 2005-2006.

Infrastructure (CCI) grants have the potential to have by far the largest educational impact. Based on data gathered by NTIA shortly after the award phase, over 8,000 educational anchor institutions will be affected.

Online learning has also been implemented at community anchor institutions, but its applicability appears to be anchor type specific. For example, interviews with library patrons as part of one study revealed in that instance library "public access is not suitable for online courses owing to limited hours of operation, short duration appointments for library computers, and overcrowding. Improved broadband access for educational purposes is thus in need of further attention."⁹⁰

C.3.2 K-12 Institutions

K-12 institutions are by far the majority of institutions affected by BTOP grants. NTIA has identified 6,882 K-12 anchor institutions potentially affected by CCI grants, exclusive of the effects of PCC and SBA grants.

We will be able to obtain a list of the K-12 anchor institutions affected by the CCI grants based on Quarterly and Annual PPRs by the grantees and information provided to NTIA. We will use data from the multiple sources (described below) to compile a list of anchor institutions that are receiving BTOP funding. Access will be measured by matching this list to the list of anchor institutions provided in the NTIA broadband mapping data.

⁹⁰ LaRose, Robert, Jennifer L. Gregg, Sharon Stover, Joseph Straubhaar, and Nobuya Inagaki. 2008. "Closing the Rural Broadband Gap." Final Technical Report, Michigan State University. Report published November 30. https://www.msu.edu/~larose/ruralbb/.



⁸⁶ Ito, Mizuko, Heather Horst, Matteo Brittanit, Danah Boyd, Becky Herr-Stephenson, Patricia G. Lange, C.J. Pascoe, and Laura Robinson. 2008. "Living and Learning with New Media Summary of Findings from the Digital Youth Project." Article published November. http://digitalyouth.ischool.berkeley.edu/files/report/digitalyouth-WhitePaper.pdf.

⁸⁷ Passey, Don, Colin Rogers, Joan Machell, and Gilly McHugh. 2004. "The Motivational Effect of ICT on Pupils." http://www.dcsf.gov.uk/research/data/uploadfiles/RR523new.pdf.

⁸⁸ Becta, Minister. 2008. "Minister's Taskforce on Home Access to Tech." *Extending Opportunity* 4.

http://partners.becta.org.uk/upload-dir/downloads/page_documents/partners/home_access_report.pdf.

⁸⁹ Picciano, Anthony, and Jeff Seaman. 2009. "K–12 Online Learning: A 2008 Follow-up of the Survey of U.S. School District Administrators." *Sloan Consortium*. http://sloanconsortium.org/sites/default/files/k-12_online_learning_2008.pdf

Limited information is available on educational outcomes on a school-by-school basis. The National Assessment of Educational Progress (NAEP) is the largest nationally representative and continuing assessment of what America's students know and can do in various subject areas. Assessments are conducted periodically in mathematics, reading, science, writing, the arts, civics, economics, geography, and U.S. history. NAEP assessments are administered uniformly using the same sets of test booklets across the nation. NAEP results serve as a common metric for all states and selected urban districts. The assessment stays essentially the same from year to year, with only carefully documented changes. This permits NAEP to provide a clear picture of student academic progress over time.

NAEP, also known as The Nation's Report Card, is the only nationally representative and continuing assessment of what America's students know and can do in various subject areas. Since 1969, assessments have been conducted periodically in reading, mathematics, science, writing, U.S. history, civics, geography, and the arts. The NAEP budget supports the following program components:

- National NAEP—National NAEP reports information for the nation and specific geographic regions of the country, includes students drawn from both public and nonpublic schools, and reports results for student achievement in grades 4, 8, and 12;
- State NAEP—These assessments provide reliable state-level student achievement data in reading, mathematics, science, and writing;
- NAEP Trial Urban District Assessment—Federal appropriations authorized for the No Child Left Behind Act supported a multiyear study of the feasibility of a Trial Urban District Assessment of Educational Progress, with the first assessment occurring in reading and writing in 2002 for five urban districts, and 2003 reading and mathematics assessment results available for 10 such districts; and
- Long-Term Trend (LTT)—NAEP LTT assessments, designed to give information on the changes in the basic achievement of America's youths, are administered nationally and report student performance at ages 9, 13, and 17 in reading and mathematics.

Based on the review of the literature and the availability of data from the NAEP data set, the following categories contain measures that could be applied at the anchor institution level to measure the education-related impacts of BTOP grants:

- Major Reporting Groups (includes subcategories Student Factors, School Factors, and Community Factors)
 - Generally seen as the most important and popular data in NAEP. Consisting of more than a dozen variables, their availability depends on the assessment and criteria selected. (For more information, see the glossary on their Web site or search their Web site for "reporting groups.")
- Student Factors (includes subcategories Demographics, Affective disposition, Academic record and school experience)
 - Data about students, their parents, students' level of interest and activity in school, student opinions about the subject being assessed, absenteeism, academic record, and school experience.
- Instructional Content and Practice (includes subcategories Curriculum, Course offerings, Classroom management, Grouping, Modes of instruction/classroom activities)
 - Course offerings, emphasis and time spent on topic matter, role of parents and guests in teaching, students' ability levels, modes of instruction, classroom activity, and computer use (not available in LTT).
- Teacher Factors



- Teacher race, teacher education and background, attitudes, expectations, availability of support for teachers, and teacher satisfaction. Teacher variables are not a part of LTT, and are not generally part of grade 12 assessments.
- School Factors
 - Demographics, percentage of students taking particular classes or coursework, average class size and duration, school type, assignment of coursework by ability, governance, resources (computers, field trips, tech support, summer programs), test requirements for students, school climate, parental involvement, teacher morale, student attitudes, enrollment, safety, absenteeism, cheating, and extracurricular activities (not available in LTT).
- Community Factors
 - Region of the country or school location.
- Factors Beyond School
 - Home environment, media at home, students' use of time outside school, homework, parents and homework, and more.
- Government Factors
 - State and district requirements for students (not in LTT).

We will use measures from NAEP, where comparable, to illustrate the differences or similarities between results obtained at K-12 institutions receiving BTOP funds and their counterparts elsewhere. Statistical analyses will be applied to determine if causality or correlation can be detected between BTOP grant funding and student achievement scores as reported in NAEP.

Another potential source of data that could be used in our assessment of the impact of broadband on education is Local Education Agency Universe Survey Dropout and Completion annual data.⁹¹ These data are available at the county level for the 2002-2003 and 2005-2006 school years, while more recent years are available at the state level.

We will match these dropout data to Quarterly / Annual PPRs collected by NTIA from grantees. This could provide a measure of dropout rates in areas impacted by BTOP, and those rates could potentially be compared historically or against areas not impacted by BTOP grant funding, although, the evaluation study team does not anticipate being able to observe any direct impacts from BTOP grant activity at the state level.

We will further examine these dropout data sources, and, where possible, use these to compare anchor institutions based on broadband availability and adoption. Statistical analysis will be applied to determine causality or correlation between these statistics and BTOP funding.

C.3.3 Community College & Four Year Institutions

As in the case of K-12 institutions, community colleges and four-year institutions will receive broadband service as a result of BTOP grants, although these institutions are not as numerous as K-12 institutions.

The Integrated Postsecondary Education Data System (IPEDS) provides a detailed look at postsecondary institutions in the United States. IPEDS data reports are available online from the U.S. Department of Education, Institute of Education Sciences. We will identify the community colleges affected by BTOP grants and assemble their IPEDS data on an ongoing basis. We will

⁹¹ National Center for Education Statistics. *Local Education Agency Universe Survey Dropout and Completion Annual Data*, 1997 – 2008. Accessed January 27, 2012. http://nces.ed.gov/ccd/drpagency.asp.



then look for differences in IPEDS results between those institutions receiving BTOP funds and those receiving no such funding.

Based on the review of the literature and the availability of data from the IPEDS data set, the following categories contain measures that could be applied at the anchor institution level to measure the education-related impacts of BTOP grants:

- **Institutional Characteristics:** These include basic institutional contact information, tuition and fees, room and board charges, control or affiliation, type of calendar system, levels of awards offered, types of programs, and admissions requirements.
- **Institutional Prices:** This includes tuition and fee data as well as information on the estimated student budgets for students based on living situations (on-campus or off-campus).
- **Enrollment:** This includes Fall Enrollment, Residence of First-Time Students, Age Data, Head Count, Instructional Activity, and Total entering class.
- Student Financial Aid: This includes the number of full-time, first-time degree/certificateseeking undergraduate students who receive different types of student financial aid, including grants and loans, from different sources at each institution. IPEDS also collects data to show the average dollar amount of aid received by these students.
- **Degrees and Certificates Conferred (Completions):** This section includes data on the number of students who complete a postsecondary education program by type of program and level of award (certificate or degree).
- **Student Persistence and Success:** This includes data on First-Year Retention Rates and Graduation Rates.
- Institutional Resources: This includes Human Resources and Finance related data.

While IPEDS will provide a valuable source of information for outcomes such as graduation rates, the specific impacts of broadband on the institution might not be evident. Also, depending on the size of the broadband investment versus the size of the institution, the overall size of the impact might be too small to be captured in an aggregate measure of institution performance.

The National Broadband Mapping data may be used to identify institutions that are similar to those selected for BTOP CCI grants in terms of broadband access. In addition, we can further improve the match between comparable institutions by examining key IPEDS variables available for both. This could include class size, class composition, and graduation rates. We will develop a list of comparable institutions that will be used to serve as a control for the analysis. Statistical analysis will be applied to IPEDS data elements such as graduation rates, in order to determine the degree of causality or correlation with BTOP grant funding.

C.4 Measuring Impacts on Healthcare

Hospitals, clinics, and physicians deliver quality care on a daily basis, but they are faced with an aging population and rising healthcare costs. Improving Americans' health is one of the most important tasks for the nation. Healthcare already accounts for 17 percent of U.S. gross domestic product (GDP); by 2020, it will top 20 percent. America is aging and by 2040 there will be twice as many Americans older than 65 as there are today, and healthcare costs will likely increase as a consequence.⁹²

⁹² Center for Medicare & Medicaid Services. 2008. "National Health Expenditure Projections 2008–2018." Accessed January 27, 2012. http://www.cms.hhs.gov/NationalHealthExpendData/downloads/proj2008.pdf.



Though the United States spends more annually on healthcare than any other nation (nearly \$2 trillion), the system has not overcome some issues of quality, access, and affordability.⁹³ Daniel Castro, in his Information Technology and Innovation Foundation report on healthcare, explicitly notes, "[s]ome individuals do not receive the proper amount of care. Those who receive too much care or the wrong kind of care waste resources, while those who receive insufficient care may develop additional health problems. In many instances, the healthcare that people receive is not based on the best available scientific evidence."⁹⁴ It is this disparate level of healthcare access that has slowed the improvement of healthcare quality.⁹⁵ The services available and the care received are influenced in part by geographic location. Those in densely populated metropolitan areas have access to a wider variety of services, providers, and specialists. As Stenberg and Low point out, "[r]ural residents often face challenges accessing a full range of healthcare services. Today, due to the availability of broadband Internet, rural healthcare providers can more easily link with urban providers through the use of health IT."⁹⁶

In 1999, the FCC noted that broadband access to the Internet was in a position to "meaningfully improve [the Nation's]...healthcare services."⁹⁷ Broadband enhances healthcare efficiency via strengthened connectivity, driving innovation, and cutting-edge approaches to healthcare that are expected to lead to "vast individual and national cost savings and to an increase in the availability of quality health solutions."⁹⁸ Though its benefits are clear, access to broadband service is not identically available across the nation. In addition, from the fact that availability is not ubiquitous, the United States currently invests very little in healthcare IT relative to the other nations. Atkinson points out that total investment in healthcare IT by the United States government in 2005 was only \$0.43 per capita, which is extremely low when compared to the United Kingdom's per capita investment of \$192.79.⁹⁹

Healthcare IT includes several components, including:

- **Health IT:** Information driven health practices and the technologies that enable them. It includes billing and scheduling systems, e-care, electronic health records, telehealth, and mobile health.¹⁰⁰
- **E-Care:** The electronic exchange of information (e.g., data, images, and video) to aid in the practice of medicine and advanced analytics. It encompasses technologies that enable video consultation, remote monitoring, and image transmission over fixed and mobile networks.¹⁰¹
- Electronic Health Record (EHR): An electronic health record is a digital record of patient health information generated by one or more encounters in any care delivery setting. Included in this information are patient demographics, progress notes, diagnoses,

¹⁰¹ Ibid.



⁹³ Atkinson, Robert D., and Daniel Castro. 2008. "Digital Quality of Life: Understanding the Personal and Social Benefits of the Information Technology Revolution." *Information Technology and Information Foundation*. Article published October. http://www.itif.org/index.php?id=179.

⁴⁴ Castro, Daniel. 2007. "Improving Health Care: Why a Dose of IT May Be Just What the Doctor Ordered." *Information Technology* and *Information Foundation*, October 25. http://www.itif.org/index.php?id=88.

⁹⁵ Davidson, Charles M. and Michael J. Santorelli. 2009. "The Impact of Broadband on Telemedicine: A Study Commissioned by the U.S. Chamber of Commerce." Working paper, The Advanced Communications Law & Policy Institute at New York Law School. http://www.uschamber.com/sites/default/files/about/0904Broadband_and_Telemedicine.pdf.

⁹⁶ Low, Sarah and Peter Stenberg. 2009. "Rural Broadband at a Glance." Working paper, United States Department of Agriculture. http://www.ers.usda.gov/publications/eib47/eib47.pdf.

⁹⁷ Davidson, Charles M., and Michael J. Santorelli. 2009. "The Impact of Broadband on Telemedicine: A Study Commissioned by the U.S. Chamber of Commerce." Working paper, The Advanced Communications Law & Policy Institute at New York Law School. http://www.uschamber.com/sites/default/files/about/0904Broadband_and_Telemedicine.pdf.

⁹⁸ Ibid.

⁹⁹ Atkinson, Robert D., and Daniel Castro. 2008. "Digital Quality of Life: Understanding the Personal and Social Benefits of the Information Technology Revolution." *Information Technology and Information Foundation*. http://www.itif.org/index.php?id=179.

¹⁰⁰ Federal Communications Commission. "National Broadband Plan: Connecting America." Accessed January 27, 2012. http://www.broadband.gov/plan/10-healthcare/.

medications, vital signs, medical history, immunizations, laboratory data, and radiology reports.¹⁰²

- **Telehealth:** Often used as a synonym for e-care, but includes non-clinical practices such as continuing medical education and nursing call centers.¹⁰³
- **Mobile Health:** The use of mobile networks and devices in supporting e-care. Emphasizes leveraging health-focused applications on general-purpose tools such as smart phones and Short Message Service (SMS) messaging to drive active health participation by consumers and clinicians.¹⁰⁴

To improve its functionality, it has been widely recommended the United States healthcare adopt a system of EHRs, which contain a complete listing of patients' medical histories. The absence of single, individualized, consolidated healthcare records makes it difficult for patients to actively manage their healthcare. Healthcare providers often receive incomplete information when they treat their patients, which can result in medical errors.¹⁰⁵ A study of radiologists found that 62 percent cite a "[lack of consistent] access to patient medical records as an impediment to their work," and stating that 96 percent agreed or strongly agreed that this problem poses medical risks for patients.¹⁰⁶ The ability to create an EHR system is dependent on healthcare providers' having broadband connectivity.¹⁰⁷ Though the benefits of EHRs are widely recognized, cost is a substantial barrier to implementing such a system. Castro goes on to emphasize that the overwhelming majority of hospitals find the initial cost of EHR adoption as a "significant barrier or somewhat of a barrier."¹⁰⁸ Cost is a more significant issue for individual and smaller medical practices. Approximately two-thirds of U.S. physicians work in this type of practice, which Castro notes results in a major impediment in the development of healthcare IT.¹⁰⁹

Currently, an estimated 61 percent of adults use the Internet to obtain health information (a portion of the population Pew refers to as "e-patient"). Additionally, Pew reports 42 percent of all adults say they or someone they know has been helped by following medical advice or health information found on the Internet. This represents a significant increase from 2006 when this figure was 25 percent of all adults.¹¹⁰ More specifically, Pew reports the percentage of American adults getting exercise and fitness information online has jumped from 21 percent in 2002 to 38 percent now – an 88 percent increase^{.111} Broadband service enables the population to combat and prevent health issues with easy access to such information. Ackerman argues that patient demand, as a market force, will drive adoption of telemedicine.¹¹² This demand is self-evident in many arenas, including the cell phone market:¹¹³

¹¹³ Davidson, Charles M. and Michael J. Santorelli. 2009. "The Impact of Broadband on Telemedicine: A Study Commissioned by the United States Chamber of Commerce." Working paper, The Advanced Communications Law & Policy Institute at New York Law School. http://www.uschamber.com/sites/default/files/about/0904Broadband_and_Telemedicine.pdf.



¹⁰² Federal Communications Commission. "National Broadband Plan: Connecting America." Accessed January 27, 2012. http://www.broadband.gov/plan/10-healthcare

¹⁰³ *Ibid*.

¹⁰⁴ Ibid.

¹⁰⁵ Castro, Daniel D. 2007. "Improving Health Care: Why a Dose of IT May Be Just What the Doctor Ordered." Working paper, Information Technology and Innovation Foundation.

¹⁰⁶ Davidson, Charles M. and Michael J. Santorelli. 2009. "The Impact of Broadband on Telemedicine: A Study Commissioned by the U.S. Chamber of Commerce." Working paper, The Advanced Communications Law & Policy Institute at New York Law School. http://www.uschamber.com/sites/default/files/about/0904Broadband_and_Telemedicine.pdf.

¹⁰⁷ Low, Sarah and Peter Stenberg. 2009. "Rural Broadband at a Glance." Working paper, United States Department of Agriculture. http://www.ers.usda.gov/publications/eib47/eib47.pdf.

¹⁰⁸ Castro, Daniel D. 2007. "Improving Health Care: Why a Dose of IT May Be Just What the Doctor Ordered." Working paper, Information Technology and Innovation Foundation.

¹⁰⁹ Castro, Daniel D. 2007. "Improving Health Care: Why a Dose of IT May Be Just What the Doctor Ordered." Working paper, Information Technology and Innovation Foundation.

¹¹⁰ Fox, Susannah. 2009. "The Social Life of Health Information: Americans' Pursuit of Health Takes Place within a Widening Network of both Online and Offline Sources." *Pew Research Center: Pew Internet & American Life Project*, June 11. http://www.pewinternet.org/Reports/2009/8-The-Social-Life-of-Health-Information.aspx.

¹¹¹ Ibid.

¹¹² Ackerman, Michael J. 2010. "Telemedicine: Networks in the Service of Healthcare." Presentation, National Library of Medicine. Accessed January 27, 2012. http://courses.mbl.edu/mi/2010/presentations_spring/Telemed.ppt.

- Verizon phones contain a "Pill Phone" option with dosage, facts, side effects, and notifications on more than 1,800 common drugs.
- T-Mobile provides download options for personal healthcare applications, including a Health Tracker that files a consumers health notes.
- AT&T's iPhone allows physicians and patients to view medical images remotely.

It is this access to technology and innovation that will shift the focus from disease treatment towards personal wellness and preventative care, which in turn will improve both individual and country-wide economies.

Though notable adoption progress has been made, there exists a division among the population in terms of Internet use for healthcare services. The Pew study reports that 85 percent of college graduates compared with only 25 percent of those without high school diplomas are e-patients; and 82 percent of adults with household incomes greater than \$75,000 are e-patients compared with 44 percent of those with household income lower than \$30,000.¹¹⁴ The need to expand such service capabilities to reach all households is evident given the potential benefits.

The advancement of healthcare IT may prove remarkably beneficial to consumers and providers alike. Perhaps the most notable improvements to the system are the cost savings from keeping patients and providers local, and the reduction in administrative workloads. Often, especially in less populated regions, it is necessary that patients leave the community to receive healthcare, requiring travel expenses and, in some cases, time taken off work. One study estimates that healthcare IT "could save the U.S. healthcare system \$4.28 billion [annually] from just reducing transfers of patients from one location, such as a nursing home for medical exams at hospitals, physicians' offices, or other caregiver locations."¹¹⁵ In addition to any financial and time burdens, in this case consumer purchasing dollars also go with them to the urban community ¹¹⁶ This scenario is less than ideal for small, rural economies and burdensome for low-income patients. Additionally, because much of healthcare involves generating, processing, and transmitting information, implementing a system of EHRs and other electronic administrative management has potential to substantially reduce costs. ¹¹⁷ Atkinson reports that overall, societal cost savings from implementing and using healthcare IT in the United States has been estimated (by two studies) at approximately \$80 billion per year.

In addition to lowering costs, enhanced healthcare IT allows consumers access to better quality information. Atkinson points out that by providing such information, IT empowers patients to make more sound healthcare decisions. Individuals are able to access not only higher quality information regarding conditions and treatments, but also information pertaining to the quality of services available through different providers.¹¹⁹ Internet health resources enhance information available to patients and consumers by establishing online communities and social networks as knowledge exchange forums. Consumers can share with one another thoughts and experiences, aiding in the decision making process. The Internet also creates an opportunity for individuals to communicate with specialists and experts located anywhere in the world. This lessens the influence of

¹¹⁸ Ibid. ¹¹⁹ Ibid.



¹¹⁴ Ackerman, Michael J. 2010. "Telemedicine: Networks in the Service of Healthcare." Presentation, National Library of Medicine. Accessed January 27, 2012. http://courses.mbl.edu/mi/2010/presentations_spring/Telemed.ppt.

¹¹⁵ Davidson, Charles M. and Michael J. Santorelli. 2009. "The Impact of Broadband on Telemedicine: A Study Commissioned by the U.S. Chamber of Commerce." Working paper, The Advanced Communications Law & Policy Institute at New York Law School. http://www.uschamber.com/sites/default/files/about/0904Broadband_and_Telemedicine.pdf.

¹¹⁶ Ackerman, Michael J. 2010. "Telemedicine: Networks in the Service of Healthcare." Presentation, National Library of Medicine. Accessed January 27, 2012. http://courses.mbl.edu/mi/2010/presentations_spring/Telemed.ppt.

¹¹⁷ Atkinson, Robert D., and Daniel Castro. 2008. "Digital Quality of Life: Understanding the Personal and Social Benefits of the Information Technology Revolution." Working paper, Information Technology and Information Foundation.

geographic proximity on availability and quality of care. A few examples of the cost and health savings associated are outlined below: ¹²⁰

- A telemedicine network in Alaska has provided the ability for doctors to communicate with expecting mothers upwards of 200 miles away, providing prenatal care and assistance to safely deliver babies.
- A 2002 study by the Veterans Association found in-home chronic disease management tools, including teleconsulations and remote monitoring, resulted in "40 percent fewer emergency room visits and a 63 percent reduction in hospital admissions."
- In-home monitoring systems are being tested to detect the early onset of cognitive diseases, including Alzheimer's. It is estimated that the early "interventions that could delay the onset of Alzheimer's disease by as little as one year would reduce prevalence of the disease by 12 million fewer cases in 2050," which could lead to dramatic cost savings for this disease alone.

Internet health resources create opportunities for providers to increase service provision and offer higher quality care. Previously, Atkinson notes, it was generally accepted that, "[i]f you want access to the best healthcare in the world, you should live close to a major world-class research hospital, usually located in an urban area...if you do not live near such a facility, you may not get the best treatment." Atkinson notes "that IT is helping to break down geographical barriers to healthcare by enabling patients to have access to top-quality care without necessarily being physically close to it."121 Often sparsely populated and economically depressed areas lack a full spectrum of healthcare options. Generalists may be unwilling to practice in areas without specialty backup.¹²² A 2005 study furthers this point, finding that only three percent of medical students expressed interest in working in a rural area.¹²³ Telemedicine addresses the issue of doctor or specialist shortages in rural communities with low population densities.¹²⁴ Additionally, the increase in quality and availability of information will aide in reducing medical errors. For example, Atkinson points out that in the United States adverse drug events account for 19 percent of injuries in hospitalized patients and cost hospitals over \$2 billion per year, not including malpractice costs or the costs of injuries to patients. One study found that health IT, including such elements as e-prescribing, could improve physician accuracy and eliminate around 200,000 adverse drug events in the U.S., resulting in an annual savings of \$1 billion.125

Broadband will play an invaluable role in transforming healthcare for seniors and people with disabilities. Seniors currently account for 12 percent of the population and are set to double by 2050.¹²⁶ Additionally, there are currently over 50 million people with disabilities in the U.S., many of whom are older. Healthcare costs are going to drastically increase in the coming years, and broadband can help manage and contain these costs. It is estimated that broadband-based health resources can save around \$927 billion in healthcare costs for seniors and people with disabilities between 2005 and 2030.¹²⁷ Services such as in-home monitoring will allow seniors to stay in place,

 ¹²⁶ Davidson, Charles M. and Michael J. Santorelli. 2009. "The Impact of Broadband on Telemedicine: A Study Commissioned by the U.S. Chamber of Commerce." Working paper, The Advanced Communications Law & Policy Institute at New York Law School. http://www.uschamber.com/sites/default/files/about/0904Broadband_and_Telemedicine.pdf.
 ¹²⁷ Ibid



¹²⁰ Davidson, Charles M. and Michael J. Santorelli. 2009. "The Impact of Broadband on Telemedicine: A Study Commissioned by the U.S. Chamber of Commerce." Working paper, The Advanced Communications Law & Policy Institute at New York Law School. http://www.uschamber.com/sites/default/files/about/0904Broadband_and_Telemedicine.pdf.

¹²¹ Atkinson, Robert D., and Daniel Castro. 2008. "Digital Quality of Life: Understanding the Personal and Social Benefits of the Information Technology Revolution." Working paper, Information Technology and Information Foundation.

¹²² Ackerman, Michael J. 2010. "Telemedicine: Networks in the Service of Healthcare." Presentation, National Library of Medicine. Accessed January 27, 2012. http://courses.mbl.edu/mi/2010/presentations_spring/Telemed.ppt.

¹²³ Davidson, Charles M. and Michael J. Santorelli. 2009. "The Impact of Broadband on Telemedicine: A Study Commissioned by the U.S. Chamber of Commerce." Working paper, The Advanced Communications Law & Policy Institute at New York Law School. http://www.uschamber.com/sites/default/files/about/0904Broadband and Telemedicine.pdf.

¹²⁴ Low, Sarah and Peter Stenberg. 2009. "Rural Broadband at a Glance." Working paper, United States Department of Agriculture. http://www.ers.usda.gov/publications/eib47/eib47.pdf.

¹²⁵ Atkinson, Robert D., and Daniel Castro. 2008. "Digital Quality of Life: Understanding the Personal and Social Benefits of the Information Technology Revolution." Working paper, Information Technology and Information Foundation.

stay connected, and take advantage of lifesaving applications, while decreasing dependency on nursing homes and care givers. In addition to providing access to health Web sites, broadband "facilitates efforts by seniors and people with disabilities to stay in touch with family, friends, and community, and to participate in an array of activities, all of which may decrease debilitating symptoms of depression and sustain mental acuity."¹²⁸ Many seniors are working past retirement and many of these jobs require computer skills. Additionally, seniors are increasingly enrolling in computer courses and Internet training classes, with top Internet uses being searching for health information and keeping in touch with loved ones. This sets the stage for successfully bringing more seniors into broadband. In fact, the largest increase in broadband adoption from 2005-2008 fell in the 65+ age group.¹²⁹

Implementing a new, widely adopted healthcare IT system has the potential to create substantial consumer surplus and save the country billions of dollars. Much of the estimated national savings comes from increases in efficiency including shorter hospital stays and less frequent visits, improved nurse productivity, and more efficient drug utilization. EHRs provide doctors with more complete information about their patients, which reduces the need for duplicative and unnecessary medical tests.¹³⁰ Patients would be equipped with access to more information aiding them in managing their health and making more informed decisions. Atkinson explains, "by increasing patients' access to their own medical records and to a plethora of information to help patients make better decisions, the Internet has the potential to improve healthcare."¹³¹ IBM has observed that in making consumers more directly accountable for their own health and healthcare choices, "they can also become wiser, more value-based purchasers, improving their health through better choices, and at the same time exert pressure to keep system costs in line."¹³² As stated in the National Broadband Plan, "[t]hese [broadband-enabled] solutions...offer the potential to improve healthcare outcomes while simultaneously controlling costs and extending the reach of the limited pool of healthcare professionals. Furthermore, as a major area of innovation and entrepreneurial activity, the health IT industry can serve as an engine for job creation and global competitiveness."¹³³ One study, for example, estimates that an investment of \$10 billion in health IT in one year has the ability to create or retain 212,000 U.S. jobs for a year.¹³⁴ The National Broadband Plan aims to improve quality of life by providing better healthcare at lower costs, to a greater portion of the population, while simultaneously creating jobs to aid in recovery.

Many communities lack sufficient numbers of primary care clinicians.¹³⁵ Limited research suggests that family physicians, nurse practitioners, and physician assistants are especially likely to practice in rural communities.^{136,137,138} They are also more likely to care for low-income patients.^{139,140} Other

¹³⁹ Moody, NB, PL Smith, and LL Glenn. 1999. "Client Characteristics and Practice Patterns of Nurse Practitioners and Physicians." Nurse Practitioners 24: 94–96, 99–100, 102–103.



¹²⁸ Davidson, Charles M. and Michael J. Santorelli. 2009. "The Impact of Broadband on Telemedicine: A Study Commissioned by the U.S. Chamber of Commerce." Working paper, The Advanced Communications Law & Policy Institute at New York Law School. http://www.uschamber.com/sites/default/files/about/0904Broadband_and_Telemedicine.pdf.

¹²⁹ *Ibid.*

¹³⁰ Castro, Daniel D. 2007. "Improving Health Care: Why a Dose of IT May Be Just What the Doctor Ordered." Working paper, Information Technology and Innovation Foundation.

¹³¹ Atkinson, Robert D., and Daniel Castro. 2008. "Digital Quality of Life: Understanding the Personal and Social Benefits of the Information Technology Revolution." Working paper, Information Technology and Information Foundation.

¹³² Davidson, Charles M. and Michael J. Santorelli. 2009. "The Impact of Broadband on Telemedicine: A Study Commissioned by the U.S. Chamber of Commerce." Working paper, The Advanced Communications Law & Policy Institute at New York Law School. http://www.uschamber.com/sites/default/files/about/0904Broadband and Telemedicine.pdf.

¹³³ Federal Communications Commission. 2010. "Connecting America: The National Broadband Plan." http://www.broadband.gov/plan/10-healthcare/.

¹³⁴ Davidson, Charles M. and Michael J. Santorelli. 2009. "The Impact of Broadband on Telemedicine: A Study Commissioned by the U.S. Chamber of Commerce." Working paper, The Advanced Communications Law & Policy Institute at New York Law School. http://www.uschamber.com/sites/default/files/about/0904Broadband_and_Telemedicine.pdf.

¹³⁵ Council on Graduate Medical Education. 1998. "Tenth Report: Physician Distribution and Health Care Challenges in Rural and Inner-City Areas." Washington DC: Government Printing Office.

¹³⁶ Ibid.

¹³⁷ Fryer GE, LA Green, SM Dovey, and RI Phillips Jr. 2001. "The United States Relies on Family Physicians Unlike any other Specialty." *American Family Physician* 63:1669–1703.

¹³⁸ Larson, EH, LG Hart, and J Hummel. 1994. "Rural Physician Assistants: a Survey of Graduates of MEDEX Northwest." *Public Health Reports* 266–274.
studies restricted to analyses of physician distribution have indicated that family physicians are more likely than other primary care physicians to work in rural and underserved communities.^{141,142} The conclusions that can be drawn from the available research indicate that non-physician primary care clinicians and family physicians have a greater propensity to care for underserved populations than do primary care physicians in other specialties.¹⁴³ No study, however, has comprehensively compared the geographic distribution and patient populations of clinicians across the different primary care disciplines. But these are the very individuals that would benefit the most from health IT applications.

The Office of the National Coordinator for Health IT (ONC) supports two national health IT adoption surveys: one of physician offices and one of hospitals.

Summarized in Table 6, the surveys:

- Assess the current state of health IT adoption;
- Specify measurable goals and methods for evaluating strategies; and,
- Determine approaches that can accelerate health IT adoption in a cost-effective manner

Setting	2006	2007	2008	2009*
Physicians offices (basic**)	11%	13%	17%	21%
Physicians offices (full**)	3%	4%	4%	6%
Hospitals (basic**)	N/A	N/A	8%	N/A
Hospitals (full**)	N/A	N/A	2%	N/A

Table 6: Adoption Rates¹⁴⁴

*2009 statistics are preliminary.

Efforts to quantify health IT benefits are underdeveloped. Researchers have tried to measure the success, but many "focus on only one metric of their success – such as user acceptance, economic benefits, usefulness, or improvement in patient safety – rather than conducting a comprehensive evaluation."¹⁴⁵ One of the primary sources of data that will be used in our assessment of the impact of broadband on healthcare is the American Hospital Associations (AHA) annual survey. This survey effort has been in annual circulation since 1980 and is designed to assess many factors at over 6,500 hospitals nationwide. The survey covers over 1,000 data items, which touch on areas such as organization of hospitals, facilities and services, community benefits the hospitals may fulfill, and general facility items (e.g., bed count, utilization, revenue).

Our evaluation study will use Quarterly / Annual PPRs collected by NTIA from grantees and match that data to the AHA data. This will provide a sample of hospitals that are affected by BTOP grant awards. An additional sample of hospitals that are deemed statistically similar (based on size,

¹⁴⁵ Atkinson, Robert D., and Daniel Castro. 2008. "Digital Quality of Life: Understanding the Personal and Social Benefits of the Information Technology Revolution." Working paper, Information Technology and Information Foundation.



 ¹⁴⁰ Mainous, Ag, JG Bertolino, and PL Harrell. 1992. "Physician Extenders: Who is Using Them?" Family Medicine 24:201–204.
¹⁴¹ Council on Graduate Medical Education. 1998. Tenth Report: Physician Distribution and Health Care Challenges in Rural and

Inner-City Areas. Washington DC: Government Printing Office.

¹⁴² Burnett, WH, DH Mark, JE Midtling, and BB Zellner. 1995. "Primary Care Physicians in Underserved Areas: Family Physicians Dominate." Western Journal of Medicine 163:532–536.

¹⁴³ Grumbach, Kevin, Gary L. Hart, Elizabeth Mertz, Janet Coffman, and Lorella Palazzo. 2003. "Who is Caring for the Underserved? A Comparison of Primary Care Physicians and Nonphysician Clinicians in California and Washington." *Annals of Family Medicine* 1, no.2 (July): 97-104. http://www.ncbi.nlm.nih.gov/pmc/articles/PMC1466573/.

¹⁴⁴ The Office of the National Coordinator for Health Information Technology. 2010. "Health IT adoption." Last modified May 25. http://healthit.hhs.gov/portal/server.pt?open=512&objID=1152&parentname=CommunityPage&parentid=28&mode=2&in_hi_u serid=11113&cached=true.

demographics, surrounding area's population) will be drawn that have not been affected by BTOP grant awards, and the difference/similarities will be statistically analyzed.

Based on this review of the literature and the availability of data from the AHA survey described above, the following measures could be analyzed at the anchor institution level to measure the healthcare-related impacts of BTOP grants:

- Broadband technology infrastructure available to healthcare organizations and practitioners
- Access to Health IT care products and services
- Type of Health IT applications sought
- Type of Health IT applications utilized
- Notable changes in the quality of healthcare services before and after introduction of broadband service/improvements
- Notable changes in the cost of healthcare services before and after introduction of broadband service/improvements

More specifically, looking at the AHA survey in terms of E-Care, Electronic Health Records, Telehealth, and Mobile Health, we will investigate causality or correlation through statistical analysis of the following statistics and BTOP funding:

- Availability of E-Care:
 - Home health services
 - Electronic consultant reports
 - Electronic laboratory reports
 - Electronic radiology reports
 - Electronic radiology images
 - Electronic diagnostic test results
 - Electronic diagnostic test images
 - Sharing of electronic patient-level clinical data through an electronic health information exchange or a regional health information organization
 - Level of participation in a regional health information exchange or regional health information organization
 - Electronic exchange of patient data (e.g., patient demographics, clinical care record, laboratory results, medication history, radiology reports)
 - Electronic checks of insurance eligibility
 - Electronic claims submission to both public and private payers
 - Use of electronic reminders to patients for pre-admission and/or follow-up care
 - Electronic capture of patient consents or authorizations
 - Capability to provide patients with an electronic copy of their health record
- Availability of Electronic Health Record:
 - EHRs
 - Electronic patient demographics



- Electronic physician notes
- Electronic nursing notes
- Electronic problem lists
- Electronic medication lists
- Electronic discharge summaries
- Electronic advanced directives
- Electronic laboratory reports
- Electronic radiology reports
- Electronic radiology images
- Electronic diagnostic test results
- Electronic diagnostic test images
- Availability of Telehealth:
 - Electronic clinical guidelines
 - Electronic clinical reminders
 - Electronic drug allergy alerts
 - Electronic drug-drug interaction alerts
 - Electronic drug-lab interaction alerts
 - Electronic drug dosing support
 - Telemedicine
- Availability of Mobile Health:
 - Mobile health devices
 - Physician use of personal data assistant

C.5 Measuring Impacts on Public Safety

For the purposes of this project, we have defined public safety agencies to include law enforcement agencies, fire safety agencies, and emergency medical services. The aim of this review is to provide an overview of the current impact and future impact potential of broadband ICT on law enforcement, public safety, and emergency medical services. Broadband technology can potentially affect public safety in two ways. First, dedicated broadband network resources have the potential to improve communication between disparate public safety institutions. Several of the BTOP grants, particularly the 700MHz grants, address this type of initiative. These grants are not included in the scope of this study. A second benefit of broadband ICT on public safety occurs when public safety institutions are connected to broadband communications and community members. We investigate the extent of these benefits as part of this study and present more information on this type of benefit below.

The presence of broadband and advanced IT systems amplifies the protection officials are able to provide. In a recent report, Atkinson explains, "by using IT, a police force can be more productive, solve more crimes, and better protect the lives and property of the public."¹⁴⁶ Law enforcement

¹⁴⁶ Atkinson, Robert D., and Daniel Castro. 2008. "Digital Quality of Life: Understanding the Personal and Social Benefits of the Information Technology Revolution." Working paper, Information Technology and Information Foundation.



agencies need public safety IT to understand the community and the nature of criminal activity within the jurisdiction. As technology advances, systems designed to enhance public safety are becoming more accurate and sophisticated. For example, Atkinson points out, "GPS devices can similarly be used to enforce restraining orders... if domestic violence offenders violate restrictions on visiting their spouse at their home or place of work, the police will be notified in real time."¹⁴⁷ Such systems not only improve the efficiency of law enforcement, but also provide comfort and reassurance to citizens. Similar advanced IT systems exist which notify victims of an offender's release or escape from custody, enabling affected parties to take proper precautions.

In addition to IT enabling greater protection services, it also works to prevent and prepare communities for instances of accidents and crime. Atkinson mentions, "[g]overnments and nongovernmental organizations are using IT to predict, respond to, and manage accidents at dangerous facilities, as well as natural phenomena such as hurricanes, wildfires, tsunamis, and landslides." He explains that through satellite and aerial images, as well as first hand ground inspections, organizations can locate at-risk populations and unstable locations and determine how to respond after or amidst a disaster.¹⁴⁹ Broadband and public safety IT can also boost efficiency through cost and time savings. Advanced technologies "automate many of the time-consuming tasks associated with police work and free police resources to be used on other effective programs." At the most basic level, efficiency gains are realized as less time needs to be devoted to paper work and administrative tasks. A more complex example, Atkinson, points out, "[u]sing GPS devices for remote monitoring can reduce expenses and improve outcomes for the criminal justice system.... [r]ather than being put in jail, some low-risk offenders can be required to stay at home...incarceration can cost a state approximately \$25,000 per inmate per year, whereas the cost of remotely monitoring an inmate can cost less than \$5,000 per year."¹⁵⁰ By incorporating advanced technologies, public safety agencies are able to devote more efficient use of their time and resources.

Developing broadband and IT systems improves information sharing between citizens and public safety entities. Atkinson notes one of the primary benefits of advanced mobile technologies in combating crime, "with so many cell phones now equipped with digital cameras, when a crime occurs witnesses are more likely to be able to provide digital evidence."¹⁵¹ It therefore is important that both the public and law enforcement agencies have access to, and the means of integrating, such information. Additionally, IT systems enhance the transparency of public safety agencies. Atkinson explains, "[p]olice departments that have computerized their incident and arrest reports can more easily share their data with nonprofit organizations and journalists. Using these data, researchers can monitor police performance and analyze the data for evidence of impropriety."¹⁵² Greater transparency implies more efficient agency operations.

The adoption of broadband and advanced IT systems has notable benefits for the country's public safety. However, according to the NTIA there are significant barriers to such realization. They state, "the public safety community must overcome a number of practical hurdles, ranging from the technical to the financial and regulatory."¹⁵³ To improve public safety it is necessary that all areas of the country have access to broadband services. NTIA emphasized, "[f]ederal agencies will require ubiquitous broadband coverage to support a variety of voice, video, and data requirements. In many instances, these mission-critical capabilities will be needed in regions that are unlikely to be served by commercial providers."¹⁵⁴ As a result, government must directly support or create the

¹⁵⁴ Ibid.



¹⁴⁷ Atkinson, Robert D., and Daniel Castro. 2008. "Digital Quality of Life: Understanding the Personal and Social Benefits of the Information Technology Revolution." Working paper, Information Technology and Information Foundation.

¹⁴⁸ *Ibid.*

¹⁴⁹ *Ibid.*

¹⁵⁰ Atkinson, Robert D., and Daniel Castro. 2008. "Digital Quality of Life: Understanding the Personal and Social Benefits of the Information Technology Revolution." Working paper, Information Technology and Information Foundation.

¹⁵¹ *Ibid.*

¹⁵² *Ibid.*

¹⁵³ National Telecommunications and Information Administration. 2009. "Executive Branch Views on Public Safety, Homeland Security and Cybersecurity Elements of a National Broadband Plan." United States Department of Commerce.

appropriate incentives for the deployment of services and communication infrastructure that public safety agencies require.

Finally, the National Broadband Plan seeks to increase availability, adoption, and the use of broadband for "national purposes, such as economic development, healthcare, education, public safety, and government transparency. Public safety is also included in two of the long-term goals of the National Broadband Plan:¹⁵⁵

- Every American community should have affordable access to at least one gigabit per second broadband service to anchor institutions such as schools, hospitals, and government buildings.
- To ensure the safety of the American people, every first responder should have access to a nationwide, wireless, interoperable broadband public safety network.

Potential sources of data that could be used in our assessment of the impact of broadband on public safety are: Crime & Justice Electronic Data (Bureau of Justice Statistics), Law Enforcement Management and Administrative Statistics (LEMA), and the National Incident-Based Reporting System (NIBRS).^{156,157,158}

The Bureau of Justice Statistics reports are available from 1977 – 2008 and are collected on a yearly basis. It is aggregated data from a wide variety of published sources and intended for analytic uses. These measures cover topics such as crime, justice, and socio-demographics. These data are available nationwide at the state level and more granularly for specific areas of the country. Given the level of analysis of these data, we feel it could be best used while performing certain case studies.

LEMA is available every 3-4 years from 1987 – 2007 and collects data from over 3,000 state and local law enforcement agencies, including all those that employ 100 or more sworn officers and a nationally representative sample of smaller agencies. Data are obtained on the organization and administration of police and sheriffs' departments, including agency responsibilities, operating expenditures, job functions of sworn and civilian employees, officer salaries and special pay, demographic characteristics of officers, weapons and armor policies, education and training requirements, computers and information systems, vehicles, special units, and community policing activities. Specific items of interest are:

- Number of computers in the field
- Number of vehicle mounted computers
- Number of other computers
- Digital fingerprints used
- Digital mug shots used
- Digital suspect composites used, etc

The evaluation study could track these data in areas known to be impacted by BTOP compared to statistically similar areas that are not impacted by BTOP. This could provide a measure of

http://www.icpsr.umich.edu/icpsrweb/ICPSR/series/00128/studies?archive=ICPSR&sortBy=7.



¹⁵⁵ Federal Communications Commission. 2010. "Connecting America: The National Broadband Plan." http://www.broadband.gov/plan/10-healthcare/.

¹⁵⁶ Bureau of Justice Statistics. *Crime & Justice Electronic Data*, 1977-2008. http://bjs.ojp.usdoj.gov/content/dtdata.cfm#index. Last modified January 27, 2012.

¹⁵⁷ Bureau of Justice Statistics. *Law Enforcement Management and Administrative Statistics*: 1987, 1990, 1993, 1997, 1999, 2000, 2003, 2007. Last modified April 19, 2010. http://bjs.ojp.usdoj.gov/index.cfm?ty=dcdetail&iid=248#Documentation.

¹⁵⁸ Federal Bureau of Investigation. "National Incident-Based Reporting System, 1991 – 2008." Uniform Crime Reporting Program. Accessed January 27, 2012.

digitalization of law enforcement agencies. Given the level of these data, it could potentially be a viable measure of the impact of BTOP on public safety via law enforcement agencies.

NIBRS is an incident-based reporting system for crimes known to the police. For each crime incident coming to the attention of law enforcement, variety of data are collected about the incident. These data include:

- The nature and types of specific offenses in the incident
- Characteristics of the victim(s) and offender(s)
- Types and value of property stolen and recovered
- Characteristics of persons arrested in connection with a crime incident

These data are available from 1991 – 2008 and published yearly at the city level. The evaluation study could track these data in areas known to be impacted by BTOP compared to statistically similar areas that are not impacted by BTOP. This could provide a measure of criminal activity. Given the level of these data, it could potentially be a viable measure of the impact of BTOP on public safety via crime rates.



C.6 Measuring Impacts Quality of Life/Standard of Living and Civic Engagement

As discussed in their paper, *The Social Impact of Broadband Household Internet Access*, Anderson and Raban point out the relative lack of research on the social impacts of broadband as compared to the economic impacts.¹⁵⁹ Anderson and Raban note that, "[i]ndeed recent authors have noted that few academic publications focus on the impact of broadband on social and personal issues in contrast to the developmental and macro-economic issues."¹⁶⁰ Early studies on impacts of broadband computer networks on society mainly focus on offering expert opinions, forecasting future trends, and speculating about the potential of the technology. More recent studies attempted to assess the penetration rate of broadband by focusing on specific types of applications requiring high speed connections.^{161,162,163,164} These studies also attempted to quantify the subscriber volume, which stands for the number and percent of customers purchasing the service.¹⁶⁵

Moreover, research has found that it is difficult to discern social outcomes specifically attributable to broadband (as opposed to narrowband Internet use). Beyond this, when actual outcomes have been identified, some research has shown that the outcome or change is correlated more to the previous behavior of the individual, as opposed to the switch from narrow-band to broadband.¹⁶⁶ Anderson and Raban, for example, looked at Internet time, TV time, amount of money spent online, and out-of-home social leisure activities as a result of a switch to broadband, and found that the greatest effect on time spent online is not moving to broadband, although this was significant, but the previous behavior of an individual. Similarly, regarding the amount of money individuals spend online. Anderson and Raban found, "a steady progression of online spend driven largely by experience not only in terms of years spent online but in terms of breadth of Internet use."167 Similarly, with respect to time spent watching TV, Anderson and Raban found that, contrary to previous findings using longitudinal data that getting household Internet access had a negative effect on television use, switching from narrowband to broadband Internet at home did not have a similar effect.¹⁶⁸ Their results, instead, demonstrated the resilience of time spent watching TV to a range of life transitions including switching to broadband. Anderson and Raban's research was based on surveys conducted in 2001 and 2002 and, as such, represent results gleaned from very early adopters of broadband. More recent adopters may exhibit different behavioral change.¹⁶⁹

Other studies that have attempted to measure social impacts related to broadband use have noted difficulties in identifying appropriate indicators and/or data sources to support those indicators. When discussing IT productivity in terms of broadband at home, Gillett, Lehr, Osorio, and Sirbu, for example, note that, "home-based access may improve quality of life, for example by enabling more participation in community and civic activities, making a locale more attractive to potential

¹⁶⁹ Anderson, Ben, and Yoel Raban. 2005. "The Social Impact of Broadband Household Internet Access." Chimera Working Paper 2005-2006. Colchester, University of Essex.



¹⁵⁹ Anderson, Ben., and Yoel Raban. 2005. "The Social Impact of Broadband Household Internet Access." Chimera Working Paper 2005-2006, Colchester, University of Essex.

¹⁶⁰ Ibid.

¹⁶¹ Chang, S., Lee, H., and C. Middleton. 2004. "The Deployment of Broadband Internet in Australia: Areas for Attention and Implications from Canada and Korea." Retrieved from www.ryerson.ca/~cmiddlet/pubs/bband_aus.pdf.

¹⁶² Cohill, A.M. 2005. "Community Web Portals and Economic Development". Presented at the 9th Annual Rural Telecommunications Congress Annual Meeting, Lexington, Kentucky, October 9-12.

¹⁶³ Cohill, A.M. 2005. "Broadband Policy for Rural Communities." Presented to the Canadian Government Telecommunications Policy Group, Whitehorse, Yukon Territory, Canada, September 9.

¹⁶⁴ Lee, H., O'Keefe, R., and K. Yun. 2003. "The Growth of Broadband and Electronic Commerce in South Korea: Contributing Factors." *The Information Society*, 19(1): 81-93.

¹⁶⁵ Chang, S., Lee, H., and C. Middleton. 2004. "The Deployment of Broadband Internet in Australia: Areas for Attention and Implications from Canada and Korea." Retrieved from www.ryerson.ca/~cmiddlet/pubs/bband_aus.pdf.

¹⁶⁶ Anderson, Ben, and Yoel Raban. 2005. "The Social Impact of Broadband Household Internet Access." Chimera Working Paper 2005-2006, Colchester, University of Essex.

¹⁶⁷ Ibid.

¹⁶⁸ Ibid.

residents."¹⁷⁰ However, Gillett, Lehr, Osorio, and Sirbu found that most of these hypothesized impacts of home-based Internet are not directly measurable, so they focused instead on how broadband is likely to change other indicators that describe local economies, including community participation and quality of life as measured by voting participation, mortality rates, or local prices.¹⁷¹ However, as Gillett, Lehr, Osorio, and Sirbu found, the ability to test some of these indicators is limited by the collection frequency for different types of Census data, and geographic unit limitations for other types of data (for example, voting participation is not tallied by ZIP Code).¹⁷² In addition, for most indicators, it is reasonable to expect that broadband's impacts will be felt only after some time lag – broadband has to be not only available, but adopted, and then used.

Others have also noted difficulties in measuring quality of life impacts. In "How the Public Perceives Community Information Systems," the Pew Internet and American Life Project and the Monitor Institute noted that several of the indicators for measuring a citizens' sense of how their community information system is performing and their overall satisfaction with their community are difficult to measure and assess independently without complicated and expensive methodologies.¹⁷³ Specific areas noted as difficult to measure, among others, were the effectiveness of technology programs at schools, the availability of "quality of life" information from community organizations, and "effective opportunities" for citizens to have their voices heard.¹⁷⁴

Despite this, several studies have attempted to develop guality of life or standard of living indicators to measure broadband's social impacts. In their article, "Closing the Rural Broadband Gap" LaRose, Gregg, Strover, Straubhaar and Inagaki note that, "[b]roadband development may produce social benefits as well as economic ones."¹⁷⁵ They summarize the literature addressing the impacts of rural broadband networks, stating that, "[r]ural broadband networks could improve some of the conditions of rural life that lead to depopulation and despair, including access to healthcare and education."¹⁷⁶ In their survey and case study research, LaRoase, Gregg, Strover, Straubhaar, and Inagaki look at community attachment, perceived social support, relocations intentions, number of voluntary members, and Internet self-efficacy as measures of broadband's social impacts in rural communities.¹⁷⁷ They find that the precursors of broadband adoption were the perceived benefits of high speed Internet connections, the ability to experience those benefits for oneself, and a sense of efficacy when using the Internet, noting that, "Ithese are factors amendable to community-based, self development interventions that can close the broadband gap despite the challenging demographics of rural communities." LaRose, Gregg, Strover, Straubhaar and, Inagaki also found that, "social uses of the Internet increased the social support experienced by rural residents, leading to higher levels of community satisfaction and attachment, and ultimately lower intentions to relocate away from rural communities."¹⁷⁸ Those impacts were counterbalanced, however, by the development of social connections and interests beyond the local community increasing intentions to relocate. This led to the conclusion that the development of local web content and a focus on local social networks, "is important to sustain rural populations."¹⁷⁹

¹⁷⁹ *Ibid.*



¹⁷⁰ Gillett, Sharron, William Lehr, Carlos Osorio, and Marvin Sirbu. 2006. "Measuring Broadband's Economic Impact." U.S. Department of Commerce, Economic Development Administration.

 $http://web.si.umich.edu/tprc/papers/2005/475/TPRC2005_Gillett\%20Lehr\%20Sirbu\%20Osorio\%20submitted.pdf. \end{tabular} \label{eq:linear} If the second secon$

¹⁷² *Ibid.*

¹⁷³ Rainie, Lee, Kristen Purcell, Tony Siesfeld, and Mayur Patel. 2011. "How the Public Perceives Community Information Systems." *Pew Internet and American Life Project*, March 11. http://pewinternet.org/Reports/2011/Community-Information-Systems.aspx.

¹⁷⁴ Ibid.

¹⁷⁵ LaRose, Robert, Jennifer L. Gregg, Sharon Stover, Joseph Straubhaar, and Nobuya Inagaki. 2008. "Closing the Rural Broadband Gap." Final Technical Report, Michigan State University. Report published November 30. https://www.msu.edu/~larose/ruralbb/.

¹⁷⁶ *Ibid.*

¹⁷⁷ LaRose, Robert, Jennifer L. Gregg, Sharon Stover, Joseph Straubhaar, and Nobuya Inagaki. 2008. "Closing the Rural Broadband Gap." Final Technical Report, Michigan State University. Report published November 30. https://www.msu.edu/~larose/ruralbb/.

¹⁷⁸ *Ibid.*

Other case study research has tried to identify and measure, where possible, the social transformational changes that can be attributed to the uptake of broadband technologies within communities. In their case study of the Shippagan area, a rural area on the Acadian Peninsula in North-eastern New Brunswick Canada, Sid-Ahmed Selouani, and Habib Hamam surveyed individuals throughout the Shippangan area, which had obtained broadband access at different times over the six years prior to their survey.¹⁸⁰ In terms of transformational changes, Selouani and Hamam looked at use of communications and informational tools; changes in leisure time activities and community involvement, such as time together as a family and participation in community meetings or activities; mobility and migration; changes in uses of the Internet at home; changes in online purchases; establishment and/or expansion of home-based businesses; and the financial impacts on home based businesses.¹⁸

In summary, Selouani and Hamam found in terms of transformational change and community impacts that the introduction of broadband services to households does not diminish the time that the family spends together, allows for daily online banking, and increases the household purchases of electronic devices.¹⁸² Moreover, household users reported positive impacts, generally in the areas of:

- Facilitating communication with friends •
- Making education projects easier
- Improving the quality of life since it allows the payment of bills from home

Selouani and Hamam note that the identified positive changes are closely related to the broadband connection as opposed to access to the Internet in general, as many of the tasks are very difficult or quite impossible to do with a dial-up connection.¹⁸³ In addition, respondents' indication that in the near future a proportion of surveyed persons plan to use their high-speed connection to get access to e-learning, to find information about real estate values and purchasing, stock trading/tracking, investment purchases, and local government services.

Volunteerism is another quality of life indicator utilized by some researchers in looking at the social impacts the Internet in general. The Pew Research Center's Internet & American Life Project has found that 75 percent of all American adults are active in some kind of voluntary group or organization and Internet users are more likely than others to be active: 80 percent of Internet users participate in groups, compared with 56 percent of non-Internet users.¹⁸⁴ The Pew survey found that use of the Internet is having a wide-ranging impact on users' engagement with civic, social, and religious groups. Moreover, the Pew research found that Internet users are more active participants in their groups than other adults, and are more likely to feel pride and a sense of accomplishment.

Some research has found that found that broadband users are sometimes less satisfied than others with community life, which raises the possibility that upgrades in local information systems might produce more critical, activist citizens. Research suggests that social media like Facebook and Twitter are emerging as key parts of the civic landscape and mobile connectivity is beginning to affect people's interactions with civic life.¹⁸⁵

¹⁸⁵ Rainie, Lee, Kristen Purcell, Tony Siesfeld, and Mayur Patel. 2011. "How the Public Perceives Community Information Systems." Pew Internet and American Life Project, March 11. http://pewinternet.org/Reports/2011/Community-Information-Systems.aspx.



¹⁸⁰ Habib, Haman, and Sid-Ahmed Selouani. 2007. "Social Impact of Broadband Internet: A Case Study in the Shippagan Area, a Rural Zone in Atlantic Canada." Journal of Information, Information Technology, and Organizations, Vol. 2: 79-94.

¹⁸¹ *Ibid.* ¹⁸² *Ibid.*

¹⁸³ Ibid.

¹⁸⁴ Rainie, Lee, Kristen Purcell, and Aaron Smith. 2011. "The Social Side of the Internet." Pew Internet and American Life Project, January 18. http://pewinternet.org/Reports/2011/The-Social-Side-of-the-Internet.aspx.

Based on a review of the literature, potential measures for quality of life impacts of broadband could include:

- Civic engagement, including voter registration and/or participation and membership in civic associations
- Homelessness
- Volunteerism
- Telecommuting rates
- Establishment of home businesses
- Home schooling rates
- Time spent online
- Internet purchases

Potential sources of data that could be used to support some of these measures are the CPS and ACS. $^{\rm 186,187}$

The CPS is a monthly survey of about 50,000 households conducted by the Bureau of the Census for the Bureau of Labor Statistics. This survey covers Employment, Unemployment, Earnings, Educational Attainment, Income, Poverty, Health Insurance coverage, Job Experience and Tenure, School Enrollment, Voting and Registration, Computer Usage, and Internet Usage. These data are available at the county level, and there are four specific supplements to this monthly survey that would be of particular value to the study:

- 1. Schooling and Computer / Internet usage supplement (1998, 2000, 2001, 2003, 2005, 2010, 2011) a nationwide supplement to the CPS, which covers items such as:
 - a. (Do you/Does anyone in this household) use the Internet at any location?
 - b. (Do you/Does anyone in this household) connect to the Internet from home?
 - c. Do you currently access the Internet at home using...?
 - d. What is the main reason that you do not have high-speed (that is, faster than dialup) Internet access at home?
- 2. Voter registration and participation supplement (1996, 1998, 2000, 2002, 2004, 2006, 2008) a nationwide supplement to the CPS, which covers items such as:
 - a. Were you registered?
 - b. Did you vote?
 - c. Main reason not registered?
 - d. Main reason did not vote?
- 3. Membership in civic associations supplement (2008, 2009, 2010, 2011) a nationwide supplement to the CPS, which covers items such as:
 - a. Contacted a public official?
 - b. Bought or boycotted based on political views?
 - c. Type of group joined?

 ¹⁸⁷ United States Census Bureau. The American Community Survey, 2005-2009. Accessed January 27, 2012. http://dataferrett.census.gov/.



¹⁸⁶ United States Census Bureau. *Current Population Survey*, 1996 – 2010 Accessed January 27, 2012. http://dataferrett.census.gov/.

- d. How often did you communicate via email or internet?
- 4. Volunteer Workers Supplement (2002-2011) a nationwide supplement to the CPS, which covers items such as:
 - a. Volunteer activities for an organizations, including time spent, types of organizations, and types of volunteer activities, and location of volunteer activities (state)
 - b. Volunteer activities for children's school or youth organizations, including time spent and types of volunteer activities and location of volunteer activities (state)
 - c. Charitable donations
- 5. Telecommuting rates (1997, 2001, 2004) a nationwide supplement to the CPS, which covers items such as:
 - a. Type of telecommuting, e.g. employer-sponsored telework vs. home business telework
 - b. Frequency per week (days and hours) of telework
 - c. Computer use while working from home

The ACS is a nationwide survey designed to provide communities a fresh look at how they are changing. It is a critical element in the Census Bureau's decennial census program. The ACS collects information such as age, race, income, commute time to work, home value, veteran status, and other important data. These data are available yearly at the county level, and a specific point of interest to the evaluation study is the "Means of Transportation to Work" section, which contains a question that address "working from home."

These data will be tracked in areas known to be impacted by BTOP and compared to statistically similar areas that are not impacted by BTOP. Correlation and causality of these measures will be tested with respect to BTOP funding.

For identified measures that are not supported through these two data sources, we will attempt to collect data from state and local level government sources for particular geographies, perhaps those associated with our case study locations. To the extent possible, statistically significant causation or correlation findings for these limited geographic-specific instances will be extrapolated to other similar areas or nationwide, as appropriate.



Appendix D. Glossary

- ACS American Community Survey
- ACTION Access to Computer Technology and Instruction in Online Networking
- AHA American Hospital Association
- ARE-ON Arkansas Research & Education Optical Network
- ASR ASR Analytics, LLC
- ATOM Arkansas Telehealth Oversight & Management
- BAA Broadband Awareness and Adoption
- BTOP Broadband Technology Opportunities Program
- CAI Community Anchor Institutions
- CCI Comprehensive Community Infrastructure
- CETF California Emerging Technology Fund
- CPS Current Population Survey
- EDA Economic Development Administration
- EHR Electronic health records
- ERS Economic Research Service
- ESOL English as a Second Language
- FCC Federal Communication Commission
- FCCC Foundation for California Community Colleges
- GDP Gross domestic product
- GPUAC Greater Philadelphia Urban Affairs Coalition
- ICN Illinois Century Network
- ICT Information communication technology
- IPEDS Integrated Postsecondary Education Data System
- IT Information Technology
- LEMA Law Enforcement Management and Administrative Statistics
- LTT Long-term trend
- MESA Mathematics, Engineering, Science Achievement
- NAEP National Assessment of Educational Progress



- NBM National Broadband Map
- NIBRS National Incident–Based Reporting System
- NIU Northern Illinois University
- NOFA Notice of Funds Availability
- NTIA National Telecommunications and Information
- ONC Office of the National Coordinator for Health IT
- OSHEAN Ocean State Higher Education Economic Development Administration Network
- PAM Post-Award Monitoring
- PCC Public Computer Centers
- PPR Performance Progress Report
- Recovery Act American Recovery and Reinvestment Act of 2009
- RUS Rural Utilities Service
- SBA Sustainable Broadband Adoption
- SBDD State Broadband Data Development
- SDB Socially and Economically Disadvantaged Business
- SDN South Dakota Network
- SMS Short message service
- SOW Statement of work
- UAMS University of Arkansas Medical Services
- USDA U.S. Department of Agriculture
- ZBP ZIP Code Business Pattern
- ZCTA ZIP Code Tabulation Areas



Appendix E. Bibliography

"Commerce's NTIA Unveils National Broadband Map and New Broadband Adoption Survey Results." 2011. Last modified February 17, 2011. http://www.ntia.doc.gov/press/2011/NationalBroadbandMap_02172011.html.

"EPA: Value of American Life Drops to \$6.9 Million." 2008. Associated Press, July 11.

"Exploring the Benefits of Opening Accounts Online." 2006. America's Community Banker.

"Towards Access, Adoption, and Inclusion: A Call for Digital Equality and Broadband Opportunity." 2009. A statement from The National Black Caucus of State Legislators, November 28. http://www.thehispanicinstitute.net/files/u2/TAAI_MultiCaucus_Statement.pdf.

"South Korea Tops in Broadband Penetration Study." 2009. Last modified July 19, 2009. http://www.physorg.com/news164595432.html. Abdul Alkalimat, Doug Gills, and Kate Williams. 1995. Job-Tech: The Technological Revolution and its Impact on Society. Chicago: Twenty-First Century Books. http://www.cpsr.cs.uchicago.edu/jobtech/.

Abramowitz, Amy, and Michael Clements. "The Deployment and Adoption of Broadband Service: A Household Level Analysis. U.S. Government Accountability Office." http://web.si.umich.edu/tprc/papers/2006/518/TPRC2006.pdf.

Abu, M., B. Aris, and W.C. Tan. 2006. "GLOOTT Model: A Pedagogically-Enriched Design Framework of Learning Environment to Improve Higher Order Thinking Skills." AACE Journal 14, no. 2: 139-153. http://www.editlib.org/?fuseaction=Reader.ViewFullText&paper_id=6198

Ackerman, Michael J. 2010. "Telemedicine: Networks in the Service of Healthcare." Presentation, National Library of Medicine. Accessed January 27, 2012. http://courses.mbl.edu/mi/2010/presentations_spring/Telemed.ppt.

ACT. 2005. "Issues in College Readiness: Do Current State Standards and Assessments Reflect College Readiness?: A Case Study." http://www.act.org/research/policymakers/pdf/current_standards.pdf.

Adler, Richard. 2007. "Health Care Unplugged: The Evolving Role of Wireless Technology." Prepared for the California Health Care Foundation, November. http://www.chcf.org/~/media/Files/PDF/H/PDF%20HealthCareUnpluggedTheRoleOfWireless.pdf.

Alkalimat, Abdul, and Kate Williams. 2001. "Social Capital and Cyberpower in the African American Community: A Case Study of a Community Technology Center in the Dual City." In *Community Informatics: Shaping Computer Mediated Social Relations*, edited by Leigh Keeble and Brian Loader. London: Routledge. http://www.communitytechnology.org/cyberpower/cyberpower.pdf

Allen, I. Elaine, and Jeff Seaman. 2010. "Learning on Demand: Online Education in the United States, 2009." Babson Survey Research Group. http://sloanconsortium.org/publications/survey/pdf/learningondemand.pdf.

American Society for Training & Development. 2006. "Bridging the Skills Gap: How the Skills Shortage Threatens Growth and Competitiveness... and What to do About It. Alexandria, VA: American Society for Training Development." http://www.astd.org/NR/rdonlyres/FB4AF179-B0C4-4764-9271-17FAF86A8E23/0/BridgingtheSkillsGap.pdf.

America's Small Business Development Center Network (ASBDC). 2010. "About Us." Accessed November 15, 2010. http://www.asbdc-us.org/About_Us/aboutus.html.



- Anderson, Ben, and Yoel Raban. 2005. "The Social Impact of Broadband Household Internet Access." Chimera Working Paper 2005-2006, Colchester, University of Essex.
- Anderson Court Reporting. 2009. "National Broadband Plan Workshop Opportunities For Small And Disadvantaged Businesses." Remarks at FCC Opportunities for Small and Disadvantaged Businesses Workshop, August 18. http://www.broadband.gov/docs/ws_08_op_small_dis_biz.pdf.
- Anderson, Simon P., Andre de Palma, and Jacques-Francois Thisse. 1992. *Discrete Choice Theory of Product Differentiation*. Cambridge, Massachusetts: MIT Press.
- Annenberg Institute for School Reform. 2004. "Professional Learning Communities: Professional Development Strategies that Improve Instruction." Providence, RI: Annenberg Institute for School Reform. http://www.annenberginstitute.org/pdf/ProfLearning.pdf.
- Appropriate Framework for Broadband Access to the Internet over Wireline Facilities. 2005. FCC 05-150, 2005 WL 2347773, Federal Communications Commission, September 23.
- Arbore, Alessandro, and Andrea Ordanini. 2006. "Broadband Divide Among SMEs: The Role of Size, Location and Outsourcing Strategies." *International Small Business Journal*, 24 no. 1: 83-99.
- Aron, D. and D. Burnstein. 2003. "Broadband Adoption in the United States: An Empirical Analysis." Paper presented at the 31st Research Conference on Communication, Information and Internet Policy, Arlington, VA. http://intel.si.umich.edu/tprc/papers/2003/180/aronburnstein broadband adoption paper.pdf.
- Aron, Debra J., and Steven S. Wildman. 1996. "The Pricing of Customer Access in Telecommunications." Industrial and Corporate Change 5, no. 4: 1029-1047.
- Asia-Pacific Development Information Programme. 2007. "The Role of Governments in Promoting ICT Access and Use by SMEs: Considerations for Public Policy." APDIP e-Note 12, United Nations Development Programme. http://www.apdip.net/apdipenote/12.pdf/.
- Atkinson, Robert C, and Ivy E. Schultz. 2009. "Broadband in America: Where It Is and Where It Is Going." Preliminary Report for the Staff of the FCC's Omnibus Broadband Initiative, *Columbia Institute for Tele-Information*.
- Atkinson, Robert D. "Framing a National Broadband Policy." 2007. *The Journal of Communications Law* 16: 145-177. http://commlaw.cua.edu/res/docs/articles/v16/16-1/atkinson.pdf
- Atkinson, Robert D. 2010. "Network Policy and Economic Doctrines. The Information Technology and Innovation Foundation." Paper Presented at the 2010 Telecommunications Policy Research Conference, Arlington, Virginia, October. http://www.itif.org/files/2010-network-policy.pdf.
- Atkinson, Robert D., and Andrew S. McKay. 2007. "Digital Prosperity: Understanding the Economic Benefits of the Information Technology Revolution. *Information Technology and Innovation Foundation*, March 13. http://www.itif.org/files/digital_prosperity.pdf.
- Atkinson, Robert D., and Daniel Castro. 2008. "Digital Quality of Life: Understanding the Personal and Social Benefits of the Information Technology Revolution." *Information Technology and Information Foundation*. www.itif.org/index.php?id=179.
- Atkinson, Robert D., and Phil Weiser. 2009. "A Roundtable on the End of Scarcity, Open Architecture, and the Future of Broadband Competition Policy." *Information Technology and Innovation Foundation*, June. http://www.siliconflatirons.org/documents/publications/report/AtkinsonWeiserEndofScarcity.pdf.



- Atkinson, Robert D., and Richard Bennett. 2010. "ITIF Analysis of FCC Broadband Deployment Report." *Information Technology and Innovation Foundation*, July 21. http://www.itif.org/publications/itifanalysis-fcc-broadband-deployment-report.
- Atkinson, Robert D., and Richard Bennett. 2009. "A National Broadband Plan for Our Future: Reply Comments of Information Technology and Innovation Foundation." *Information Technology and Innovation Foundation*, July 21. http://www.itif.org/files/7.21.09Comments-final.pdf.
- Atkinson, Robert D., and Stephen J. Ezell. 2010. "The Good, the Bad, and the Ugly (and the Self-Destructive) of Innovation Policy: A Policymaker's Guide to Crafting Effective Innovation Policy." *Information Technology and Innovation Foundation*, October. http://www.itif.org/files/2010-good-badugly.pdf.
- Atkinson, Robert D. 2009. "Policies to Increase Broadband Adoption at Home." *Information Technology and Innovation Foundation*, November 5. http://www.itif.org/files/2009-demand-side-policies.pdf.
- Atkinson, Robert D. 2007. "The Case for a National Broadband Policy." *Information Technology and Innovation Foundation*, June. http://www.itif.org/files/CaseForNationalBroadbandPolicy.pdf.
- Atkinson, Robert D. 2010. "The Truth, and Nothing But the Truth about U.S. International Broadband Rankings." *Information Technology and Innovation Foundation*, May. http://www.itif.org/files/2010-BB-rankings.pdf.
- Atkinson, Robert D. 2008. "Time for a Post-Partisan Broadband Debate." *Information Technology and Innovation Foundation*, September 9. http://www.itif.org/files/Post-PartisanBroadbandDebate.pdf.
- Atkinson, Robert D. U.S. 2006. "Continues to Tread Water in Global Broadband Adoption." *Information Technology and Innovation Foundation*, April 12. http://www.itif.org/files/oecd-article1.pdf.
- Atkinson, Robert D. 2010. "Who's Who in Internet Politics: A Taxonomy of Information Technology Policy." *Information Technology and Innovation Foundation*, October. http://www.itif.org/files/2010-digital-politics.pdf.
- Atkinson, Robert D. 2009. "Wireless Taxation, Economic Growth and Economic Opportunity." *Information Technology and Innovation Foundation*, June 9.
- Atkinson, Robert D., Daniel Castro, and Stephen Ezell. 2009. "The Digital Road to Recovery: A Stimulus Plan to Create Jobs, Boost Productivity and Revitalize America." *Information Technology and Innovation Foundation*, January 7. http://www.itif.org/files/roadtorecovery.pdf.
- Atkinson, Robert D., Daniel Castro, Scott M. Andes, Stephen Ezell, Darrene Hackler, and Richard Bennett. "Innovation Policy on a Budget: Driving Innovation in a Time of Fiscal Constraint." *Information Technology and Innovation Foundation*, September 24. http://papers.ssrn.com/sol3/Delivery.cfm/SSRN_ID1722870_code666235.pdf?abstractid=1722870& mirid=1.
- Atkinson, Robert D., Daniel K. Correa, and Julie A. Hedlund. 2008. "Explaining International Broadband Leadership." *Information Technology and Innovation Foundation*, May. http://www.itif.org/files/ExplainingBBLeadership.pdf.
- Atkinson, Robert D., Stephen Ezell, Daniel Castro, and George Ou. 2009. "The Need for Speed: The Importance of Next-Generation Broadband Networks." *Information Technology and Innovation Foundation*, March 5. http://www.itif.org/files/2009-needforspeed.pdf.
- Atkinson, Robert D., Stephen Ezell, Scott M. Andes, Daniel Castro, and Richard Bennett. 2010. "The Internet Economy 25 Years After .com." *Information Technology and Innovation Foundation*. http://www.itif.org/files/2010-25-years.pdf.



Atkinson, Robert D. 2009. "The Role of Competition in a National Broadband Policy." *Journal on Telecommunications and High Technology Law*, 7. http://www.itif.org/files/JTHTL.pdf.

Atrostic, B. K., and Sang V. Nguyen. 2005. "It and Productivity in U.S. Manufacturing: Do Computer Networks Matter?" *Economic Inquiry*, 43 no. 3: 493-506. http://papers.ssrn.com/sol3/papers.cfm?abstract_id=906247.

Atrostic, B.K. 2007. "Measuring U.S. Innovative Activity." Working paper, U.S. Census Bureau, Center for Economic Studies. http://econpapers.repec.org/paper/cenwpaper/07-11.htm.

Atrostic, B.K. and Sang Nguyen. 2002. "Computer Networks and U.S. Manufacturing Plant Productivity: New Evidence from the CNUS Data." Working Papers from U.S. Census Bureau, Center for Economic Studies. http://www.ces.census.gov/index.php/ces/cespapers?down_key=101637.

Atrostic, B.K. and Sang Nguyen. 2005. "It and Productivity in U.S. Manufacturing: Do Computer Networks Matter?" *Economic Inquiry*, 43 no. 3: 493-506. http://papers.ssrn.com/sol3/papers.cfm?abstract_id=906247

Atrostic, B.K. and John Gates. 2001. "U.S. Productivity and Electronic Processes in Manufacturing." Working Papers from U.S. Census Bureau, Center for Economic Studies. http://www.ces.census.gov/index.php/ces/cespapers?down_key=101626.

Atrostic, B.K., John Gates and Ron Jarmin. 2000. "Measuring the Electronic Economy: Current Status and Next Steps." Working Papers, U.S. Census Bureau, Center for Economic Studies. http://www.census.gov/econ/estats/papers/3.pdf

Atrostic, B.K., John Gates, and Ron Jarmin. 2000. "Measuring the Electronic Economy: Current Status and Next Steps." Working Papers, U.S. Census Bureau, Center for Economic Studies. http://www.census.gov/econ/estats/papers/3.pdf.

Aufderheide, Patricia. 1999. Communications Policy and the Public Interest: The Telecommunications Act of 1996. New York: The Guilford Press, Inc..

Augereau, Angelique, and Shane Greenstein. 2001. "The Need for Speed in Emerging Communications Markets: Upgrades to Advanced Technology at Internet Service Providers." *International Journal of Industrial Organization* 19, no. 7: 1085-1102. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/The%20Need %20for%20Speed%20in%20Emerging%20Communications%20Markets%20-

%20Upgrades%20to%20Advanced%20Technology%20at%20Internet%20Service%20Providers.pdf.

Augereau, Angelique, Shane Greenstein, and Marc Rysman. 2008. "Coordination versus Differentiation in a Standards War: 56K modems." *Rand Journal of Economics* 37, no. 4: 887-909. http://www.wcas.northwestern.edu/csio/Papers/2004/CSIO-WP-0062.pdf

Badasyan, Narine, David Shideler, and Laura Taylor. 2007. "The Economic Impact of Broadband Deployment in Kentucky." *Regional Economic Development* 3, no. 2: 88-118.

Bangemann, Martin et al. 1994. "Europe and the Global Information Society: Bangemann Report Recommendations to the European Council." Bangemann Report Recommendations to the European Council, UMIC-Agency for the Knowledge Society. http://www.umic.pt/images/stories/publicacoes200801/raport_Bangemanna_1994.pdf.

Bargh, John A., and Katelyn Y. A. McKenn. 2004. "The Internet and Social Life." *Annual Review of Psychology* 55: 537-590.

Bartelsman, Eric, and Mark Doms. 2000. "Understanding Productivity: Lessons from Longitudinal Microdata." *Journal of Economic Literature* 38, no. 3: 569-594. http://papers.ssrn.com/sol3/papers.cfm?abstract_id=231129



- Bauer, Johannes M., Steven S. Wildman, and Bella Mody. 2006. "Making U.S. Telecommunications Policy: Who Participates and Who is Heard? The Roles of Research and Ideas." White paper, Department of Telecommunication, Information Studies, and Media Quello Center for Telecommunication Management and Law Michigan State University. http://mediaresearchhub.ssrc.org/making-u-s-telecommunications-policy-who-participates-and-whois-heard-the-roles-of-research-and-ideas/attachment.
- Bauer, Johannes M., Sungjoong Kim, and Steven S. Wildman. 2006. "Making U.S. Telecommunications Policy: Who Participates and Who is Heard? The Roles of Research and Ideas." White paper, Michigan State University.
- Bauer, Johannes, Sungjoong Kim, Bella Mody, and Steven Wildman. 2009. "The Role of Research in Communications Policy: Theory and Evidence." Paper presented at the annual meeting of the International Communication Association, Sheraton New York, New York City, NY, May 25. http://www.allacademic.com//meta/p_mla_apa_research_citation/0/1/4/9/6/pages14962/p14962-1.php.
- Beard, T. Randolph, George S. Ford, and Richard P. Saba. 2010. "Internet Use and Job Search." *Phoenix Center Policy Paper*, no. 39 (January). http://www.phoenixcenter.org/pcpp/PCPP39Final.pdf.
- Beard, T. Randolph, and George Ford. 2005. "Are Unbundled and Self-supplied Telecommunications Switching Substitutes? An Empirical Study." *International Journal of the Economics of Business* 12, no. 2. http://www.aestudies.com/library/ijeb05.pdf.
- Beard, T. Randolph, George Ford, and Thomas W. Koutsky. 2005. "Mandated Access and the Make-or-Buy Decision: The Case of Local Telecommunications Competition." *Quarterly Review of Economics and Finance* 45: 28-47.http://www.aestudies.com/library/qref05.pdf.
- Beard, T. Randolph, George Ford, Thomas M. Koutsky, Lawrence J. Spiwak. 2007. "Network Neutrality and Industry Structure." *Hastings Communications and Entertainment Law Journal* 29: 149-170. http://www.phoenix-center.org/CommEntNetworkNeutrality.pdf
- Beard, T. Randolph, George S. Ford., Lawrence J. Spiwak, and Michael Stern. 2009. "The Broadband Adoption Index: Improving Measurements and Comparisons of Broadband Deployment and Adoption." *Phoenix Center Policy Paper Series*, Phoenix Center for Advanced Legal & Economic Public Policy Studies. http://www.law.indiana.edu/fclj/pubs/v62/no2/9-SPIWAK_FINAL.pdf.
- Beard, T.R., R. Ekelund Jr., and George Ford. 2003. "Pursuing Competition in Local Telephony: The Law and Economics of Unbundling and Impairment." *Journal of Law, Technology and Policy.* http://www.aestudies.com/library/befimpair.pdf.
- Beaudry, Paul, Mark Doms, and Ethan Lewis. 2009. "Endogenous Skill Bias in Technology Adoption: City-Level Evidence from the IT Revolution." Working Paper Series 2006-24, Federal Reserve Bank of San Francisco. http://ideas.repec.org/p/nbr/nberwo/12521.html.
- Becta, Minister. 2008. "Taskforce on Home Access to Tech.". *Extending Opportunity* 4. http://partners.becta.org.uk/uploaddir/downloads/page_documents/partners/home_access_report.pdf.
- Beil, Richard, George Ford, and John Jackson. 2005. "On the Relationship between Telecommunications Investment and Economic Growth: Three Empirical Studies." International Economic Journal 19: 3-9. http://www.aestudies.com/library/granger.pdf.



- Beltran, Daniel, Kuntal Dias, and Robert Fairlie. 2010. "Are Computers Good for Children: The Effects of Home Computers on Educational Outcomes." *Economic Inquiry* 48: 771–792. http://people.ucsc.edu/~rfairlie/papers/published/econ%20inquiry%202010%20-%20home%20computers.pdf.
- Berry, Steven. 1994. "Estimating Discrete Choice Models of Product Differentiation." *Rand Journal of Economics* 25: 242-62.
- Bertrand, M., Duflo, E., and Mullainathan, S. 2004. "How Much Should We Trust Differences-in-Differences Estimates?" *The Quarterly Journal of Economics*, vol. 119, no. 1 (February): 249-275.
- Bing, Benny. Broadband Video Networking. Nowood, MA: ARTECH House, 2010.
- Bivens, Josh. 2003. "Updated Employment Multipliers for the U.S. Economy." Working Paper No. 268, Economic Policy Institute. www.epi.org/workingpapers/epi_wp_268.pdf.
- Booz Allen Hamilton, and Intel Corporation. 2005. "Enabling the Worldwide Information Society— Assessing the Impact of Government Assisted PC Purchase Programs." White paper, Government Assisted PC Programs. http://www.boozallen.com/media/file/150730.pdf.
- Borzekowski, Ron. 2002. "Measuring the Cost Impact of Hospital Information Systems: 1987-1994." Board of Governors of the Federal Reserve System. Article published September 10. http://www.federalreserve.gov/pubs/feds/2002/200242/200242pap.pdf
- Brandow Company, and the Economic Development Research Group. 2000. "Evaluation of the Appalachian Regional Commission's Infrastructure and Public Works Program Projects." http://www.arc.gov/assets/research_reports/EvaluationofARCsInfrastructureandPublicWorksProgram Projects3.pdf.
- Bresnahan, Tim, and Shane Greenstein. 1996. "Technical Progress and Co-Invention in Computing and in the Use of Computers." *Brookings Papers on Economics Activity: Microeconomics*: 1-78. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Technical%20 Progress%20and%20Co-Invention%20in%20Computing%20and%20in%20the%20Use%20of%20Computers.pdf
- Bresnahan, Tim, and Shane Greenstein. 1999. "Technological Competition and the Structure of the Computing Industry." *Journal of Industrial Economics* 47: 1-40. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Technological %20Competition%20and%20the%20Structure%20of%20the%20Computing%20Industry.pdf.
- Bresnahan, Tim, and Shane Greenstein. 1996. "The Competitive Crash in Large Scale Commercial Computing." In *The Mosaic of Economic Growth*, edited by Ralph Landau, Timothy Taylor, and Gavin Wright, Stanford: Stanford University Press: 357-397. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/chapters/The%20Com petitive%20Crash%20in%20Large-Scale%20Commercial%20Computing.pdf.
- Bresnahan, Tim, and Shane Greenstein. 2001. "The Economic Contribution of Information Technology: Towards Comparative and User Studies." *Journal of Evolutionary Economics* 11: 95-118. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/The%20Econo mic%20Contribution%20%20of%20Information%20Technology%20-%20Towards%20Comparative%20and%20User%20Studies.pdf.
- Bretschneider, Stuart, Jon P. Gant, and Lili Wang. 2005. "Evaluating Web-based E-Government Services with a Citizen-Centric Approach." Paper presented at Proceedings of the 38th Annual Hawaii International Conference on System Sciences (HCISS'05), Institute of Electrical and Electronics Engineers (IEEE). http://connections.ideals.illinois.edu/works/1545.



Broadband Stakeholders Group. 2004. "UK National Broadband Strategy." UK Department for Business Innovation and Skills. http://www.bis.gov.uk/files/file13444.pdf.

Brock, Gerald W. 1996. "Telecommunication Policy for the Information Age: From Monopoly to Competition." Information Economics and Policy 8, no. 1.

Brodey, Benjamin B., Keith H. Claypoole, Jeffrey Motto, Robert G. Arias, and Richard Goss. 2000. "Satisfaction of Forensic Psychiatry Patients with Remote Telepsychiatric Evaluation." *Psychiatric Services* 51, no. 10. http://ps.psychiatryonline.org/cgi/reprint/51/10/1305.pdf.

Broadband in Europe for All (BREAD). 2004. "Newtime FP6 Project." http://www.istbread.org/fp6_projects.asp?page=121.

Brown, Ken, and Shane Greenstein. 2000. "Identifying the Demand for Features: An application to Mainframe Computers." *Economics of Innovation and New Technology* 9: 353-383. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Identifying%20t he%20Demand%20for%20Features%20-%20An%20Application%20to%20Mainframe%20Computers.pdf.

Bruinsma, Frank, Sytze A. Rienstra, and Piet Rietveld. 1997. "Economic Impacts of the Construction of Transport Corridor: A Multi-level and Multi-approach Case Study for the Construction of A1 Highway in the Netherlands." *Regional Studies* 31: 391-402.

Brynjolfsson, Erik, Michael D. Smith, and Yu (Jeffrey) Hu. 2003. "Consumer Surplus in the Digital Economy: Estimating the Value of Increased Product Variety on Online Booksellers." *Management Science*, November. http://web.mit.edu/sloan-msa/Papers/2.4.pdf.

- Bureau of Labor Statistics. "Employment Projections 2008-18." United States Department of Labor. http://www.bls.gov/news.release/pdf/ecopro.pdf
- Bureau of Justice Statistics. *Crime & Justice Electronic Data*, 1977-2008. http://bjs.ojp.usdoj.gov/content/dtdata.cfm#index.

Bureau of Justice Statistics. *Law Enforcement Management and Administrative Statistics*: 1987, 1990, 1993, 1997, 1999, 2000, 2003, 2007. http://bjs.ojp.usdoj.gov/index.cfm?ty=dcdetail&iid=248#Documentation.

Burkart, Patrick, and Sharon Strover. 2003. "Local Services and Broadband Telecommunications Competition in Texas, Missouri and Kansas." Report from the Telecommunications and Information Policy Institute, University of Texas, May. http://www.utexas.edu/research/tipi/reports2/loc_services.pdf.

Burton, M.L. and Hicks, M.J. "The Residential and Commercial Benefits of Rural Broadband: Evidence from Central Appalachia." Center for Business and Economic Research, Marshall University, 2005. http://www.marshall.edu/cber/research/broadband/Final%20Rural%20Broadband%20July%202005.p df.

Burnett, WH, DH Mark, JE Midtling, and BB Zellner. 1995. "Primary Care Physicians in Underserved Areas: Family Physicians Dominate." *Western Journal of Medicine* 163:532–536.

Business-Higher Education Forum. 2006. "The American Competitiveness Initiative: Addressing the STEM Teacher Shortage and Improving Student Academic Readiness." *BHEF Issue Brief.* http://www.bhef.com/publications/documents/brief3_s06.pdf.

Cain Miller, Claire. 2010 "More Shoppers Go Online, Hungry for Deals." *New York Times,* February 15. http://bits.blogs.nytimes.com/2010/02/15/more-shoppers-go-online-hungry-for-deals/.



- Carbonara, Peter. 1997. "Here's a Business Plan to Fight Poverty." Last modified December 31. http://www.fastcompany.com/magazine/12/fightpoverty.html.
- Cardona, Melisande, Anton Schwarz, B. Burcin Yurtoglu, and Christine Zulehner. 2009. "Demand Estimation and Market Definition for Broadband Internet services." *Journal of Regulatory Economics* 35: 70-95.
- CareerOneStop. 2010. "America's Service Locator." Accessed November 15, 2010. http://www.servicelocator.org/
- Carnevale, Mary Lu. 2009. "White House Launches Recovery.gov to Track Stimulus Spending." *Wall Street Journal*, February 17. http://blogs.wsj.com/washwire/2009/02/17/white-house-launches-recoverygov-to-track-stimulus-spending/.
- Castro, Daniel. 2007. "Improving Health Care: Why a Dose of IT May Be Just What the Doctor Ordered." Information Technology and Information Foundation, October 25. www.itif.org/index.php?id=88.
- CBG Communications. 2008. "High-Speed Internet Deployment and Adoption Strategy Recommendations for the State of Washington." Prepared for Department of Information Service, December 1. http://www.broadband.wa.gov/sites/default/files/docs/HSISWG%20-%20Final%20Report%20-%201Dec08.pdf.
- Center for Medicare & Medicaid Services. 2008. "National Health Expenditure Projections 2008–2018." http://www.cms.hhs.gov/NationalHealthExpendData/downloads/proj2008.pdf.
- Chapman, Gary, Sharon Strover, and Jody Waters. 2003. "Beyond Community Networking and CTCs: Access, Development and Public Policy." Presented at the Telecommunications Policy and Research Conference, Washington, D.C., September. http://intel.si.umich.edu/tprc/papers/2003/230/TIF_evaluation.pdf.
- Chappell, Bill. 2010. "Google, FCC Look to Speed America's Lagging Broadband." *National Public Radio*, February 17.
 - $http://www.npr.org/blogs/alltechconsidered/2010/02/google_fcc_to_prod_americas_la.html.$
- Chaudhry, MD, Basit; Jerome Wang, MD; Shinyi Wu, PhD; Margaret Maglione, MPP; Walter Mojica, MD; Elizabeth Roth, MA; Sally C. Morton, PhD; and Paul G. Shekelle, MD, PhD. 2006. "Systemic Review: Impact of Health Information Technology on Quality, Efficiency, and Cost of Medical Care." *Annals of Internal Medicine* 144, no. 10.
- Chaudhuri, Anindya and Kenneth Flamm. 2005. "An Analysis of the Determinants of Broadband Access." Presented at the Telecommunications Policy Research Conference Washington, DC, September 24. http://web.si.umich.edu/tprc/papers/2005/485/Broadband.pdf.
- Childress, Ronald, Michael Cunningham, and Rebecca Goodwin. 2003. "The Changing Role of the Secondary Principal." *NASSP Bulletin* 87: 26-42.
- Chinn, Menzie, and Robert Fairlie. 2010. "ICT Use in the Developing World: An Analysis of Differences in Computer and Internet Penetration." *Review of International Economics* 18, no. 1: 153-167. http://people.ucsc.edu/~rfairlie/papers/published/rie%202010%20%20ict%20use%20in%20developin g%20countries.pdf.
- Chinn, Menzie, and Robert Fairlie. 2007. "The Determinants of the Global Digital Divide: A Cross-Country Analysis of Computer and Internet Penetration." *Oxford Economic Papers* 59: 16-44. http://people.ucsc.edu/~rfairlie/papers/published/oep%202007%20-%20global%20digital%20divide.pdf



- Choi, Soon-Yung, and Andrew B. Whinston. 2003. "The IT Revolution in the USA: The Current Situation and Problems." *The Internet Revolution: A Global Perspective*, edited by E. Giovannetti, M. Kagami, and M. Tsuji. Cambridge: University of Cambridge.
- Chongthammakun, R., and S. J. Jackson. "Extending Virtual Organizations in the Public Sector: Lessons from CSCW, STS, and Organization Science." Presented at the Hawaii International Conference on System Sciences, Los Alamitos, CA.
- Chow, Clifton, Jan Ellis, June Mark, and Bart Wise. 1997. "Impact of CTCNet Affiliates: Findings from a National Survey of Users." Report supported by National Science Foundation Grant No. ESI-9552567, Community Technology Centers' Network Research and Evaluation Team, Newton, MA.
- Cisco Systems. 2009. "Virtual Sales Expertise: How Cisco Supports Virtual Access to Technical Experts." http://www.cisco.com/web/about/ciscoitatwork/downloads/ciscoitatwork/pdf/Cisco_IT_Case_Study_S OAR_presentation.pdf.
- Clark, Drew. 2011. "Illinois and the National Broadband Map... Make That a Mashup!" Last modified February 2011.
- Cleary, Paul. 2009. "The Impact of Broadband on Education in the USA." In *Encyclopedia of Multimedia Technology and Networking Second Edition*, edited by Margherita Pagani. IGI Global.
- Coffman, Kerry G., and Andrew Odlyzko. 2002. "Growth of Internet." *Optic Fiber Telecommunications IVB*, edited by Ivan Kaminow and Tingye Li. San Diego: Academic Press.
- Collins, Hilton. 2009. "Nebraska Online Vehicle Tax Estimator Gives Citizens Tax and Fee Estimates." Government Technology, April 27. http://www.govtech.com/e-government/Nebraska-Online-Vehicle-Tax-Estimator-Gives.html.
- Colombo, Massimo G., and Luca Grilli. 2009. "The Diffusion of Broadband-Based Applications Among Italian Small and Medium Enterprises." *Contributions to Economics* 3: 174-185.
- Columbia Telecommunications Corporation. 2010. "The Impact of Broadband Speed and Price on Small Business." Report for SBA under contract number SBAHQ-09-C-0050. Report released November. http://archive.sba.gov/advo/research/rs373tot.pdf.
- Communications Workers of America. 2009. "A Report on Internet Speeds in All 50 States." http://cwafiles.org/speedmatters/state_reports_2009/CWA_Report_on_Internet_Speeds_2009.pdf?n ocdn=1.
- Community Service Society. 2009. "The Unheard Third 2009: Job Loss, Economic Insecurity, and a Decline in Job Quality." http://www.cssny.org/userimages/downloads/Unheard%20Third%202009%20Report%2010-7-09.pdf.
- Compaine, Benjamin M, ed. 2001. *The Digital Divide: Facing a Crisis or Creating a Myth?* Massachusetts: MIT Press.
- Connected Nation, Inc. 2008. "The Economic Impact of Stimulating Broadband Nationally." Last modified February 21, 2008. Washington, DC: Connection Nation Inc. http://connectednation.org/_documents/Connected_Nation_EIS_Study_Executive_Summary_02212 008.pdf.

Coomes, Paul A., and Thomas E. Lambert. 2001. "An Evaluation of the Effectiveness of Louisville's Enterprise Zone." *Economic Development Quarterly* 15: 168-80.



Cooper, Mark. 2004. "Expanding the Digital Divide & Falling Behind on Broadband: Why a Telecommunications Policy of Neglect is not Benign." Consumer's Union. http://www.consumerfed.org/elements/www.consumerfed.org/file/digitaldivide.pdf.

Corbishey, Jeffrey, Sherryl Hauser, and Catherine A. Little. 2009. "Constructing Complexity for Differentiated Learning." *Mathematics Teaching in the Middle School* 15, no. 1: 34-42. http://www.eric.ed.gov/ERICWebPortal/search/detailmini.jsp?_nfpb=true&_&ERICExtSearch_Search Value_0=EJ851904&ERICExtSearch_SearchType_0=no&accno=EJ851904.

Cornebise, Janet, June Mark, and Ellen Wahl. 1997. "Community Technology Centers: Impact on Individual Participants and Their Communities." Newton, MA: Education Development Center, Inc.

Correa, Daniel K. 2007. "Assessing Broadband in America: OECD and ITIF Broadband Rankings." Information Technology and Innovation Foundation. www.itif.org/files/BroadbandRankings.pdf.

Council of Economic Advisors. 2009. "Preparing the Workers of Today for the Jobs of Tomorrow." White House Report, Executive Office of the President. Last modified July. http://www.whitehouse.gov/assets/documents/Jobs_of_the_Future.pdf.

Council of Economic Advisors. 2009. "Preparing the Workers of Today for the Jobs of Tomorrow." Executive Office of the President. Last modified July 13, 2009. http://www.whitehouse.gov/assets/documents/Jobs_of_the_Future.pdf.

Council on Graduate Medical Education. 1998. *Tenth Report: Physician Distribution and Health Care Challenges in Rural and Inner-City Areas.* Washington, DC: Government Printing Office.

Crandall, Robert, and Kenneth Flamm. 1989. Changing the Rules: Technological Change, International Competition, and Regulation in Communications. Brookings Institution Press.

Crandall, Robert W. 2003. "Broadband Communications". Preliminary draft, The Brookings Institution, May. http://www.brookings.edu/views/papers/crandall/200305.pdf.

Crandall, Robert W., and Charles L. Jackson. 2003. "The \$500 Billion Opportunity: The Potential Economic Benefit of Widespread Diffusion of Broadband Internet Access." In *Down to the Wire: Studies in the Diffusion and Regulation of Telecommunications Technologies*, edited by Allan L. Shampine. Haupaugge, NY: Nova Science Press.

Crandall, Robert, William Lehr and Robert Litan. 2007. "The Effects of Broadband Deployment on Output and Employment: A Cross-sectional Analysis of U.S. Data." *The Brookings Institution: Issues in Economic Policy*, no. 6 (July). http://www.brookings.edu/views/papers/crandall/200706litan.pdf.

Crandall, Robert, Charles L. Jackson, and Hal J. Singer. 2003. "The Effect of Ubiquitous Broadband Adoption on Investment, Jobs, and the U.S. Economy." Report for the New Millennium Research Council, Criterion Economics, LLC. http://newmillenniumresearch.org//archive/bbstudyreport_091703.pdf.

Crawford, Susan P. 2007. "Internet Think." *Journal on Telecommunications and High Technology Law* 5, no. 2. http://www.jthtl.org/content/articles/V5I2/JTHTLv5i2_Crawford.PDF.

Crawford, Susan P. 2006. "Network Rules." Working paper, Cardozo Legal Studies Research Paper No. 159. http://ssrn.com/abstract=885583.

Crawford, Susan P. 2005. "Shortness of Vision: Regulatory Ambition in the Digital Age." Cardozo Legal Studies Research Paper, No. 102. *Fordham Law Review*. http://scrawford.net/display/Shortness_of_Vision.pdf.

Crawford, Susan P. 2006. "The Ambulance, the Squad Car, and the Internet." *Berkeley Technology Law Journal* 21, no. 2: 874-944. http://www.btlj.org/data/articles/21_02_04.pdf.



- Crawford, Susan P. 2008. "The Radio and the Internet." Cardozo Legal Studies Research Paper, No. 197, *Berkeley Technology Law Journal*.
- Crawford, Susan P. 2009. "Transporting Communications." *Boston University Law Review* 89: 871. http://ssrn.com/abstract=1254983.
- Crawford, Susan P. 2007. "The Internet and the Project of Communications Law." Working paper, Cardozo Law School. http://www.scrawford.net/courses/Crawford,%20The%20Internet%20and%20the%20Project%20of% 20Communications%20Law.pdf.
- Cregan, Patrick, J Li, LS Wilson, RY Qiao, T Percival, A Krumm-Heller, and S. Stapleton. 2006. "Development of a Broadband TeleHealth System for Critical Care: Process and Lessons Learned." *Telemedicine and e-Health* 12, no. 5: 552-560.
- Dalbey, Dennis, and Paul Rappoport. 2010. "Broadband in Pennsylvania: An Essential Service?" *Speakers Journal* 9, no. 9. http://www.house.state.pa.us/SpkrJournal/documents/9/v9_a9.pdf.
- Daly, Mary, Mark Doms, and Jeremy Gerst. 2009. "Regional Growth and Resilience: Evidence from Urban Growth Centers." *Federal Reserve Bank of San Francisco Economic Review*. http://www.frbsf.org/publications/economics/review/2009/er1-12.pdf.
- Darby, Larry F. 2008. "Broadband Tax Relief Will Contribute Substantially to Efforts to Create Jobs." *American Consumer Institute Center for Citizen Research.* www.theamericanconsumer.org/2008/11/25/broadband-tax-relief-will-contribute-substantially-toefforts-to-create-jobs/.
- Darby, Larry F., Jospeh P. Fuhr Jr., and Stephen B. Pociask. 2010. "The Internet Ecosystem: Employment Impacts of National Broadband Policy". *The American Consumer Institute*. http://www.theamericanconsumer.org/wp-content/uploads/2010/01/aci-jobs-study-final1.pdf.
- Darkins, M.D., Patricia Ryan, R.N., M.S., Rita Kobb, M.N., A.P.R.N., Linda Foster, M.S.N., R.N., Ellen Edmonson, R.N., M.P.H., Bonnie Wakefield, Ph.D., R.N., and Anne E. Lancaster, B.Sc. 2008. "Care Coordination/Home Telehealth: The Systematic Implementation of Health Informatics, Home Telehealth, and Disease Management to Support the Care of Veteran Patients with Chronic Conditions." *Telemedicine and e-Health,* December.
- Card, David, and Alan B. Krueger. 1994. "Minimum Wages and Employment: A Case Study of the Fast-Food Industry in New Jersey and Pennsylvania." *American Economic Review*, vol. 84, no. 4: 774– 775.
- David, Paul, and Shane Greenstein. 1990. "The Economics of Compatibility of Standards: A Survey." *Economics of Innovation and New Technology* 1: 3-41. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/The%20Econo mics%20of%20Compatibility%20Standards%20-%20A%20Survey.pdf.
- Davidson, Charles M., and Michael J Santorelli. 2008. "The Impact of Broadband on Senior Citizens: A study commissioned by the U.S. Chamber of Commerce." The Advanced Communications Law & Policy Institute at New York Law School, December. http://www.nyls.edu/user_files/1/3/4/30/83/BroadbandandSeniors.pdf.

Davidson, Charles M., and Michael J. Santorelli. 2010. "The Impact of Broadband on Education: A study commissioned by the U.S. Chamber of Commerce." Working paper, The Advanced Communications Law & Policy Institute at New York Law School, December. http://www.uschamber.com/sites/default/files/about/US_Chamber_Paper_on_Broadband_and_Educ ation.pdf.



- Davidson, Charles M., and Michael J. Santorelli. 2009. "The Impact of Broadband on Telemedicine: A study commissioned by the U.S. Chamber of Commerce." Working paper, The Advanced Communications Law & Policy Institute at New York Law School, April. http://www.uschamber.com/sites/default/files/about/0904Broadband_and_Telemedicine.pdf.
- Davis, Denise M. 2009. "The Condition of Libraries: Trends 1999-2009." Report prepared for ALA Senior Management and Unit Managers. Report prepared for ALA Senior Management and Unit Manager, American Library Association, Office for Research & Statistics, December. http://www.ala.org/ala/research/initiatives/Condition_of_Libraries_1999.20.pdf.
- Davis, Denise M., John Carlo Bertot, and Charles R. McClure. 2008. "Findings from the State Library Agency Chief Officers' Qualitative Questionnaire." Chicago, IL: American Library Association. http://www.ala.org/ala/research/initiatives/plftas/previousstudies/0708/contentsacknowledgements.pdf
- DeBakey, Michael E. 2006. "Hurricanes Katrina and Rita: Role of Individuals and Collaborative Networks in Monitoring/Coordinating Societal and Professional Resources for Major Disasters." *Critical Care* 10, no. 1: 205.
- Del Aguila-Obra, Ana R., and Antonio Padilla-Melendez. 2006. "Organizational Factors Affecting Internet Technology Adoption." *Internet Research* 16, no. 1: 94-110.
- Department of Culture, Media, and Sport and Department of Business Innovation & Skills. 2009. "Digital Britain: Final Report." Report presented to Parliament, Department of Culture, Media, and Sport. http://www.culture.gov.uk/images/publications/digitalbritain-finalreport-jun09.pdf.
- Department Of Housing And Urban Development. 2005. "Empowerment Zones: Performance Standards for Utilization of Grant Funds." Proposed rule, *Federal Register* 70, no. 109 (June).
- Department of Veteran Affairs. 2009. "VA Data Show Home Health Technology Improves Access to Care." Last modified January 5, 2009. http://www.va.gov/opa/pressrel/pressrelease.cfm?id=1637.
- Deubel, Patricia. 2008. "K-12 Online Teaching Endorsements: Are They Needed?" *T.H.E. Journal*, January 10. http://thejournal.com/Articles/2008/01/10/K12-Online-Teaching-Endorsements-Are-They-Needed.aspx.
- Digital Impact Group and Econsult Corporation. 2010. "The Economic Impact of Digital Exclusion." Article published March 5. http://www.econsult.com/articles/030810_costofexclusion.pdf
- DiMaggio, Paul, Eszter Hargittai, Coral Celeste and Steven Shafer. 2004. "Digital Inequality: From Unequal Access to Differentiated Use." *Social Inequality*, edited by Kathryn Neckerman. New York: Russell Sage Foundation: 355-400. http://www.webuse.org/pdf/DiMaggioEtAl-DigitalInequality2004.pdf.
- DiMaggio, Paul, Eszter Hargittai, Coral Celeste, and Steven Shafer. 2001. "From the 'Digital Divide' to `Digital Inequality': Studying Internet Use As Penetration Increases." Working paper, Princeton University.
- DiMaggio, Paul, Eszter Hargittai, Coral Celeste, and Steven Shafer. 2001. "From Unequal Access to Differentiated Use: A Literature Review and Agenda for Research on Digital Inequality." Report prepared for the Russell Sage Foundation, Sociology Department of Princeton University. http://www.eszter.com/research/pubs/dimaggio-etal-digitalinequality.pdf.
- DiMaggio, Paul, Eszter Hargittai, W.Russell Neuman, and John P. Robinson. 2001. "The Social Implications of the Internet." *Annual Review of Sociology* 27: 307-336. http://www.webuse.org/pdf/DiMaggioEtAl-SocialImplications2001.pdf.



- Dinc, Mustafa, and Serdar Yilmaz. 2002. "Telecommunications and Regional Development: Evidence From the U.S. States." *Economic Development Quarterly* 16: 211-228.
- Dobransky, Kate, and Eszter Hargittai. 2006. "The Disability Divide in Internet Access and Use." *Information, Communication and Society* 9, no. 2: 313-334. http://www.webuse.org/pdf/DobranskyHargittai-DisabilityDivide2006.pdf.
- Doms, Mark, and Chris Forman. 2005. "Prices for Local Area Network Equipment." FRBSF Working Paper 2003-13, Information Economics and Policy 17, no. 3: 365-388. http://www.frbsf.org/publications/economics/papers/2003/wp03-13bk.pdf.
- Doms, Mark, and Ethan Lewis. 1998. "Labor Supply and Personal Computer Adoption." Working Papers 06-10, Federal Reserve Bank of Philadelphia. http://ideas.repec.org/p/fip/fedfwp/2006-18.html.
- Doms, Mark, and Timothy Dunne. 1998. "Capital Adjustment Patterns in Manufacturing Plants." *Review of Economic Dynamics* 1, no. 2: 409-429.
- Doms, Mark, and Timothy Dunne. 1995. "Energy Intensity, Electricity Consumption, and Advanced Manufacturing Technology Usage." *Technological Forecasting and Social Change* 49: 297-310. http://www.ces.census.gov/index.php/ces/cespapers?down_key=100218.
- Doms, Mark, Robert McGuckin, and Mary Streitwieser. 1998. "The Effect of Technology Use on Productivity Growth." *Economics of Innovation and New Technology* 7: 1-26. http://www.ces.census.gov/index.php/ces/cespapers?down_key=100250.
- Doms, Mark, Timothy Dunne, and Kenneth Troske. 1997. "Workers, Wages, and Technology." *Quarterly Journal of Economics* 112, no. 1: 253-290.
- Doms, Mark, Timothy Dunne, and Mark J. Roberts. 1995. "The Role of Technology Use in the Survival and Growth of Manufacturing Plants in Innovation, Evolution of Industry and Economic Growth: The International Library of Critical Writings in Economics." *International Journal of Industrial Organization* 13, no. 4: 523-542.
- Doms, Mark, Wendy Dunn, Stephen Oliner, and Daniel Sichel. 2004. "How Fast Do Personal Computers Depreciate?" *Concepts and New Estimates In NBER: Tax Policy and the Economy* 18: 37-79. http://www.nber.org/chapters/c10869.pdf.
- Doms, Mark, Ron Jarmin, and Shawn Klimek. 2004. "Information Technology Investment and Firm Performance in U.S. Retail Trade." *Economics of Innovation and New Technology* 13: 595-614.
- Doms, Mark. 2003. "Communications Equipment: What Has Happened to Prices?" FRBSF Working Paper 2003-15, Federal Reserve Bank of San Francisco. http://www.frbsf.org/publications/economics/papers/2003/wp03-15bk.pdf.
- Doms, Mark. 1996. "Estimating Capital Efficiency Schedules within Production Functions." *Economic Inquiry* 34: 78-92. http://www.ces.census.gov/index.php/ces/cespapers?down_key=100181
- Doms, Mark. 2005. "IT Investment: Will the Glory Days Ever Return?" *The Federal Reserve Bank of San Francisco Economic Letter*. http://www.frbsf.org/publications/economics/letter/2005/el2005-13.pdf.
- Doms, Mark. 2004. "The Boom and the Bust in Information Technology Investment." *The Federal Reserve Bank of San Francisco Economic Review*. http://www.frbsf.org/publications/economics/review/2004/er19-34bk.pdf.
- Doms, Mark. 2005. "The Diffusion of Personal Computers across the U.S." *The Federal Reserve Bank of San Francisco Economic Letter*. http://www.frbsf.org/publications/economics/letter/2005/el2005-37.pdf



Doug McKessock, Clare Chua, and Mary Foster. 2005. "Dynamic Online Homework System: An Enabler of Learning." Paper presented at the Ascilite Conference, December. http://www.ascilite.org.au/conferences/brisbane05/blogs/proceedings/47_McKessock.pdf;

Downes, Tom, and Shane Greenstein. 2002. "Universal Access and Local Commercial Internet Markets." *Research Policy 31*: 1035-1052. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Universal%20A ccess%20and%20Local%20Commercial%20Internet%20Markets.pdf.

Downes, Tom, and Shane Greenstein. 2007. "Understanding Why Universal Service Obligations May Be Unnecessary: The Private Development of Local Internet Access Markets." *Journal of Urban Economics* 62: 2-26.

http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Accessentry.pd f

- Downes, Tom, and Shane Greenstein. 1999. "Do Commercial ISPs Provide Universal Access?" *Competition, Regulation and Convergence: Current Trends in Telecommunications Policy Research,* edited by Sharon Gillett and Ingo Vogelsang. New Jersey: Lawrence Erlbaum Associates, 195-212. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/chapters/Do%20Com mercial%20ISPs%20Provide%20Universal%20Access.pdf
- Duffy-Deno, K.T. 2003. "Business Demand For Broadband Access Capacity." *Journal of Regulatory Economics* 24, no. 3: 359-372.

Dutz, Mark, Jonathan Orszag, and Robert Willig. 2009. "The Substantial Consumer Benefits Of Broadband Connectivity For U.S. Households." Report commissioned by the Internet Innovation Alliance, July.

- Duvall, Jerry, and George Ford. 2000. "Changing Industry Structure: The Economics of Entry and Price Competition." *Telecommunications and Space Journal* 7. http://www.phoenix-center.org/pcpp/PCPP10Final.pdf.
- Eastin, Matthew S., and Robert LaRose. 2000. "Internet Self-Efficacy and the Psychology of the Digital Divide." *Journal of Computer-Mediated Communication* 6, no. 1. http://jcmc.indiana.edu/vol6/issue1/eastin.html#abstract.
- Economics & Statistics Administration and the National Telecommunications and Information Administration. "Exploring the Digital Nation: Home Broadband Internet Adoption in the United States." U.S. Department of Commerce. http://www.ntia.doc.gov/report/2010/exploring-digital-nationhome-broadband-internet-adoption-united-states.
- Economic Development Administration. "Comprehensive Economic Development Strategies: Summary of Requirements." Economic Development Administration, U.S. Department of Commerce. http://www.eda.gov/PDF/CEDSFlyer081706.pdf.
- Edwards, Paul N., Steven J. Jackson, Geoffrey C. Bowker, and Cory P. Knobel. 2006. "Understanding Infrastructure: Dynamics, Tensions, and Design." Report of the workshop, History and Theory of Infrastructure: Lessons for New Scientific Cyberinfrastructures, Ann Arbor, MI, September 28 -October 1.

http://deepblue.lib.umich.edu/bitstream/2027.42/49353/3/UnderstandingInfrastructure2007.pdf.

Eisner Gillett, Sharon, and William Lehr. 1999. "Availability of Broadband Internet Access: Empirical Evidence." Paper prepared for the Twenty-Seventh Annual Telecommunications Policy Research Conference, Alexandria, Virginia, September 25-27. http://dspace.mit.edu/bitstream/handle/1721.1/1480/LehrGillettTPRC99_0523.pdf?sequence=1



Ekelund, Robert Jr., and George Frank. 2003. "Innovation, Investment, and Unbundling: An Empirical Update." Yale Journal on Regulation 20. http://www.aestudies.com/library/efupdate.pdf.

- Ellin, Abby. 2009. "Entrepreneurs Leverage New Orleans's Charm to Lure Small Businesses." *New York Times*, July 29,.
 - http://www.nytimes.com/2009/07/30/business/smallbusiness/30sbiz.html?pagewanted=all.
- E. J. Ourso College of Business, Louisiana State University. "Louisiana Business and Technology Center Mobile Classroom: About Us." Accessed November 15, 2010. http://www.bus.lsu.edu/centers/lbtc/mobileclassroom.asp
- Fairlie, Robert, and Rebecca London. 2010. "The Effects of Home Computers on Educational Outcomes: Evidence from a Field Experiment with Community College Students." Working Paper, University of California, Santa Cruz.
- Fairlie, Robert W., Rebecca A. London, Rachel Rosner, and Manuel Pasto. 2006. "Crossing the Divide: Immigrant Youth and Digital Disparity in California." Santa Cruz, CA: University of California at Santa Cruz & Center for Justice, Tolerance, and Community. http://www2.ucsc.edu/cjtc/docs/digital.pdf.
- Fairlie, Robert. 2007. "Explaining Differences in Access to Home Computers and the Internet: A Comparison of Latino Groups to Other Ethnic and Racial Groups." *Electronic Commerce Research* 7: 265-291. http://people.ucsc.edu/~rfairlie/papers/published/ecr%202007%20-%20latino%20technology.pdf
- Fairlie, Robert. 2004. "Race and the Digital Divide." *Contributions to Economic Analysis and Policy* 3, no. 1. http://people.ucsc.edu/~rfairlie/papers/published/bejournals%202004%20-%20digitaldividefinal.pdf.
- Fairlie, Robert. 2005. "The Effects of Home Computers on School Enrollment." *Economics of Education Review* 24, no. 5: 533–547. http://www2.ucsc.edu/cjtc/docs/r_schoolcomp6.pdf.
- Fairlie, Robert. 2006. "The Personal Computer and Entrepreneurship." *Management Science* 52, no. 2: 187-203.
- Federal Emergency Management Agency. 2007. "Technical Report #166: I-35 Bridge Collapse and Responses." *Technical Report Series*, U.S. Fire Administration. http://www.usfa.dhs.gov/downloads/pdf/publications/tr_166.pdf.
- Fellowes, Matt. 2006. "The High Cost of Being Poor: Reducing the Costs of Living for Working Families." Brookings Institution.
- Ferguson Ph.D., A. Stewart, John Kokesh M.D., Chris Patricoski M.D., Phil Hofstetter, and Nathan Hogge. 2008-2009. "Impact of Store-And-Forward Telehealth in Alaska: A Seven Year Retrospective." Anchorage, AK: Alaska Native Tribal Health Consortium. http://www.hss.state.ak.us/healthcommission/meetings/200908/pdf/AFHCAN_impact.pdf.
- Ferguson, Charles H. 2004. "The Broadband Problem: Anatomy of a Market Failure and a Policy Dilemma." *Brookings Institute Press*.

http://www.floridataxwatch.org/resources/pdf/110507FinalReportFLVS.pdf.

Firth, Lucy and David Mellor. "Broadband: Benefits and Problems." Telecommunications Policy 29, no. 2-3 (March-April 2005): 223-236.

Flamm, Kenneth and Anindya Chaudhuri. "The Determinants of Broadband Access." Paper presented at the Telecommunications Policy Research Conference, Washington, DC, September 24, 2005. http://web.si.umich.edu/tprc/papers/2005/485/Broadband.pdf.



- Flamm, Kenneth. Targeting the Computer: Government Support and International Competition. Washington, DC: Brookings Institution Press, 1987.
- Flamm, Kenneth. Creating the Computer: Government, Industry, and High Technology. Washington, DC: Brookings Institution Press, 1988.
- Florida TaxWatch Center for Educational Performance and Accountability. 2007. "Final Report: A Comprehensive Assessment of Florida Virtual School."
- Ford, George S. 2007. "Does Municipal Supply of Communications Crowd-Out Private Communications Investment? An Empirical Study." *Energy Economics* 29, no. 3 (May): 467-478.
- Ford, George S. 2009. "Normalizing Broadband Connections." Phoenix Center Perspectives, May 12.
- Ford, George S., and John D. Jackson. 2004. "Competition and Investment in Telecommunications." *Atlantic Economic Journal* 32, no. 1: 71.
- Ford, George S., and John D. Jackson. 2004. "Demand Elasticities for International Message Telephone Service." *Applied Economics* 36, no. 14: 1523-1527.
- Ford, George S., and John D. Jackson. 1998. "On the Interpretation of Policy Effects from the Estimates of Simultaneous Systems of Equations." *Applied Economics* 30: 995.
- Ford, G.S. and T.M. Koutsky. 2005. "Broadband and Economic Development: a Municipal Case Study from Florida." *Applied Economic Studies* 1. http://www.designnine.com/library/docs/other_papers/BroadbandFactFiction.pdf.
- Ford, George, Thomas Koutsky, and Lawrence J. Spiwak. 2007. "The Broadband Performance Index: A Policy-Relevant Method of Assessing Broadband Adoption Among Countries." *Phoenix Center Policy Paper* No. 29 (June).
- Ford, George, Thomas Koutsky, and Lawrence Spiwak. 2006. "The Economics of Build-out Rules in Cable Television." *Hastings Communications and Entertainment Law Journal* 28: 207.
- Ford, George, Thomas Koutsky, and Lawrence Spiwak. 2005. "The Impact of Video Service Regulation on the Construction of Broadband Networks to Low-Income Households." *Phoenix Center Policy Paper*, September 23.
- Forgrave, Reid. 2008. "Living on the Edge: Disabled Become Able to Work." *DesMoines Register*, March 20. http://www.desmoinesregister.com/apps/pbcs.dll/article?AID=/20080320/NEWS/803200376/-1/SPORTS09.
- Forman, Chris, Avi Goldfarb, and Shane Greenstein. 2003. "Which Industries use the Internet?" *Organizing the New Industrial Economy*, edited by Michael Baye. New York: Elsevier: 47-72. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/chapters/Which%20Ind ustries%20use%20the%20Internet%20-%20Baye%20Volume.pdf.
- Forman, Chris, Avi Goldfarb, and Shane Greenstein. 2005. "Geographic Location and the Diffusion of Internet Technology." *Electronic Commerce Research and Applications* 4, no. 1: 1-13.
- Forman, Chris, Avi Goldfarb, and Shane Greenstein. 2005. "How did Location Affect Adoption of the Commercial Internet? Global Village vs. Urban Leadership." *Journal of Urban Economics* 58, no. 3: 389-420.
- Forman, Chris, Avi Goldfarb, and Shane Greenstein. 2005. "How Do Industry Features Influence the Role of Location on Internet Adoption?" *Journal of the Association for Information Systems* 6, no. 12: 383-408



Forman, Chris, Avi Goldfarb, and Shane Greenstein. 2008. "Understanding the Inputs into Innovation: Do Cities Substitute for Internal Firm Resources?" *Journal of Economics & Management Strategy* 17, no. 2: 295-316.

Forman, Chris, Avi Goldfarb, and Shane Greenstein. 2005. "Geographic Location and the Diffusion of Internet Technology." *Electronic Commerce and Research Applications* 4: 1-13. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/ecra.pdf.

Forman, Chris, Avi Goldfarb, and Shane Greenstein. 2005. "Industry Features Influence the Role of Location on Internet Adoption?" *Journal of the Association for Information Systems* 6: 383-408. http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.84.3727&rep=rep1&type=pdf.

Forman, Chris, Avi Goldfarb, and Shane Greenstein. 2003. "The Geographic Dispersion of Commercial Internet Use." *Rethinking Rights and Regulations: Institutional Responses to New Communications Technologies*, edited by Lorrie Cranor and Steven Wildman. Boston: MIT Press: 113-146. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/chapters/The%20Geog raphic%20Dispersion%20of%20Commercial%20Internet%20Use.pdf.

Forman, Chris, Avi Goldfarb, and Shane M. Greenstein. 2009. "The Internet and Local Wages: Convergence or Divergence?" NBER Working Paper No. w14750, NBER Working Paper Series.

Fornefeld, Martin, Gilles Delaunay, and Dieter Elixmann. 2008. "The Impact of Broadband on Growth and Productivity." A study on behalf of the European Commission, MICUS Management Consulting GmbH, Düsseldorf, Germany.

Fornefeld, Martin, Gilles Delaunay, and Dieter Elixmann. 2008. "The Impact of Broadband on Growth and Productivity. A Study on Behalf of the European Commission." DG Information Society and Media. MICUS.

Forrester Research, Inc. 2003. "The Wide Range of Abilities and Its Impact on Computer Technology." Research study commissioned by Microsoft Corporation.

Fox, Susannah. 2010. "FCC: Broadband Adoption and Use in America." *Pew Research Center*. http://www.pewinternet.org/Commentary/2010/February/FCC-Broadband-Adoption-and-Use-in-America.aspx.

Fox, Susannah. 2009. "The Social Life of Health Information: Americans' Pursuit of Health Takes Place within a Widening Network of both Online and Offline Sources." *Pew Research Center: Pew Internet* & American Life Project. http://www.pewinternet.org/Reports/2009/8-The-Social-Life-of-Health-Information.aspx.

Federal Bureau of Investigation. "National Incident-Based Reporting System, 1991 – 2008." Uniform Crime Reporting Program.

http://www.icpsr.umich.edu/icpsrweb/ICPSR/series/00128/studies?archive=ICPSR&sortBy=7.

Federal Communications Commission. 2010. "Broadband Plan's Working Recommendations for Key National Priorities Unveiled."

Federal Communications Commission. 2010. "National Broadband Plan: Connecting America." Accessed January 27, 2012. http://www.broadband.gov/plan/10-healthcare/.

Freedman, Gordon. 2009. "From an Education Pipeline to Cycles of Learning: Is the Tipping Point for Education in Sight?" Summary from the Pipeline Matters Council: Improving K20 Student Progression, Blackboard Institute, July 13. http://www.inacol.org/research/docs/Blackboard%20K20%20CouncilSummaryReport.pdf



Freese, J., S. Rivas, and Eszter Hargittai. 2006. "Cognitive Ability and Internet Use Among Older Adults." *Poetics* 34, no. 4 (August-October): 236-249. http://www.webuse.org/pdf/FreeseRivasHargittai-CognitiveAbility2006.pdf.

Frieden, Rob. 2005. "Lessons from Broadband Development in Canada, Japan, Korea and the United States." *Telecommunications Policy* 29: 595–613. http://front.sjtu.edu.cn/datacomm/reader/Lessons%20from%20broadband%20development.pdf.

Fryer GE, LA Green, SM Dovey, and RI Phillips Jr. 2001. "The United States Relies on Family Physicians Unlike any other Specialty." *American Family Physician* 63:1669–1703.

Fuhr, Joseph J., and Stephen B. Pociask. 2007. "Broadband Services: Economic and Environmental Benefits." *American Consumer Institute*. http://www.nederlandbreedbandland.nl/uploaded/FILES/Final_Green_Benefits.pdf.

- Future Foundation. 2009. "Broadband has Positive Impact on Education Says Research." Last modified September 14, 2009. http://www.publictechnology.net/content/1743.
- Gajda, Rebbeca, and Matthew Militello. 2008. "Retaining School Principals: What We Can Learn From Practicing Administrators." AASA Journal of Scholarship and Practice 5: 14-20.
- Galster, George, Chris Hayes, and Jennifer Johnson. 2005. "Identifying Robust, Parsimonious Neighborhood Indicators." *Journal of Planning Education and Research* 24: 265-280.
- Gant, Diana B., and John P. Gant. 2003. "Enhancing E-service Delivery in State Government." *E-Government*, edited by Mark Abramson and Therese Morin. Maryland: Rowman and Littlefield Publishers, 53-80. http://www.alibris.com/search/books/isbn/9780742527973.
- Gant, Jon P. and Donald S. Ijams. 2004. "Digital Government and Geographic Information Systems." *Digital Government: Principles and Best Practices*, edited by Alexei Pavlichev and David Garson. Hershey: Idea Group Publishing, 238-262.
- Gillett, Sharon Eisner, and Marvin A. Sirbu. 2006. "Measuring Broadband's Economic Impact." Prepared for the U.S. Department of Commerce, Economic Development Administration National Technical Assistance, Training, Research, and Evaluation Project #99-07-13829, Washington, DC, February.
- Gillett, Sharon Eisner, and William Lehr. 1999. "Availability of Broadband Internet Access: Empirical Evidence." Twenty-Seventh Annual Telecommunications Policy Research Conference, Alexandria, Virginia, September 25-27. http://mit.dspace.org/bitstream/handle/1721.1/1480/LehrGillettTPRC99_0523.pdf?sequence=1.
- Gillett, Sharron, William Lehr, Carlos Osorio, and Marvin Sirbu. 2006. "Measuring Broadband's Economic Impact." U.S. Department of Commerce, Economic Development Administration. http://web.si.umich.edu/tprc/papers/2005/475/TPRC2005_Gillett%20Lehr%20Sirbu%20Osorio%20su bmitted.pdf.
- Gilroy, Angele A., and Lennard G. Kruger. 2007. "Broadband Internet Access and the Digital Divide: Federal Assistance Programs" *Broadband Internet: Access, Regulation and Policy*, edited by Ellen Cohen. New York: Nova Science Publishers, Inc.
- Glass, Victor, and Stela K. Stefanova. 2010. "An Empirical Study of Broadband Diffusion in Rural America." *Journal of Regulatory Economics* 38: 70-85.
- Global Initiative for Inclusive Information and Communication Technologies. 2007. "The Accessibility Imperative."



Global Sustainability Initiative and Boston Consulting Group. 2008. "SMART 2020: Enabling the Low Carbon Economy in the Information Age." *United States Report Addendum* 49. http://www.gesi.org/LinkClick.aspx?fileticket=cOArprYnXWY%3D&tabid=60.

Goetz, Stephan J. 2008. "Self-Employment in Rural America: The New Economic Reality." *Rural Realities*. Columbia, MO: The Rural Sociological Society. http://ruralsociology.org/StaticContent/Publications/Ruralrealities/pubs/RuralRealities2-3.pdf.

Goolsbee, Austan, and Peter J. Klenow. 1999. "Evidence on Learning and Network Externalities in the Diffusion of Home Computers." NBER Working Paper No. 7329, September. http://www.nber.org/papers/w7329.

Goolsbee, Austan. 2006. "The Value of Broadband and the Deadweight Loss of Taxing New Technology." *Contributions to Economic Analysis & Policy* 5. http://faculty.chicagobooth.edu/austan.goolsbee/research/broadb.pdf

Government Accounting Office. 2004. "Human Capital: Opportunities to Improve Federal Continuity Planning Guidance." Report to the Chairman, Subcommittee on Oversight of Government Management, the Federal Workforce, and the District of Columbia, Committee on Governmental Affairs, U.S. Senate, GAO-04-384. http://www.gao.gov/new.items/d04384.pdf.

Greenstein Shane. 1997. "Lock-in and the Costs of Switching Mainframe Computer Vendors: What Do Buyers See?" *Industrial and Corporate Change* 6: 247-273. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Lock-in%20and%20the%20Costs%20of%20Switching%20Mainframe%20Computer%20Vendors%20in% 20the%20US%20Federal%20Government%20in%20the%201970s.pdf.

Greenstein, Shane, and Ed Jimeniz. 2008. "The Emerging Internet Retailing Market as a Nested Diffusion Process." *International Journal of Innovation Management* 2: 281-308.

- Greenstein, Shane, and Eduardo Luzio. 1995. "Measuring the Performance of a Protected Infant Industry: The Case of Brazilian Micro-Computers." *Review of Economics and Statistics* 77: 622-633. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Measuring%20 the%20Performance%20of%20a%20Protected%20Infant%20Industry%20-%20The%20Case%20of%20Brazilian%20Micro-Computers.pdf.
- Greenstein, Shane, and Garey Ramey. 1998. "Market Structure, Innovation, and Vertical Product Differentiation." *International Journal of Industrial Organization* 16, no. 3 (May): 285-311. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Market%20Stru cture,%20Innovation,%20and%20Vertical%20Product%20Differentiation.pdf.
- Greenstein, Shane, and Greg Stranger. 2008. "ISP Pricing in the Shadow of Firm Turnover: ISPs During the 1990s." *International Journal of Industrial Organization* 26: 625-642. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/ISPpricing_and _Entry_final.pdf.
- Greenstein, Shane, and Greg Stranger. 2008. "Pricing at the On-ramp to the Internet: Price Indices for ISPs during the 1990s." *Hard to Measure Goods and Service: Essays in Memory of Zvi Griliches*, edited by Ernst Berndt and Charles Hulten. Chicago: University of Chicago Press, 197-234. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Pricing%20the %20Internet%20for%20Zvi%20book.pdf.

Greenstein, Shane, and Jeff Prince. 2008. "The Diffusion of the Internet and the Geography of the Digital Divide in the United States." *The Oxford Handbook of Information and Communication Technologies*, edited by Danny Quah and Robin Mansell. The Oxford University Press. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/chapters/Internet%20g eography2.pdf.



- Greenstein, Shane, and Jim Wade. 1998. "The Product Life Cycle in the Commercial Mainframe Computer Market, 1968-1983." *Rand Journal of Economics* 29: 772-790. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/The%20Produc t%20Life%20Cycle%20in%20the%20Commercial%20Mainframe%20Computer%20Market,%20196 8-1983.pdf.
- Greenstein, Shane, and Marc Rysman. 2006. "Testing for Agglomeration and Dispersion." *Economic Letters* 86: 405-411.

http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/aggdis23_note.pdf.

Greenstein, Shane, and Mark Rysman. 2007. "Coordination Costs and Standard Setting: Lessons from 56K." *Standards and Public Policy*, edited by Shane Greenstein and Victor Stango. New York: The Cambridge University Press, 123-159. http://www.csio.econ.northwestern.edu/Papers/2004/CSIO-WP-0056.pdf.

Greenstein, Shane, and Mercedes Lizardo. 1999. "Determinants of the Regional Distribution of Information Technology Infrastructure in U.S." In *The Electronic Village: Public Policy Issues of the Information Economy*, edited by Dale Orr and Tom Wilson. Toronto: C.D. Howe Institute. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/chapters/Determinants %20of%20the%20Regional%20Distribution%20of%20Information%20Technology%20Infrastructure %20in%20U.S..pdf.

Greenstein, Shane, and Mike Mazzeo. 2005. "Differentiated Entry into Competitive Telephony." *Journal of Industrial Economics*: 323-350. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/clec_entry_mar ch05.pdf.

Greenstein, Shane, and Pablo Spiller. 1995. "Modern Telecommunications Infrastructure and Economic Activity: An Empirical Investigation." *Industrial and Corporate Change* 4: 647-665. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Modern%20Tel ecommunications%20Infrastructure%20and%20Economic%20Activity%20-%20An%20Empirical%20Investigation.pdf.

Greenstein, Shane, and Ryan C. McDevitt. 2009. "The Broadband Bonus: Accounting for Broadband Internet's Impact on U.S. GDP." *Technology Policy Institute Publication of Notice*, January. http://www.techpolicyinstitute.org/files/greenstein_and_mcdevitt_2009.pdf.

Greenstein, Shane, and Ryan McDevitt. 2010. "Evidence of a Modest Price Decline in US Broadband Services." NBER Working Paper No. w16166, NBER Working Paper Series, July.

Greenstein, Shane, and Tarun Khanna. 1997. "What Does it Mean for Industries to Converge?" Competing in an Age of Digital Convergence, edited by David Yoffie. Cambridge, Massachusetts: Harvard University Press, 201-226.

http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/chapters/What%20Doe s%20Industry%20Convergence%20Mean.pdf

Greenstein, Shane, and Tom Downes. 2002. "Entry into the Schooling Market: How is the Behavior of Private Suppliers Influenced by Public Sector Decisions?" *Bulletin of Economic Research* 54, no. 4: 373-392.

http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Entry%20into% 20the%20Schooling%20Market%20-

%20How%20is%20the%20Behavior%20of%20Private%20Suppliers%20Influenced%20by%20Public%20Sector%20Decisions.pdf.



- Greenstein, Shane, and Tom Downes. 1996. "Understanding the Supply Decisions of Non-Profits: Modeling the Location of Private Schools." *Rand Journal of Economics* 27, no. 2: 365-390. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Understanding %20the%20Supply%20Decisions%20of%20Non-Profits%20-%20Modeling%20the%20Location%20of%20Private%20Schools.pdf.
- Greenstein, Shane M., and Jeffrey Prince. 2006. "The Diffusion of the Internet and the Geography of the Digital Divide in the United States." NBER Working Paper Series, Vol. w12182, May.
- Greenstein, Shane, Mercedes Lizardo, and Pablo Spiller. 2003. "The Evolution of Advanced Large Scale Information Infrastructure in the United States." *Down to the Wire: Studies in the Diffusion and Regulation of Telecommunications Technologies*. New York: Nova Science Publishers. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/chapters/The%20Evol ution%20of%20Advanced%20Large%20Scale%20Scale%20Information%20Infrastructure%20in%2 0the%20United%20States.pdf.

Greenstein, Shane, Neil Gandal, and Dave Salant. 1999. "Adoptions and Orphans in the Early Microcomputer Market: Chickens and Eggs, Hardware and Software." *Journal of Industrial Economics* XLVII, no. 1: 87-105. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Adoptions%20

and%20Orphans%20in%20the%20Early%20Microcomputer%20Market%20-%20Chickens%20and%20Eggs,%20Hardware%20and%20Software.pdf.

Greenstein, Shane, Susan McMaster, and Pablo Spiller. 1995. "The Effect of Incentive Regulation on Local Exchange Companies' Deployment of Digital Infrastructure." *Journal of Economics, Management, and Strategy* 4: 187-236. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Infrastructure.pdf.

Greenstein, Shane. 2000. "Building and Developing the Virtual World: The Commercial Internet Access Market." *Journal of Industrial Economics* 48, no. 4. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Building%20an d%20Delivering%20the%20Virtual%20World%20-%20Commercializing%20Services%20for%20Internet%20Access.pdf.

Greenstein, Shane. 2001. "Commercialization of the Internet: The Interaction of Public Policy and Private Actions." *Innovation, Policy and the Economy*, edited by Adam Jaffe, Josh Lerner and Scott Stern. Cambridge, Massachusetts: MIT Press, 151-186.

http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/chapters/Commercializ ation%20of%20the%20Internet%20-

%20The%20INteraction%20of%20Public%20Policy%20and%20Private%20Choices%20or%20Why%20Introducing%20the%20Market%20Worked%20So%20Well.pdf.

- Greenstein, Shane. 1993. "Did Installed Base Give an Incumbent any (Measurable) Advantages in Federal Computer Procurement?" *Rand Journal of Economics* 24, no. 1: 19-39. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Did%20Installe d%20Base%20Give%20and%20Incumbent%20any%20(Measurable)%20Advantages%20in%20Fe deral%20Computer%20Procurement.pdf.
- Greenstein, Shane. 2008. "Economic Experiments and Neutrality in Internet Access." *Innovation, Policy, and the Economy* 8, edited by Adam Jaffe, Josh Lerner, and Scott Stern. Chicago: University of Chicago Press, 59-110.

http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/exp%20and%2 0neut%20final.pdf.



Greenstein, Shane. 2000. "Empirical Evidence on Commercial Internet Access Providers' Propensity to Offer new Services." In *The Internet Upheaval, Raising Questions and Seeking Answers in Communications Policy*, edited by Benjamin Compaign and Ingo Vogelsang, 253-275. Boston: MIT Press.

http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/chapters/Empirical%20 Evidence%20on%20Advanced%20Services%20at%20Commercial%20Internet%20Access%20Prov iders.pdf.

Greenstein, Shane. 2000. "Framing Empirical Research on the Evolving Structure of Commercial Internet Markets." In *Understanding the Digital Economy: Data, Tools and Research*, edited by Erik Brynolfsson and Brian Kahin, 151-184. Cambridge, Massachusetts: MIT Press. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/understanding. pdf.

Greenstein, Shane. 1997. "From Super-minis to Super Computers: Estimating Surplus in the Computing Market." In *The Economics of New Goods*, edited by Tim Bresnahan and Robert J. Gordon, 329-362. Chicago: University of Chicago Press.

http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/chapters/From%20Sup er-minis%20to%20Super%20Computers%20-

%20Estimating%20Surplus%20in%20the%20Computing%20Market.pdf.

Greenstein, Shane. 1999. "Industrial Convergence." In *The Technology Management Handbook*, edited by Richard Dorf. CRC Press. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Technology%2 0Management.pdf.

- Greenstein, Shane. 1998. "Industrial Economics and Strategy: Computing Platforms." *IEEE Micro: Chips, Systems, and Applications, Special issue on Standardization* 18, no. 3: 43-53. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Industrial%20E conomics%20and%20Strategy%20-%20Computing%20Platforms.pdf.
- Greenstein, Shane. 2008. "Innovation and the Evolution of Market Structure for Internet Access in the United States." In *The Internet and American Business*, edited by William Aspray and Paul E. Ceruzzi. Cambridge, Massachusetts: MIT Press, 47-104.

Greenstein, Shane. 1996. "Invisible Hands versus Invisible Advisors: Coordination Mechanisms in Economic Networks." *Public Networks, Public Objectives*, edited by Eli Noam and Aine Nishuilleabhain. Amsterdam, Netherlands: Elsevier Science B. V., 135-160. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/chapters/Invisible%20 Hands%20versus%20Invisible%20Advisors%20-%20Coordiantion%20Mechanisms%20in%20Economic%20Networks.pdf

- Greenstein, Shane. 1995 "Lock-in and the Costs of Switching Mainframe Computer Vendors in the US Federal Government in the 1970s." *IEEE Annals of the History of Computing* 17, no. 3 (Winter): 58-66. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Lock-in%20and%20the%20Costs%20of%20Switching%20Mainframe%20Computer%20Vendors%20in% 20the%20US%20Federal%20Government%20in%20the%201970s.pdf
- Greenstein, Shane. 1993. "Markets, Standardization, and the Information Infrastructure." *IEEE Micro, Chips, Systems, and Applications, Special Issue on Standards* 13, no. 6 (December): 36-51. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Markets,%20St andardization,%20and%20the%20Information%20Infrastructure.pdf.
- Greenstein, Shane. 2010. "Standardization, Complexity and Transitional Technology Markets." *The Standards Edge*, edited by Sherrie Bolin and Carl Cargill. Ann Arbor, MI: Sheridan Books. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/chapters/Standards%2 0&%20complexity.pdf.



- Greenstein, Shane. 2001. "Technological Mediation and Commercial Development in the Early Internet Access Market." *California Management Review* 43: 75-94. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Technological %20Mediation%20and%20Commercial%20Development%20in%20the%20Early%20Internet%20Ac cess%20Market.pdf.
- Greenstein, Shane. 2004. "Technological Mediation as Strategy: The Case of ISPs." *Next Generation Business Handbook*, edited by Subir Chowdhury. New York: John Wiley and Sons, 1128-1156. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/chapters/Technological %20Mediation%20as%20Strategy.pdf.

Greenstein, Shane. 2005. "The Economic Geography of Internet Infrastructure in the United States." *Handbook of Telecommunication Economics Volume II*, edited by Martin Cave, Sumit Majumdar, and Ingo Vogelsang. New York: Elsevier, 289-364. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/geography.pdf.

Greenstein, Shane. 1990. "Creating Economic Advantage by Setting Compatibility Standards: Can Physical Tie-ins Extend Monopoly Power?" *Economics of Innovation and New Technology* 1: 63-84. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Creating%20E conomic%20Advantage%20by%20Setting%20Compatibility%20Standards%20-%20Can%20Physical%20Tie-ins%20Extend%20Monopoly%20Power.pdf.

Greenstein, Shane. 1992. "Invisible Hands and Visible Advisors: An Economic Evaluation of Network Standardization." *Journal of the American Society for Information Science* 43, no. 8 (September): 538-549.

http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Invisible%20Ha nds%20and%20Visible%20Advisors%20-%20An%20Economic%20Evaluation%20of%20Network%20Standardization.pdf

Greenstein, Shane. 1993. "Procedural Rules and Procurement Regulations: Complexity Creates Tradeoffs." *Journal of Law, Economics, and Organizations* 9, no. 1 (April): 159-180. http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Procedural%20 Rules%20and%20Procurement%20Regulation%20-%20Complexity%20Creates%20Trade-offs.pdf.

Greenstein, Shane. 1995. "Sole-Sourcing versus Competitive Bidding: U.S. Government Agencies' Procedural Choices for Mainframe Computer Procurement." *Journal of Industrial Economics XLIII*, no. 2 (June): 125-140.

http://www.kellogg.northwestern.edu/faculty/greenstein/images/htm/Research/articles/Sole-Sourcing%20versus%20Competitive%20Bidding%20-

%20U.S.%20Government%20Agencies'%20Procedural%20Choices%20for%20Mainframe%20Com puter%20Procurement.pdf.

- Grubesic, Tony H. and Elizabeth A. Mack. 2009. "Forecasting Broadband Provision." *Information Economics and Policy 21*: 297-311.
- Grubesic, Tony H. 2006. "A Spatial Taxonomy of Broadband Regions in the United States." *Information Economics and Policy 18*: 423-448.
- Grumbach, Kevin, Gary Hart, Elizabeth Mertz, Janet Coffman, and Lorella Palazzo. 2003. "Who is Caring for the Underserved? A Comparison of Primary Care Physicians and Nonphysician Clinicians in California and Washington." *Annals of Family Medicine* 1, no.2 (July): 97-104. http://www.ncbi.nlm.nih.gov/pmc/articles/PMC1466573/.

Hafner, Katie. 2000. "In Praise of the Mom-and-Pop ISP." *New York Times*, August 17. http://www.nytimes.com/2000/08/17/technology/in-praise-of-the-mom-and-pop-isp.html.


- Haire, Andrew J. 2009. "Delivering the Next Generation Network: The Singapore Model." Presented by Andrew J. Haire, Dy Director-General, IDA Singapore, October 23.
- Habib, Haman, and Sid-Ahmed Selouani. 2007. "Social Impact of Broadband Internet: A Case Study in the Shippagan Area, a Rural Zone in Atlantic Canada." *Journal of Information, Information Technology, and Organizations*, Vol. 2: 79-94.
- Hargittai, Eszter, and N. Zillien. 2009. "Digital Distinction: Status-Specific Internet Uses." *Social Science Quarterly* 90: 274-291. http://www.webuse.org/pdf/ZillienHargittai-DigitalDistinction2009.pdf.
- Hargittai, Eszter, and Y. P. Hsieh. 2010. "Predictors and Consequences of Social Network Site Usage." *Information, Communication & Society* 13, no. 4: 515-536. http://www.webuse.org/pdf/HargittaiHsiehICS10.pdf.
- Hargittai, Eszter, and G. Walejko. 2008. "The Participation Divide: Content Creation and Sharing in the Digital Age." *Information, Communication and Society* 11, no. 2: 239-256. http://www.webuse.org/pdf/HargittaiWalejko-ParticipationDivide2008.pdf.
- Hargittai, Eszter and S. Shafer. 2006. "Differences in Actual and Perceived Online Skills: The Role of Gender." *Social Science Quarterly* 87, no. 2 (June): 432-448.http://webuse.org/pdf/HargittaiShafer-GenderSkill2006.pdf.
- Hargittai, Eszter, and A. Hinnant. 2008. "Digital Inequality: Differences in Young Adults' Use of the Internet." *Communication Research* 35, no. 5: 602-621. http://www.webuse.org/pdf/HargittaiHinnant-DigitalInequality2008.pdf.
- Hargittai, Eszter, and M.A. Centeno. 2001. "Defining a Global Geography." *American Behavioral Scientist* 10, no. 44: 545-1560. http://www.webuse.org/pdf/HargittaiCenteno-GlobalGeography2001.pdf.
- Hargittai, Eszter and Y. P. Hsieh. 2010. "From Dabblers to Omnivores: A Typology of Social Network Site Usage." *A Networked Self*, edited by Z. Papacharissi. London: Routledge. http://www.webuse.org/pdf/HargittaiHsieh-SNSTypology.pdf.
- Hargittai, Eszter, E. More, and J. Pasek. 2009. "Facebook and Academic Performance: Reconciling a Media Sensation with Data." *First Monday* 14, no. 5. http://webuse.org/pdf/PasekmoreHargittai-FacebookFM09.pdf.
- Hargittai, Eszter, F. Fullerton, E. Menchen-Trevino, and K. Thomas. 2010. "Trust Online: Young Adults' Evaluation of Web Content." *International Journal of Communication* 4: 468-494. http://www.webuse.org/pdf/HargittaiEtAlTrustOnlineIJoC10.pdf.
- Hargittai, Eszter. 2007. "A Framework for Studying Differences in People's Digital Media Uses." *Cyberworld Unlimited*, edited by Nadia Kutscher and Hans-Uwe Otto. VS Verlag für Sozialwissenschaften/GWV Fachverlage GmbH, 121-137. http://www.webuse.org/pdf/Hargittai-FrameworkDigitalMediaUse2007.pdf.
- Hargittai, Eszter. 2009. "An Update on Survey Measures of Web-Oriented Digital Literacy." *Social Science Computer Review* 27, no. 1 (Winter): 130-137. http://www.webuse.org/pdf/Hargittai-UpdateSurveyMeasures2009.pdf.
- Hargittai, Eszter. 2002. "Beyond Logs and Surveys: In-Depth Measures of People's Online Skills." *Journal of the American Society for Information Science and Technology* 53, no. 14: 1239-1244. http://www.webuse.org/pdf/Hargittai-BeyondLogsSurveys2002.pdf.
- Hargittai, Eszter. 2004. "Classifying and Coding Online Actions." *Social Science Computer Review* 22, no. 2: 210-227. http://www.webuse.org/pdf/Hargittai-ClassifyingCoding2004.pdf.



- Hargittai, Eszter. 2010. "Digital Na(t)ives? Variation in Internet Skills and Uses among Members of the Net Generation." *Sociological Inquiry* 80, no. 1. http://www.webuse.org/pdf/Hargittai-DigitalNativesSI2010.pdf.
- Hargittai, Eszter. 2004. "Do you "Google"? Understanding Search Engine Popularity Beyond the Hype." *First Monday 9*, no. 3. http://webuse.org/pdf/Hargittai-GoogleFM04.pdf.
- Hargittai, Eszter. 2004. "Internet Access and Use in Context." *New Media and Society* 6, no. 1: 137-143. http://www.webuse.org/pdf/Hargittai-InternetAccessUseContext2004.pdf.
- Hargittai, Eszter. 2003. "Minimizing Online Hurdles to Accessing Government Information." *IT & Society* 1, no. 3: 27-41. http://www.webuse.org/pdf/Hargittai-AccessingGovernmentInformation2003.pdf
- Hargittai, Eszter. 2000. "Radio's Lessons for the Internet." *Communications of the ACM* 43, no. 1: 50-56. http://www.webuse.org/pdf/Hargittai-Radio2000.pdf.
- Hargittai, Eszter. 2002. "Second-Level Digital Divide: Differences in People's Online Skills." *First Monday* 7, no. 4. http://webuse.org/pdf/Hargittai-SecondLevelFM02.pdf.
- Hargittai, Eszter. 2005. "Survey Measures of Web-Oriented Digital Literacy." Social Science Computer Review 23, no. 3 (Fall): 371-379. http://www.webuse.org/pdf/Hargittai-SurveyMeasures2005.pdf.
- Hargittai, Eszter. 2003. "The Changing Online Landscape: From Free-for-All to Commercial Gatekeeping." *Community Practice in the Network Society: Local Action/Global Interaction*, edited by P. Day and D.Schuler. London: Routledge.
- Hargittai, Eszter. 2003. "The Digital Divide and What To Do About It." In *The New Economy Handbook*, edited by D. C. Jones. San Diego, CA: Academic Press. http://www.webuse.org/pdf/Hargittai-DigitalDivideWhatToDo2007.pdf.
- Hargittai, Eszter. 2008. "The Digital Reproduction of Inequality." *Social Stratification*, edited by David Grusky. Boulder, CO: Westview Press, 936-944. http://webuse.org/pdf/Hargittai-DigitalReproduction2008.pdf.
- Hargittai, Eszter. 2007. "The Social, Political, Economic, and Cultural Dimensions of Search Engines: An Introduction." *Journal of Computer-Mediated Communication* 12, no. 3. http://webuse.org/pdf/Hargittai-SocialPoliticalEconomicJCMC07.pdf.
- Hargittai, Eszter. 1999. "Weaving the Western Web: Explaining Differences in Internet Connectivity Among OECD Countries." *Telecommunications Policy* 23, no. 10/11: 701-718. http://www.webuse.org/pdf/Hargittai-WeavingWesternWeb1999.pdf.
- Hargittai, Eszter. 2007. "Whose Space? Among Users and Non-Users of Social Network Sites." *Journal of Computer-Mediated Communication* 13, no. 1. http://webuse.org/pdf/Hargittai-WhoseSpaceJCMC07.pdf.
- Harris, Blake. 2010. "Broadband's Economic Impact." *Government Technology*, February 26. http://www.govtech.com/wireless/Broadbands-Economic-Impact-Opinion.html.
- Hauge, Janice and James E. Prieger. 2009. "Demand-Side Programs to Stimulate Adoption of Broadband: What Works?" Working paper, October 14. http://papers.ssrn.com/sol3/papers.cfm?abstract_id=1492342.
- Haughwout, Andrew F. 1999. "New Estimates of the Impact of EDA Public Works Program Investments on County Labor Markets." *Economic Development Quarterly* 13: 371-382.
- Hausman, Jerry A. 1997. "Valuing the Effect of Regulation on New Services in Telecommunications." *Brookings Papers: Microeconomics*, 1-54. http://pareto.uab.es/ftrillas/docencia/Hausman.pdf.



- Henderson, Jason. 2002. "Building the Rural Economy with High-Growth Entrepreneurs." *Federal Reserve Bank of Kansas City Economic Review* 3. http://www.kc.frb.org/PUBLICAT/ECONREV/PDF/3q02hend.pdf.
- Industry Analysis and Technology Division. 2009. "High-Speed Services for Internet Access: Status as of December 31, 2007." Federal Communications Commission. Article published January.
- Hillestad, Richard, James Bigelow, Anthony Bower, Federico Girosi, Robin Meili, Richard Scoville, and Roger Taylor. 2005. "Can Electronic Medical Record Systems Transform Health Care? Potential Health Benefit, Savings, and Costs." *Health Affairs*, September-October.
- Holly, Krisztina. 2009. "IMPACT: Innovation Model Program for Accelerating the Commercialization of Technologies—A Proposal for Realizing the Economic Potential of University Research." Working paper, University of Southern California, August.
- Horrigan, John B. 2010. "Broadband Adoption and Use in America." OBI Working Paper Series No. 1, Washington, D.C., Federal Communications Commission. http://hraunfoss.fcc.gov/edocs_public/attachmatch/DOC-296442A1.pdf.
- Horrigan, John B. 2008. "Is Home Broadband Adoption Slowing?" Commentary. Washington, D.C.: *Pew Internet and American Life Project*. http://www.pewinternet.org/Commentary/2008/September/Is-Home-Broadband-Adoption-Slowing.aspx.
- Horrigan, John, Chandler Stolp and Robert H. Wilson. 2005. "Broadband Utilization in Space: Effects of Population and Economic Structure." *The Information Society* 22: 341-354.
- Horrigan, John. 2009. "Home Broadband Adoption 2009: Broadband Adoption Increases, But Monthly Prices Do Too." *Pew Research Center*. http://www.pewinternet.org/~/media//Files/Reports/2009/Home-Broadband-Adoption-2009.pdf.
- Hu, Wei-Min and James E. Prieger. 2006. "The Timing of Broadband Provision: The Role of Competition and Demographics." AEI-Brookings Joint Center Working Paper, No. 07-06.
- Hudson, Heather and Edwin B. Parker. 1992. *Electronic Byways: State Policies for Rural Development through Telecommunications*. Boulder: Westview Press.
- Hudson, Heather, Don Dillman, Edwin Parker, and A. Roscoe. 1990. "Rural America in the Information Age: Telecommunications Policy for Rural Development." *Telecommunications Policy* 12 (June): 193-205.
- Hudson, Heather. 1995. "Economic and Social Benefits of Rural Telecommunications: A Report to the World Bank." McLaren School of Business, University of San Francisco.
- Hudson, Heather. 1997. *Global Connections: International Telecommunications Infrastructure and Policy.* New York: Van Nostrand Reinhold.
- Hulbert, Linda Ann and McBride, Regina C. 2004. "Utilizing Videoconferencing in Library Education: A Team Teaching Approach." *Journal of Education for Library and Information Services*, Winter. Association for Library and Information Science Education.
- Initiative for a Competitive Inner City. "State of the Inner City Economies: Small Businesses in the Inner City." Small Business Administration Office of Advocacy. http://www.sba.gov/advo/research/rs260tot.pdf.
- International Telecommunications Union. 2007. "Chapter 2: Bridging the Digital Divide." In *World Information Society Report*. International Telecommunications Union. .http://www.itu.int/osg/spu/publications/worldinformationsociety/2007/WISR07-chapter2.pdf.



- Internet2. 2009. "Research and Commercial Network: Capacity at U.S. Research Universities." Report for the National Science Foundation. http://www.internet2.edu/government/files/200911-IS-NSF-survey3.pdf.
- Ismail, Sherille, and Irene Wu. 2003. "Broadband Internet Access in OECD Countries: A Comparative Analysis." A Staff Report of the Office of Strategic Planning and Policy Analysis and International Bureau.
- Isserman, Andrew M. and John D. Merrifield. 1987. "Quasi-Experimental Control Group Methods for Regional Analysis: An Application to an Energy Boomtown and Growth Pole Theory." *Economic Geography* 63, no. 1 (January): 3-19.
- Ito, Mizuko, Heather Horst, Matteo Brittanit, Danah Boyd, Becky Herr-Stephenson, Patricia G.Lange, C.J. Pascoe, and Laura Robinson. 2008. "Living and Learning with New Media Summary of Findings from the Digital Youth Project." Article published November. http://digitalyouth.ischool.berkeley.edu/files/report/digitalyouth-WhitePaper.pdf.
- Jackson, Kevin M., OD, MPH; Karen E. Scott, MD, MBA, Joshua Graff Zivin, PhD; David A. Bateman, MD, John T. Flynn, MD; Jeremy D. Keenan, MD, MPH, Michael F. Chiang, M. "Cost-Utility Analysis of Telemedicine and Ophthalmoscopy for Retinopathy of Prematurity Management." Arch Ophthalmol 126, no. 4. http://www.v2020la.org/pub/PUBLICATIONS_BY_TOPICS/ROP/Cost-Utility%20Analysis%20of%20Telemedicine....pdf.
- Jackson, Linda A., Alexander von Eye, Frank A. Biocca, Gretchen Barbatsis, Yong Zhao, and Hiram E. Fitzgerald. 2006. "Does Home Internet Use Influence the Academic Performance of Low-Income Children." *Developmental Psychology* 42, no. 3: 429-435.

Jackson, Steven J., Archer L. Batcheller, Paul N. Edwards, Geoffrey C. Bowker, Steve Cisler, and Susan Leigh Star. 2008. Extending African Knowledge Infrastructures: Sharing, Creating, Maintaining. Report for the World Bank, March. http://deepblue.lib.umich.edu/bitstream/2027.42/61201/1/Jackson%20et%20al,%20Extending%20Afri can%20Knowledge%20Infrastructures%20(March%202008).pdf.

- Jackson, Steven J., Paul N. Edwards, Geoffrey C. Bowker, and Cory P. Knobel. 2007. "Understanding Infrastructure: History, Heuristics, and Cyberinfrastructure Policy." *First Monday* 12, no. 6 (June). http://firstmonday.org/issues/issue12_6/jackson/index.html.
- Jackson, Steven. 2006. "Water Models and Water Politics: Deliberative Design and Virtual Accountability." Paper presented at DG.O2006: Proceedings of the 2006 Digital Government Conference.
- Jaffe, Justin. 2009. "SMB Cluster Analysis: SMB 2.0s Lead the Way Toward Next-Generation Technology." International Data Corporation.
- Jayakar, Krishna, Amit Schejter, and Richard Taylor. 2010. "Small Businesses and Broadband: Key Drivers for Economic Recovery." Working paper, Institute for Information Policy at Pennsylvania State University. http://comm.psu.edu/about/centers/institute-for-information-policy/smallbusiness.pdf.
- Jha, M.D., M.P.H., Ashish K., Catherine M. DesRoches, Dr.Ph., Eric G. Campbell, Ph.D., Karen Donelan, Sc.D., Sowmya R. Rao, Ph.D., Timothy G. Ferris, M.D., M.P.H., Alexandra Shields, Ph.D., Sara Rosenbaum, J.D., and David Blumenthal, M.D., M.P.P. 2009. "Use of Electronic Health Records in U.S. Hospitals." *New England Journal of Medicine* 360, no. 16 (March): 1628-38. http://www.nejm.org/doi/pdf/10.1056/NEJMsa0900592.
- Johnson, David R., Susan P. Crawford, and John G. Palfrey, Jr. 2004. "The Accountable Net: Peer Production of Internet Governance." Draft, Virginia Journal of Law and Technology. http://cyber.law.harvard.edu/Accountable%20Net-ITU.pdf.



- Jones, Sidney, and Mary Madden. 2008. "Networked Workers." *Pew Research Center: Internet & American Life Project.* http://www.pewinternet.org/~/media//Files/Reports/2008/PIP Networked Workers FINAL.pdf.pdf.
- Jorgenson, Dale. 2001. "Information Technology and the U.S. Economy." *American Economic Review* 91, no. 1 (March).
- Julius Genachowski. 2010. "Broadband: Our Enduring Engine for Prosperity and Opportunity." Remarks at the 2010 National Association of Regulatory Utility Commissioners Conference, February.
- Katz, R.L., P. Zenhausern, and S. Suter. 2008. "An Evaluation of Socio-Economic Impact of a Fiber Network in Switzerland." Mimeo, Polynomics and Telecom Advisory Services, LLC.
- Katz, R.L. 2009. "The Economic and Social Impact of Telecommunications Output: a Theoretical Framework and Empirical Evidence for Spain." *Intereconomics* 1, January-February.
- Katz, Raul and Stephan Suter. 2009. "Estimating the Economic Impact of the Broadband Stimulus Plan." Working Paper Columbia Institute for Tele-Information.
- Katz, Raul L.; Vaterlaus, Stephan; Zenhäusern, Patrick and Suter, Stephan. 2010. "The Impact of Broadband on Jobs and the German Economy." *Intereconomics*. http://www.teleadvs.com/images/Katz,_final.pdf.
- Kelley, Doris J. 2003. "A Study of the Economic and Community Benefits of Cedar Falls, Iowa's Municipal telecommunications Network." Iowa Association of Municipal Utilities. Last modified July 6, 2004. http://baller.com/pdfs/cedarfalls_white_paper.pdf
- Kendall, John S., Courtney Pollack, Amitra Schwols, and Christina Snyder. 2007. *High School Standards and Expectations for College and the High-Skills Workplace*. Denver, Colorado: REL Central. http://www.mcrel.org/pdf/standards/20090107_REL_HighSchBrief.pdf.
- Kim, Heekyung Hellen, Jae Yun Moon, and Shinkyu Yang. 2004. "Broadband Penetration and Participatory Politics: South Korea Case." Proceedings of the 37th Hawaii International Conference on System Sciences. http://csdl.computer.org/comp/proceedings/hicss/2004/2056/05/205650117b.pdf.
- King, Kay. 2010. "How Congress Erodes National Security." *Council on Foreign Relations*, interviewed by Robert McMahon, November 16. http://www.cfr.org/publication/23412/how_congress_erodes_national_security.html.

Kirkpatrick, David. 2008. "Muhammad Yunus on Tech, Profit and the Poor." Fortune Magazine, April 4.

- Kistner, Toni. 2004. "Fighting for Fair Telework Tax." *Network World*, June 7. http://www.networkworld.com/net.worker/news/2004/0607netlead.html.
- Kohl, Benjamin A., Frank D Sites, Jacob T Gutsche, and Patrick Kim. 2007. "Economic Impact of eICU Implementation in an Academic Surgical ICU." *Critical Care Medicine* 35, no. 12.
- Kolko, Jed. 2010. "Does Broadband Boost Local Economic Development?" *Public Policy Institute of California*. Article published January. http://www.ppic.org/content/pubs/report/R_110JKR.pdf.
- Kotval, Zenia. 1999. "Telecommunications: A Realistic Strategy for the Revitalization of American Cities." *Cities 16*, no. 1: 33-41.
- Koutroumpis, Pantelis. 2008. "Broadband Infrastructure and Economic Growth: A Simultaneous Approach." Mimeo, Imperial College, London, United Kingdom.



- Lam, P-L., and A. Shiu. 2008. "Causal Relationship Between Telecommunications and Economic Growth: a Study of 105 Countries." The 17th Biennial Conference of the International Telecommunications Society, Montreal, Quebec, June 24-27.
- LaRose, Robert, Jennifer L. Gregg, Sharon Stover, Joseph Straubhaar, and Nobuya Inagaki. 2008. "Closing the Rural Broadband Gap." Final Technical Report, Michigan State University. Report published November 30. https://www.msu.edu/~larose/ruralbb/.
- Larson, EH, LG Hart, and J Hummel. 1994. "Rural Physician Assistants: a Survey of Graduates of MEDEX Northwest." *Public Health Reports*, 266–274.
- Laura Breeden, Steve Cisler, Vivian Guilfoy, Michael Roberts, and Antonia Stone. 1998. "Computer and Communications Use in Low-Income Communities." Report for the Annie E. Casey Foundation, Education Development Center, Newton, MA. http://www.ctcnet.org/resources/reports/casey/index.htm.
- Laura D'Amico. 2005. "The Center for Learning Technologies in Urban Schools: A Case of Design-Based Research in Education." *Meta Study Case Report*, Simon Fraser University. Last modified July. http://www.sfu.ca/~ldamico/LeTUS_FullCase_Final.pdf.
- LECG Ltd. 2009. *Economic Impact of Broadband: An Empirical Stu*dy. London: LECG. http://www.connectivityscorecard.org/images/uploads/media/Report_BroadbandStudy_LECG_March 6.pdf
- Lehr, William, and Lee McKnight. 2000. "A Broadband Access Market Framework: Towards Consumer Service Level Agreements." Paper presented to the Twenty-Eighth Annual Telecommunications Policy Research Conference, Alexandria, Virginia, September 25-27. http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.31.9089&rep=rep1&type=pdf.
- Lehr, William, and Lee McKnight. 2000. "A Broadband Access Market Framework: Towards Consumer Service Level Agreements." Paper 121, Center for E-Business at MIT, Cambridge, Massachusetts, September.

http://ebusiness.mit.edu/research/papers/121%20Lehr,%20McKnight,%20Broadband%20Access.pdf

- Lehr, William, and Sharon Gillett, Marvin Sirbu, and Carlos Osorio. 2005. "Measuring Broadband's Economic Impact." Paper presented at the 33rd Annual Telecommunications Policy Research Conference, George Mason University, September. http://www.broadbandproperties.com/2005issues/dec05issues/Measuring%20Broadband%20Eco%2 0Impact,%20Lehr,%20Gilett,%20Sirbu.pdf.
- Lehr, William, and Tony Smith-Grievo. 2009. "Measurement and Assessment of Broadband Availability." Report prepared for John Adams Innovation Institute of the Massachusetts Technology Collaborative, January. http://people.csail.mit.edu/wlehr/Lehr-Papers_files/LehrJan09%20bb%20availability.pdf.
- Lehr, William, David Clark, Peyman Faratin, Rahul Sami, and John Wroclawski. 2006. "Overlay Networks and Future of the Internet." *Communications and Strategies*, no. 63, 3rd Quarter: 1-21. http://people.csail.mit.edu/wlehr/Lehr-Papers_files/Clark%20Lehr%20et%20al%20Overlays%20CandS%20Sept2006.pdf.
- Lehr, William, David Clark, Peyman Faratin, Rahul Sami, and John Wroclawski. 2005. "The Growth of Internet Overlay Networks: Implications for Architecture, Industry Structure and Policy." Paper presented at the 33rd Annual Telecommunications Policy Research Conference, George Mason University, September. http://web.si.umich.edu/tprc/papers/2005/466/TPRC_Overlays_9_8_05.pdf.



- Lehr, William, Kenneth Flamm, John Horrigan, and Amy Friedlander. 2007. "Measuring Broadband: Improving Communications Policymaking through Better Data Collection." *Pew Research Center: Internet and American Life Project,* November. http://people.csail.mit.edu/wlehr/Lehr-Papers_files/PIP_Measuring%20Broadband.pdf.
- Lehr, William, Marvin Sirbu, Jon Peha, and Sharon Gillette. 2006. "Scenarios for the Network Neutrality Arms Race." Paper prepared for the 34th Annual Telecommunications Policy Research Conference, George Mason University, September. http://web.si.umich.edu/tprc/papers/2006/561/TPRC2006_Lehr%20Sirbu%20Peha%20Gillett%20Net %20Neutrality%20Arms%20Race.pdf.
- Lehr, William, P. Faratin, D. Clark, P. Gilmore, and A. Berger. 2007. "Complexity of Internet Interconnections: Technology, Incentives and Implications for Policy." Paper prepared for 35th Annual Telecommunications Policy Research Conference, George Mason University, September. http://people.csail.mit.edu/wlehr/Lehr-

Papers_files/Clark%20Lehr%20Faratin%20Complexity%20Interconnection%20TPRC%202007.pdf.

- Lehr, William, Stephen Bauer, and David Clark. 2009. "Broadband Microfoundations: The Need For Traffic Data." Invited paper prepared for Beyond Broadband Access conference, NewAmerica Foundation, Washington DC, September 22-24. http://people.csail.mit.edu/wlehr/Lehr-Papers_files/Bauer_Clark_Lehr_Traffic.pdf.
- Lehr, William, Stephen Bauer, and David Clark. 2010. "Understanding Broadband Speed Measurements." Paper prepared for 38th Research Conference on Communication, Information and Internet Policy, Arlington, Virginia, September. http://people.csail.mit.edu/wlehr/Lehr-Papers_files/bauer_clark_lehr_2010.pdf

Lehr, William, Tom Vest, and Eliot Lear. 2008. "Running on Empty: the challenge of managing Internet addresses." Prepared for the 36th Research Conference on Communication, Information, and Internet Policy, George Mason University, Arlington, Virginia, September 26-28. http://people.csail.mit.edu/wlehr/Lehr-Papers_files/Lehr%20Lear%20Vest%20TPRC08%20Internet%20Address%20Running%20on%20E

- mpty.pdf. Lehr, William, Tony Smith-Grieco, and Grace Rusi Woo. 2008. "Broadband Metrics Best Practices: Bariow and Assessment " Bapart propared for Massachusette Technology Collaborative (John
- Review and Assessment." Report prepared for Massachusetts Technology Collaborative/John Adams Innovation Institute (MTC/JAII), February. http://people.csail.mit.edu/wlehr/Lehr-Papers_files/Broadband%20Metrics%20Best%20Practices%20Survey%20Feb08.pdf.
- Lehr, William, with Steven Bauer, and David Clark. 2009. "The Evolution of Internet Congestion." Paper prepared for 37th Research Conference on Communication, Information and Internet Policy, Arlington, Virginia, September. http://people.csail.mit.edu/wlehr/Lehr-Papers_files/Bauer_Clark_Lehr_2009.pdf.
- Lehr, William, and Frank R. Lichtenberg. 1998. "Computer Use and Productivity Growth in Federal Government Agencies, 1987-92." *Journal of Industrial Economics* XLVI, no. 2 June: 257-279. http://www.nber.org/papers/w5616.
- Lennett, Benjamin, Chiehyu Li, Dan Meredith, James Losey, Robb Topolski, and Sascha Meinrath. 2009. "Broadband Truth-in-Labeling." *New America Foundation*, September 23. http://oti.newamerica.net/publications/policy/broadband_truth_in_labeling.
- Litan, Robert. 2005. "Great Expectations: Potential Economic Benefits to the Nation from Accelerated Broadband Deployment to Older Americans and Americans with Disabilities." *New Millennium Research Council*, December.



- Litan, Robert. 2008. "Vital Signs via Broadband: Remote Health Monitoring Transmits Savings, Enhances Lives." *Better Health Care Together*, October 24. http://www.corp.att.com/healthcare/docs/litan.pdf.
- Lohrke, Franz T., Geralyn McClure Franklin, and Cynthia Frownfelter-Lohrke. 2006. "The Internet as an Information Conduit: A Transaction Cost Analysis Model of U.S. SME Internet Use." *International Small Business Journal* 24, no. 2: 159-178.
- Losen, Daniel, Gary Orfield, Christopher B. Swanson and Johanna Wald. 2004. "Losing Our Future: How Minority Youth are Being Left Behind by the Graduation Rate Crisis." *The Civil Rights Project at Harvard University*, Cambridge, Massachusetts. http://www.urban.org/UploadedPDF/410936_LosingOurFuture.pdf.
- Losh, Susan Carol. 2009. "Generation versus Aging, and Education, Occupation, Gender and Ethnicity Effects in U.S. Digital Divides." In *The Proceedings, Atlanta Conference on Science and Innovation Policy*, October 2-3. http://mailer.fsu.edu/~slosh/LoshGenerationVsAging.doc.
- Louch, H., Eszter Hargittai, and M. A. Centeno. 1999. "Phone Calls and Fax Machines: The Limits to Globalization." *The Washington Quarterly* 22, no. 2: 83-100. http://www.webuse.org/pdf/LouchHargittaiCenteno-PhoneCalls1999.pdf.
- Lovett, Marsha, Oded Meyer, and Candace Thille. 2008. "The Open Learning Initiative: Measuring the Effectiveness of the OLI Statistics Course in Accelerating Student Learning." Special issue, *Journal of Interactive Media in Education: Researching Open Content in Education.* http://jime.open.ac.uk/jime/article/view/2008-14.
- Low, Sarah, and Peter Stenberg. 2009. "Rural Broadband at a Glance." Working paper, United States Department of Agriculture. http://www.ers.usda.gov/publications/eib47/eib47.pdf.
- Lynch, Clifford. 2004. "The Future of Special and Distinctive Collections in the Digital Age." Coalition for Networked Information Video. Last modified February 24. https://scholarspace.manoa.hawaii.edu/bitstream/10125/16324/6/100224_lynch_future-special.html.
- Madden, Gary, Scott Savage, and Michael Simpson. 1996. "Information Inequality and Broadband Network Access: An Analysis of Australian Household Survey Data." *Industrial and Corporate Change* 5, no. 4: 1049-1056.
- Mainous, Ag, JG Bertolino, and PL Harrell. 1992. "Physician Extenders: Who is Using Them?" *Family Medicine* 24:201–204.
- Majumdar, S. K., O. Carare, and H. Chang. 2010. "Broadband Adoption and Firm Productivity: Evaluating the Benefits of General Purpose Technology." *Industrial and Corporate Change* 19, no. 3.
- Martin, Randolph C., and Robert E. Graham, Jr. 1980. "The Impact of Economic Development Administration Programs: Some Empirical Evidence." *The Review of Economics and Statistics* 62, no. 1 (February): 52-62.
- Matson, Malcolm, and Rod Mitchell. 2006. "Study On Local Open Access Networks For Communities and Municipalities." A study on behalf of The Oplan Foundation.
- McKinsey & Company. 2009. "Unlocking Energy Efficiency in the U.S. Economy." Last modified July 2009.

http://www.mckinsey.com/en/Client_Service/Electric_Power_and_Natural_Gas/Latest_thinking/Unloc king_energy_efficiency_in_the_US_economy.aspx.



- McKinsey & Company. 2009. "The Economic Impact of the Achievement Gap in America's Schools." Summary of findings, Social Sector Office. Last modified April 2009. http://www.mckinsey.com/App_Media/Images/Page_Images/Offices/SocialSector/PDF/achievement_ gap_report.pdf.
- McKinsey Quarterly Business Technology Office. 2009. "How Companies Are Benefiting From Web 2.0." *McKinsey Quarterly*, September. https://www.mckinsey.guarterly.com/Business_Technology/BT_Strategy/How_companies_are_benefit

https://www.mckinseyquarterly.com/Business_Technology/BT_Strategy/How_companies_are_benefiting_from_Web_20_McKinsey_Global_Survey_Results_2432.

- Mehra, Bharat, Cecelia Merkel, Ann P. Bishop. 2004. "The Internet for Empowerment of Minority and Marginalized Users." *New Media and Society* 6: 781–802. doi:10.1177/146144804047513.
- Memorandum for the Heads of Executive Departments and Agencies. 2009. M-10-01 Executive Office of the President. http://www.whitehouse.gov/sites/default/files/omb/assets/memoranda_2010/m10-01.pdf.
- Michael B. Horn. 2009. "Disrupting Class: How Disruptive Innovation Will Change the Way the World Learns." Mountain View, CA: Innosight Institute. http://net.educause.edu/ir/library/pdf/ECR0903.pdf.

Miller, Doriane C, MD. 2010. "Health and Disease Management Outside the Clinic Doors: There's an app for that." Institute of Medicine. Last modified April. http://www.iom.edu/~/media/Files/Activity%20Files/Quality/VSRT/2010-APR-01/9-Health%20and%20disease%20management%20outside%20the%20clinic%20doors.ashx.

- Mills, Karen, Elisabeth B. Reynolds, and Andrew Reamer. 2008. "Clusters and Competitiveness: A New Federal Role for Stimulating Regional Economies." *Brookings Institute*, April. http://www.brookings.edu/~/media/Files/rc/reports/2008/04_competitiveness_reamer/Clusters%20Re port.pdf.
- Miniwatts Marketing Group. 2010. "Internet World Stats: World Internet Usage and Population Statistics." Accessed June 30, 2011. http://www.internetworldstats.com/stats.htm.
- Missouri Virtual Instruction Program. 2008. *Annual Evaluation Report* 2007–2008. TA Consulting Evaluation Team. http://www.movip.org/about/evalreport2007-2008.pdf.
- Mohammed, Arshad. 2005. "FCC May Let Phone Companies Off DSL Hook." *Washington Post*, August 4. http://www.washingtonpost.com/wpdyn/content/article/2005/08/03/AR2005080302126.html.
- Mokhtarian, Patricia L. 2004. "Data and Measurement Issues in Transportation, with Telecommuting as a Case Study." University of California, Davis. http://www.its.ucdavis.edu/publications/2004/UCD-ITS-RR-04-29.pdf.
- Moller, Stephanie, and Stearns, Elizabeth. 2004. "Retention and School Dropout: Examining Connectivity Between Children and Schools." Paper presented at the annual meeting of the American Sociological Association, Hilton San Francisco & Renaissance Parc 55 Hotel, San Francisco, California, August 14. http://www.allacademic.com/meta/p_mla_apa_research_citation/1/0/8/7/6/p108764_index.html.
- Moody, NB, PL Smith, and LL Glenn. 1999. "Client Characteristics and Practice Patterns of Nurse Practitioners and Physicians." *Nurse Practitioners* 24: 94–96, 99–100, 102–103.

Morehart, Mitchell, and Peter Stenberg. 2006. "Determinants of Internet Use for Rural and Farm Economic Sectors." Economic Research Service, United States Department of Agriculture, Washington, DC. http://web.si.umich.edu/tprc/papers/2006/649/Stenberg%20TPRC2006%20v2.pdf

Morin, Rich, and Paul Taylor. 2009. "Luxury or Necessity? The Public Makes a U-Turn." *Pew Research Center*, April 23. http://pewresearch.org/pubs/1199/more-items-seen-as-luxury-not-necessity.



- Mosner, Ellen, Craig Spiezle, and Jim Emerman. 2003. "The Convergence of the Aging Workforce And Accessible Technology." Microsoft Corporation AgeLight Marketing Consultancy. Last modified July. http://www.hard2hire.com/library/boomers/The%20Convergence%20of%20the%20Aging%20Workforce%20and%20Accessible%20Technology.pdf.
- Moye, Johnny J. 2010. "Making Your Classes Come Alive." *Techniques: Connecting Education and Careers* 85: 8-9.
- Moye, Johnny J. 2009. "Technology Education Teacher Supply and Demand—A Critical Situation." *Technology Teacher* 69: 30-36.
- National Institute of Justice. 2004. "National Institute of Justice Annual Report." http://www.ncjrs.gov/pdffiles1/nij/209274.pdf.
- National Science Foundation. 2009. Academic Research and Development Expenditures, Fiscal Year 2007, Detailed Statistical Tables.
- National Skills Coalition. "Toward Ensuring America's Workers and Industries the Skills to Compete." Washington, DC: National Skills Coalition. http://www.nationalskillscoalition.org/assets/reports-/toward-ensuring-americas.pdf.
- National Telecommunications & Information Administration. 2002. "A Nation Online: How Americans Are Expanding Their Use of the Internet." United States Department of Commerce. Last modified February. http://www.ntia.doc.gov/legacy/ntiahome/dn/nationonline_020502.htm
- National Telecommunications and Information Administration. 2010. "Digital Nation: 21st Century America's Progress Toward Universal Broadband Internet Access." United States Department of Commerce. Last modified February 2010. http://www.ntia.doc.gov/reports/2010/NTIA_internet_use_report_Feb2010.pdf.
- National Telecommunications and Information Administration. 2011. "Digital Nation: Expanding Internet Usage." United States Department of Commerce. Article published February 2011. http://www.ntia.doc.gov/files/ntia/publications/ntia_internet_use_report_february_2011.pdf
- National Telecommunications and Information Administration. 2009. "Executive Branch Views on Public Safety, Homeland Security and Cybersecurity Elements of a National Broadband Plan." United States Department of Commerce.
- National Telecommunications and Information Administration. 1999. "Falling Through the Net: Defining the Digital Divide." United States Department of Commerce. http://www.ntia.doc.gov/ntiahome/fttn99/fttn.pdf.
- National Telecommunications and Information Administration. 1995. "Falling Through the Net: A Survey of the 'Have Nots' in Rural and Urban America." http://www.ntia.doc.gov/ntiahome/fallingthru.html.
- National Telecommunications and Information Administration. 2007. "Networked Nation: Broadband in America." United States Department of Commerce. http://www.ntia.doc.gov/reports/2008/NetworkedNationBroadbandinAmerica2007.pdf.
- National Telecommunications and Information Administration and United States Department of Agriculture's Rural Utilities Service. 2000. "Advanced Telecommunications in Rural America, The Challenge of Bringing Broadband Service to All Americans." Accessed November 15, 2010. http://www.ntia.doc.gov/reports/ruralbb42600.pdf.
- National Telecommunications and Information Administration. 2011. "Broadband Technology Opportunities Program Quarterly Status Report." Last modified December 2011. http://www.ntia.doc.gov/files/ntia/publications/btop-quarterly-congressional-report-dec-2011.pdf



National Telecommunication and Information Administration. 2010. Seventh Quarterly Status Report, November 2010. United States Department of Commerce.

National Emergency Number Association. 2009. "Safety Consortium to Promote Broadband-Enabled Emergency Response." *EMS Responder*, August 14. http://www.emsworld.com/web/online/Top-EMS-News/Safety-Consortium-to-Promote-Broadband-Enabled-Emergency-Response/1\$10191.

Newstrom, George C. 2002. "The Global Digital Economy and the Bold Dominion, Commonwealth of Virginia Strategic Plan for Technology." Secretary of Technology. Last modified September 26, 2002. http://www.vita.virginia.gov/uploadedFiles/Library/govTechPlanFull_020914.pdf.

Nguyen, Sang, and B. K. Atrostic. 2005. "Computer Investment, Computer Networks and Productivity." Working Papers from U.S. Census Bureau, Center for Economic Studies.

Nicole B. Ellison, Charles Steinfield, and Cliff Lamp. 2007. "The Benefits of Facebook 'Friends': Exploring the Relationship Between College Students' Use of Online Social Networks and Social Capital." *Journal of Computer-Mediated Communication* 12: 1143-1168. http://www.mvirtual.com.br/midiaedu/artigos_online/facebook.pdf.

Noguchi, Yuki. 2005. "Cable Firms Don't Have to Share Networks, Court Rules." *Washington Post*, June. http://www.washingtonpost.com/wp-dyn/content/article/2005/06/27/AR2005062700415.html.

Nokia Siemens Networks. 2009. *Connectivity Scorecard 2009.* http://www.connectivityscorecard.org/images/uploads/media/TheConnectivityReport2009.pdf

O'Brien, Kevin J. 2010. "Mobile Data, the Next Generation - High Speeds but at What Cost?" *New York Times*, February 15. http://www.nytimes.com/2010/02/16/technology/16lte.html.

O'Donnell, Susan, and Sonja Perley. 2006. "Broadband Video Communications Research in First Nation Communities." Paper presented at the Canadian Communication Association Annual Conference, York University, Toronto, Canada. http://www.susanodonnell.info/downloads/2006-Perley-CCA.pdf.

Office of Inspector General, U.S. Department Of Commerce. 2010. "National Telecommunications and Information Administration: Broadband Program Faces Uncertain Funding, and NTIA Needs to Strengthen Its Post-Award Operations." *Final Report No. OIG-11-005-A*, United States Department of Commerce, November 4. http://www.oig.doc.gov/oig/reports/2010/OIG-11-005-A.pdf.

Office of Personnel Management. 2009. "Federal Employment Statistics: Total Government Employment Since 1962." *Historical Federal Workforce Tables*. http://www.opm.gov/feddata/HistoricalTables/TotalGovernmentSince1962.asp.

Office of Small Business Development Centers Entrepreneurial Development. "Counseling & Assistance: Small Business Development Centers." United States Small Business Association. http://www.sba.gov/services/counseling/index.html.

Office of the Press Secretary. 2009. "Fact Sheet on American Graduation Initiative." White House Press Release. http://www.whitehouse.gov/the_press_office/Excerpts-of-the-Presidents-remarks-in-Warren-Michigan-and-fact-sheet-on-the-American-Graduation-Initiative/.

OMG Center for Collaborative Learning. 2009. "Learning More about What Works: Short-Term Client Outcomes of the Sustainable Broadband Adoption Pilot Program." Interim Evaluation Report, submitted to Digital Impact Group and the William Penn Foundation, July 28. http://www.digitalimpactgroup.org/documents/DIG%20Interim%20Report%207-28-09%20final.pdf.

Organization for Economic Co-operation and Development. 2010. OECD Broadband Portal. Directorate for Science, Technology and Industry. http://www.oecd.org/document/54/0,3343,en 2649 34225 38690102 1 1 1 1,00.html.



- Organization for Economic Cooperation and Development. 2001. "The Development of Broadband Access in OECD Countries." Last modified October 29. http://www.oecd.org/dataoecd/48/33/2475737.pdf.
- Organization for Economic Co-operation and Development. 2008. "Broadband and the Economy." Ministerial Background Report, presented at OECD Ministerial Meeting on the Future of the Internet Economy, Seoul, South Korea, June 17-18. http://www.oecd.org/dataoecd/62/7/40781696.pdf.
- Orlowski, Andrew. "FCC Opens Door to ISP Wipe-Out: Re Monopolizing the Phone Service." *The Andrew Orlowski Blog.* http://andreworlowski.com/2005/08/06/fcc-opens-door-to-isp-wipe-out/.
- Orr & Orr. 2010. "A ThomasNet Case Study." Accessed November 19, 2010. http://promoteyourbusiness.thomasnet.com/case_studies/orr-and-orr.html.
- Ortiz, Julio Angel and Andrea H. Tapia. 2006. "Municipal Responses to State-Level Broadband Internet Policy." Published in the Proceedings of the Telecommunication Policy Research Conference, September 30-October 2. http://web.si.umich.edu/tprc/papers/2006/554/TPRCfinal_pdf.
- Passey, Don, Colin Rogers, Joan Machell, and Gilly McHugh. 2004. "The Motivational Effect of ICT on Pupils." http://www.dcsf.gov.uk/research/data/uploadfiles/RR523new.pdf.
- Paczkowski, John. 2009. "The Median U.S. Broadband Speed? South Korea's Divided by Four." *All Things Digital*, August 26. http://digitaldaily.allthingsd.com/20090826/cwa/.
- Pamela E. Harrell, and Mary Harris. 2006. "Teacher Preparation Without Boundaries: A Two-Year Study of an Online Teacher Certification Program." *Journal of Technology & Teacher Education* 14. http://www.thefreelibrary.com/Teacher+preparation+without+boundaries:+a+two-year+study+of+an+online...-a0151387501.
- Panzar, C. and Steven S. Wildman. 1995. "Network Competition and the Provision of Universal Service." Industrial and Corporate Change 4: 711-719.
- Park, Do-Hyung, Jumin Lee, and Ingoo Han. 2007. "The Effect of On-Line Consumer Reviews on Consumer Purchasing Intention: The Moderating Role of Involvement." *Internal Journal of Electronic Commerce* 11, no. 4 (Summer).
- Partnership for 21st Century Skills. 2006. "Results That Matter, 21st Century Skills and High School Reform." Washington, DC: U.S. Department of Education. http://www.p21.org/documents/RTM2006.pdf.
- Paul, Ian. 2009. "South Korea to Get Super High-Speed Broadband." *Today* @ *PCWorld Blog*, February 3. http://www.pcworld.com/article/158799/south_korea_to_get_super_highspeed_broadband.html.
- Peha, Jon M., William H. Lehr, and Simon Wilkie. 2007. "The State of the Debate on Network Neutrality." Special Issue on Network Neutrality, co-edited with Jon Peha and Simon Wilkie, *International Journal of Communication* 1, August. http://ijoc.org/ojs/index.php/ijoc/article/view/192/100.
- Picciano, Anthony G., and Jeff Seaman. 2007. "K–12 Online Learning: A Survey of U.S. School District Administrators." *Sloan Consortium*. http://sloanconsortium.org/sites/default/files/K-12_Online_Learning_1.pdf.
- Picciano, Anthony G., and Jeff Seaman. 2007. "K–12 Online Learning: A Survey of U.S. School District Administrators." *Sloan Consortium*. http://sloanconsortium.org/sites/default/files/K-12_Online_Learning_1.pdf.

Picot, Arnold, and Christian Wernick. 2007. "The Role of Government in Broadband Access". *Telecommunications Policy* 31, no. 10-11 (November): 660-674.



- Pinkett, Randal. 2000. "Bridging the Digital Divide: Sociocultural Constructionism and an Asset-Based Approach to Community Technology and Community Building." Paper presented at the 81st Annual Meeting of the American Educational Research Association, New Orleans, Louisiana, April 24-28. http://alumni.media.mit.edu/~rpinkett/papers/aera2000.pdf.
- Pociask, Stephen B. 2004. "A Survey of Small Businesses' Telecommunications Use and Spending." U.S. Small Business Administration, Office of Advocacy.
- Pociask, Stephen B. 2002. "Building a Nationwide Broadband Network: Speeding Job Growth." TeleNomic Research, LLC. Last modified February 25. http://www.lus.org/uploads/BuildingaNationwideBroadbandNetwork.pdf.
- Pociask, Stephen. 2005. "Broadband Use by Rural Small Businesses." *Small Business Research Summary* 269. Last modified December. http://www.marketingtechie.com/articles/SmallBusiness%20-%20Dec20.pdf.
- Porter, Michael. 1985. *Competitive Advantage: Creating and Sustaining Superior Performance*. New York: Free Press.
- Putnam, Robert. 2000. *Bowling Alone: The Collapse and Revival of American Community.* New York: Simon and Schuster.
- PricewaterhouseCoopers. 2009. Champion for Digital Inclusion: The Economic Case for Digital Inclusion.
- Prieger, James E. and Wei-Min Hu. 2007. "The Broadband Digital Divide and the Nexus of Race, Competition, and Quality." Working paper series, August.
- Prieger, James. 2003. "The Supply Side of the Digital Divide: Is There Equal Availability in the Broadband Internet Access Market?" University of California-Davis Working Paper.
- Protheroe, Nancy. 2005. "Technology and Student Achievement." *Principal*, November-December. http://www.naesp.org/resources/2/Principal/2005/N-Dp46.pdf.
- Rainie, Lee, Kristen Purcell, and Aaron Smith. 2011. "The Social Side of the Internet." *Pew Internet and American Life Project*, January 18. http://pewinternet.org/Reports/2011/The-Social-Side-of-the-Internet.aspx.
- Rainey, Daniel V. 2000. "The Impact of Property Taxes and Growth Restrictions on Real Estate Prices." The Center for Applied Real Estate Education and Research, University of South Carolina.
- Rainie, Lee, Kristen Purcell, Tony Siesfeld, and Mayur Patel. 2011. "How the Public Perceives Community Information Systems." *Pew Internet and American Life Project*, March 11. http://pewinternet.org/Reports/2011/Community-Information-Systems.aspx.
- Reed, W. Robert, and Cynthia L. Rogers. 2003. "A Study of Quasi-experimental Control Group Methods for Estimating Policy Impacts." *Regional Science and Urban Economics* 33, no. 1 (January): 3-25.
- Reichert, Kent. 2001. "Use of Information Technology by Law Enforcement." University of Pennsylvania. Article published December. http://www.sas.upenn.edu/jerrylee/programs/fjc/paper_dec01.pdf.
- Reiser, Christa, Jennifer Wallin, and Kenneth Wilson. 2003. "Social Stratification and the Digital Divide." Social Science and Computer Review 21, no. 3 (June).
- Renko, Maija, and Paul Reynolds. 2006. "Profiling the Growth Oriented Nascent Entrepreneur in the US -Evidence from Representative Samples." Paper presented at Academy of Management Conference, Atlanta, August 11-16.



Rephann, Terance J. 2000. "The Economic Impacts of LULUs." *Environment and Planning C: Government and Policy* 18, no. 4: 393-407.

Riegner, Cate. 2007. "Word of Mouth on the Web: The Impact of Web 2.0 on Consumer Purchase Decisions." *Journal of Advertising Research* 47, no. 4 (December): 436-447. http://instruct.uwo.ca/mit/3771-001/Word_of_Mouth_On_the_Web__Cate_Riegner.pdf.

Rincon, Ana, Catherine Robinson, and Michela Vecchi. 2006. "The Productivity Impact of E-commerce in the UK, 2001: Evidence from Microdata." The National Institute of Economic and Social Research, United Kingdom.

Robert Geier, Phyllis Blumenfield, Ronald Marx, Joseph Krajcik, Barry Fishman, and Elliot Soloway. 2004. "Standardized Test Outcomes of Urban Students Participating in Standards and Project Based Science Curricula." Paper presented at the 6th International Conference on Learning Sciences, June 25. http://portal.acm.org/citation.cfm?id=1149126.1149150.

Robinson, John P, Paul DiMaggio, and Eszter Hargittai. 2003. "New Social Survey Perspectives on the Digital Divide." *IT & Society* 1, no. 5: 1-22. http://www.webuse.org/pdf/RobinsonEtAL-NewSocialSurveyPerspectives2003.pdf.

Rubinovitz, Robert N. 2003. "Market Power and Price Increases for Basic Cable Service Since Deregulation," *RAND Journal of Economics* 24, no. 1: 1-18.

Sandler, Todd. 2010. "Law Enforcement Information Sharing and the Implications for Local Government: A Technical Reference." Digital Communities Law Enforcement Information Technology Task Force. Last modified February. http://www.kms.ijis.org/db/attachments/public/3988/1/LE_Info_Sharing_Implications_for_Local_Govt. pdf.

Schrank, David, and Tim Lomax. 2009. *Urban Mobility Report*. Texas Transportation Institute, July. http://americandreamcoalition.org/highways/mobility_report_2009_wappx.pdf.

Sciadas, George. "The Digital Divide in Canada." Research paper, Science, Innovation and Electronic Information Division, Statistics Canada. http://dsppsd.pwgsc.gc.ca/Collection/Statcan/56F0009X/56F0009XIE2002001.pdf. Selouani, Sid-Ahmed and Habib Hamam. "Social Impact of Broadband Internet: A Case Study in the Shippagan Area, a Rural Zone in Atlantic Canada." Journal of Information, Information Technology, and Organizations, Volume 2, 2007. http://jiito.org/articles/JIITOv2p079-094Selouani74.pdf.

Servon, Lisa J. 2002. *Bridging The Digital Divide: Technology, Community, and Public Policy.* Massachusetts: Blackwell Publishers Ltd.

Shadish, William R., Thomas D. Cook and Donald T. Campbell. 2002. *Experimental and Quasi-Experimental Designs for Generalized Causal Inference*. Boston: Houghton.

Shah, Dhavan V, Jaeho Cho, William P. Eveland, Jr., Nojin Kwak. 2005. "Information and Expression in a Digital Age: Modeling Internet Effects on Civic Participation." *Communication Research* 32, no. 5 (October). http://www.journalism.wisc.edu/~dshah/CR2005.pdf.

Sherry, Lorraine. 1995. "Issues in Distance Learning." International Journal of Educational Telecommunications 4: 337-365.

Silvernail, David L. 2009. "Research and Evaluation of the Maine Learning Technology Initiative Laptop Program." Presentation, University of Southern Maine Center for Education Policy, Applied Research & Evaluation. http://maine.gov/mlti/resources/MLTI_March_09.pdf.

Small Business Administration Office of Advocacy. "Frequently Asked Questions." United States Small Business Administration. http://www.sba.gov/advo/stats/sbfaq.pdf.



- Smihly, Maria, and Hendrikus Storm. 2010. "ICT Usage in Enterprises 2009." *Data in Focus, Eurostat.* http://epp.eurostat.ec.europa.eu/cache/ITY_OFFPUB/KS-QA-10-001/EN/KS-QA-10-001-EN.PDF.
- Smith, Aaron, Kay Lehman Schlozman, Sidney Verba, and Henry Brady. 2009. "The Internet and Civic Engagement." *Pew Research Center*. http://www.pewinternet.org/~/media//Files/Reports/2009/The%20Internet%20and%20Civic%20Enga gement.pdf.
- Smith, Aaron. 2010. "Home Broadband 2010." *Pew Research Center: Pew Internet & American Life Project*. http://pewinternet.org/~/media//Files/Reports/2010/Home%20broadband%202010.pdf.
- Smith, Jeffery. 2009. "Mobilizing the 21st century workforce." *Civ Source*, June 23. http://civsourceonline.com/2009/06/23/mobilizing-the-21st-century-workforce/.
- Smith, Joel, and Candace Thile. 2009. "Continuous Improvement in Learning: Data from the Open Learning Initiative." Remarks at FCC Education Workshop on The Open Learning Initiative, August 20. http://www.broadband.gov/docs/ws_education/ws_education_smith.pdf.
- Solnik, Claude. 2010. "Telecommuting is on the Rise." Long Island Business News, February 10.
- State Educational Technology Directors Association. 2007. "Maximizing the Impact: The Pivotal Role of Technology in a 21st Century Education System." http://www.setda.org/c/document_library/get_file?folderId=191&name=P21Book_complete.pdf.
- Steven S. Wildman. 1997. "Interconnection Pricing, Standed Costs, and the Optimal Regulatory Contract." *Industrial and Corporate Change 6*, no. 4: 741-755.
- Stevenson, Betsey. 2006. "The Impact of the Internet on Worker Flows." University of Pennsylvania. Last modified December 2006. http://bpp.wharton.upenn.edu/betseys/papers/Stevenson_Internet.pdf.
- Strategic Networks Group. 2003. "Economic Impact Study of the South Dundas Township Fibre Network." Report prepared for the UK Department of Trade and Industry, Ontario, Canada.
- Strover, Sharon, and Joe Straubhaar. 2000. "E-Government Services and Computer and Internet Use in Texas." Telecommunications and Information Policy Institute, University of Texas at Austin. http://www.utexas.edu/research/tipi/reports2/dir_final3.pdf.
- Strover, Sharon, and Joe Straubhaar. 2001. "Evaluating the Community Technology Training Program at Reagan and Travis High Schools." Telecommunications and Information Policy Institute at University of Texas at Austin.
- Strover, Sharon. 1999. "Rural Internet Connectivity." Paper presented at the Telecommunications Research and Policy Conference, September. http://www.utexas.edu/research/tipi/reports2/TRPC.pdf.
- Strover, Sharon. 2003. "The Prospects for Broadband Deployment in Rural America." *Government Information Quarterly* 10, no. 2 (May): 95-106.
- Sullivan, Thomas M., Eric E. Menge, and Radwan Saade. 2003. "Ex Parte Presentation in a Non-Restricted Proceeding: Initial Regulatory Flexibility Analysis for Triennial Review of Unbundled Network Elements (CC Dkt. No. 01-338; CC Dkt. No. 96-98; CC Dkt No. 98-147)." *Letter to Federal Communications Commission Chairman Michael K. Powell*, U.S. Small Business Administration Office of Advocacy, February 5.
- Swire, Peter. 2009. "Smart Grid, Smart Broadband, Smart Infrastructure: Melding Federal Stimulus Programs to Ensure More Bang for the Buck." *Center for American Progress,* April. http://www.americanprogress.org/issues/2009/04/smart_infrastructure.html.



Takanori, Ida. 2007. *Broadband Economics: Lessons from Japan*. New York: Routledge Studies in Global Competition.

Talukdar, Debabrata. 2008. "Cost of Being Poor: Retail Price and Consumer Price Search Differs Across Inner-City and Suburban Neighborhoods." *Journal of Consumer Research,* October. http://develop.wikispaces.com/file/view/cost+of+being+poor.pdf

Terry, Ken. 2010. "Health Spending Hits 17.3 Percent of GDP In Largest Annual Jump." *BNET Today,* February 4. http://www.bnet.com/blog/healthcare-business/health-spending-hits-173-percent-of-gdpin-largest-annual-jump/1117.

Texas Center for Education Research. 2009. "Evaluation of the Texas Technology Immersion Pilot: Final Outcomes for a Four–Year Study (2004–05 to 2007–08)." http://www.etxtip.info/y4_etxtip_final.pdf.

Texas Transportation Institute. 2009. Annual Urban Mobility Report: Performance Measure Summary-Seattle, WA. TTI: Urban Mobility Information. http://mobility.tamu.edu/ums/congestion_data/tables/seattle.pdf.

The Alliance for Public Technology and the Communications Workers of America. 2009. "State Broadband Initiatives." Last modified June 2009. http://www.speedmatters.org/page/-/SPEEDMATTERS/Publications/CWA_APT_StateBroadbandInitiatives.pdf?nocdn=1.

The Berkman Center for Internet and Society. 2009. "Next Generation Connectivity: A Review of Broadband Internet Transitions and Policy from Around the World." Draft, The Berkman Center for Internet and Society, October.

The Broadband Technology Opportunities Program. 2010. "Expanding Broadband Access and Adoption in Communities Across America, Overview of Grant Awards." Last modified December 14. http://www.ntia.doc.gov/reports/2010/NTIA_Report_on_BTOP_12142010.pdf

The California Broadband Task Force. 2008. "The State of Connectivity: Building Innovation Through Broadband." Final Report of the California Broadband Task Force. Last modified January. http://www.cio.ca.gov/broadband/pdf/CBTF_FINAL_Report.pdf.

The Climate Group. 2008. "SMART 2020: Enabling the Low Carbon Economy in the Information Age." A report on behalf of Global e-Sustainability Initiative. http://www.smart2020.org/_assets/files/02_Smart2020Report.pdf.

The Library Research Center. 2002. "Public Library Use and Economic Hard times: Analysis of Recent Data." Report prepared for the American Library Association, University of Illinois at Urbana Champaign, April 18.

http://www.ala.org/ala/professionalresources/libfactsheets/alalibraryfactsheet06.cfm.

The Office of the National Coordinator for Health Information Technology: Health IT Adoption. http://healthit.hhs.gov/portal/server.pt?open=512&objID=1152&parentname=CommunityPage&parent id=28&mode=2&in_hi_userid=11113&cached=true.

The State of New Hampshire Department of Resources and Economic Development & The Telecommunications Advisory Board. 2008. "State of New Hampshire Broadband Action Plan." Last modified June 30, 2008. http://www.nheconomy.com/uploads/Final-Report-082808.pdf.

Tongia, R., and E.J. Wilson, III. 2009. "The Dark Side of Metcalfe's Law: Multiple and Growing Costs of Network Exclusion." Last modified September 22, 2009. http://www.cstep.in/docs/Network%20Exclusion%209-22-09.pdf.

Towns, Steve. 2009. "Federal CTO: Smart Grid, E-Health Records and Broadband Need CIO Input." *Government Technology*, October 29. http://www.govtech.com/pcio/Federal-CTO-Smart-Grid-E-Health-Records.html.



Tozzi, John. 2007. "New Orleans: A Startup Laboratory." *Business Week*, August 27. http://www.businessweek.com/print/smallbiz/content/aug2007/sb20070823_490984.htm.

- Tozzi, John. 2009. "The Rise of the 'Homepreneur." *Business Wee*k, October 23. http://www.businessweek.com/smallbiz/content/oct2009/sb20091023_263258.htm.
- Treasury Inspector General for Tax Administration. 2009. Reference Number: 2009-40-130, Department of the Treasury. http://www.treasury.gov/tigta/auditreports/2009reports/200940130fr.pdf
- Tsikalas, Kallen, Jihyun Lee, and Christine Newkirk. 2007. "Home Computing, School Engagement, and Academic Achievement of Low-Income Adolescents: Findings from Year One of a Three-Year Study of the CFY Intervention." Computers for Youth and Educational Testing Service, August 1. http://www.cfy.org/documents/CFY-ETS_Research_Study_FINAL.pdf.
- Turner, S. Derek. 2005. "Broadband Reality Check: The FCC ignores America's Digital Divide." *Free Press*. http://www.hearusnow.org/fileadmin/sitecontent/broadband_report_optimized.pdf.
- Turner, S. Derek. 2008. "Down Payment on Our Digital Future: Stimulus Policies for the 21st Century Economy." *Free Press*, December.
- United States Chamber of Commerce's Institute for a Competitive Workforce. 2008. "U.S. Chamber Highlights Need to Educate Next Generation of Workers." Press release, December 11. http://www.uschamber.com/press/releases/2008/december/us-chamber-highlights-need-educate-next-generation-workers.
- United States Department of Education. 2009. "The American Recovery and Reinvestment Act: Saving and Creating Jobs and Reforming Education." United States Department of Education. Last modified April 3. http://www2.ed.gov/policy/gen/leg/recovery/presentation/arra.pdf.
- United States Department of Health and Human Services. "Agencies that Provide Grants." http://grants.gov/aboutgrants/agencies_that_provide_grants.jsp.

United States Department of Housing and Urban Development. 2010. "Tour EZ/RC/ECs by State." Community Planning and Development. http://www.hud.gov/offices/cpd/economicdevelopment/programs/rc/tour/index.cfm

- United States Department of Justice. 2005. "The Herfindahl-Hirschman Index." Last modified August 15, 2005. http://www.usdoj.gov/atr/public/testimony/hhi.htm.
- United States Department of Labor. 2008. "PY 2008 WIA Annual Reports." Employment and Training Administration. http://www.doleta.gov/performance/results/AnnualReports/annual-report-08.cfm.
- United States Department of Labor. 2010. "Tools for America's Job Seekers Challenge Report. Last modified February 2, 2010. http://dolchallenge.ideascale.com/a/panel.do?id=5847.
- United States Senate Commission on Small Business and Entrepreneurship. "Minority Entrepreneurs." Democratic Page.
- Underwood, Jean, Allison Ault, Ohil Banyard, Karen Bird, Gayle Dillon, Mary Hayes, Ian Selwood, Bridget Somekh, and Peter Twining. 2005. *The Impact of Broadband in Schools*. Nottingham Trent University, British Educational Communications and Technology Agency.
- United States Census Bureau. *Current Population Survey*, 1996 2010. Accessed January 27, 2012. http://dataferrett.census.gov/.
- United States Census Bureau. *The American Community Survey*, 2005-2009. Accessed January 27, 2012. http://dataferrett.census.gov/.



United States Census Bureau. 2006. "Women-Owned Firms: 2002." United States Department of Commerce. Last modified August 2002. http://www2.census.gov/econ/sbo/02/sb0200cswmn.pdf

- United States Office of Personnel Management. 2009. "OPM Director Berry Drives Plan to Increase Telework among Federal Employees." Press release, April 29. http://www1.opm.gov/news/opm-director-berry-drives-plan-to-increase-telework-among-federal-employees, 1460.aspx.
- United States Office of Personnel Management. 2009. "Status of Telework in the Federal Government, Report to Congress." Report to Congress. Last modified August. http://www.telework.gov/Reports_and_Studies/Annual_Reports/2009teleworkreport.pdf.
- United States Small Business Administration. 2009. *The Small Business Economy: Report to the President*. Washington, DC: United States Government Printing Office.
- United States Small Business Administration. FY 2011 Congressional Budget Justification and FY2009 Annual Performance Report. http://www.sba.gov/idc/groups/public/documents/sba_homepage/fy_2011_cbj_09_apr.pdf.
- United States Department of Energy. 2009. Smart Grid System Report. Report published July. http://www.oe.energy.gov/DocumentsandMedia/SGSRMain_090707_lowres.pdf.
- Valentine, Gill, Jackie Marsh, Charles Pattie, and BMRB. 2005. "Children and Young People's Home Use of ICT for Educational. Purposes: The Impact on Attainment at Key Stages."

http://www.dcsf.gov.uk/research/data/uploadfiles/RR672.pdf.

- Van Houweling, Douglas, Guy T. Almes, and Ted Hanss. 1998. "First Internet2 Joint Applications." Presented at the Engineering QoS Workshop, May 21–22, 1998 Santa Clara, California. http://qos.internet2.edu/may98Workshop/9805-Proceedings.pdf.
- Van Nguyen, Sang, and B. K. Atrostic. 2006. "How Businesses Use Information Technology: Insights For Measuring Technology and Productivity." Working paper, United States Bureau of Census. http://www.ces.census.gov/index.php/ces/cespapers?down_key=101755.
- Varian, Hal, Robert E. Litan, Andrew Elder and Jay Shutter. 2002. The Net Impact Study: The Projected Economic Benefits of the Internet in the United States, United Kingdom, France and Germany, Version 2.0."
- Vellidis, G., V. Garrick, S. Pocknee, C. Perry, C. Kvien, and M. Tucker. 2007. "How Wireless Will Change Agriculture. "Papers presented at the 6th European Conference on Precision Agriculture, Skiathos, Greece, June 3-6.
- Michalchik, Vera, Sara Carriere, Deborah Kim Emery, Larry Gallagher, Ann House, Andres Molina, Lynne Peck Theis, and William R. Penuel. 2006. *One Economy Digital Communities: Transforming Lives for Low-Income Americans in San Jose and Miami*. Menlo Park, CA: Center for Technology in Learning.
- Wallsten, Scott. 2005. "Broadband Penetration: An Empirical Analysis of State and Federal Policies." Working Paper 05-12, AEI-Brookings Joint Center Publications, June.
- Wallsten, Scott. 2010. "The Future of Digital Communications Research and Policy." *Technology Policy Institute,* June. http://www.techpolicyinstitute.org/files/the%20future%20of%20digital%20communications%20resear

Wandner, Stephen A. 2008. "Employment Programs for Recipients of Unemployment Insurance." Monthly Labor Review, October. http://www.bls.gov/opub/mlr/2008/10/art2full.pdf.



ch%20and%20policy.pdf.

- Wang, Lili, Stuart Bretschneider, and Jon P. Gant. 2005. "Evaluating Web-based E-government Services with a Citizen-Centric Approach." Paper presented at the Proceedings of the 38th Annual Hawaii International Conference on System Sciences, Washington, DC, January.
- Waters, Jody, Sharon Strover, and Gary Chapman. 2002. "Community Networking in Texas: An Interim Evaluation Report for the Telecommunications Infrastructure Fund Board." Telecommunications and Information Policy Institute at University of Texas at Austin. http://www.utexas.edu/research/tipi/reports/2003/cn1evaluationinterimreportnovember2002.pdf.
- Watson, John, and Butch Gemin. 2008. "Promising Practices in Online Learning: Using Online Learning for At-Risk Students and Credit Recovery." *North American Council*, June. http://www.inacol.org/research/promisingpractices/NACOL_CreditRecovery_PromisingPractices.pdf.
- Waverman, Leonard, and Lars-Hendrik Röller. 1997. "Telecommunications Infrastructure & Economic Development: A Simultaneous Approach." Regulation Initiative Working Paper No. 11, August.
- WCPN 90.3 Ideastream. "Upside/Downside: Youngstown Business Incubator a Bright Spot in Region." Radio broadcast.
- Welch, Eric W., and Charles C. Hinnant. 2003. "Internet Use, Transparency, and Interactivity Effects on Trust in Government." Proceedings of the 36th Hawaii International Conference on System Sciences, IEEE Computer Society. http://codl2.computer.org/comp/proceedings/biosc/2003/1874/05/1874

http://csdl2.computer.org/comp/proceedings/hicss/2003/1874/05/187450144a.pdf.

- West, Darrel M. 2004 "E-Government and the Transformation of Service Delivery and Citizen Attitudes." *Public Administration* Review 64, no. 1 (January-February): 15-27. http://andromeda.rutgers.edu/~ncpp/cdgp/articles/West-2004.pdf.
- Westat. 1999. "Evaluation of the Telecommunications and Information Infrastructure Assistance Program for the 1994 and 1995 Grant Years." United States Department of Commerce, 1999. http://www.ntia.doc.gov/top/research/EvaluationReport/TIIAP_1994_1995.pdf.

Whisler, Arnim E. and Asheesh Saksen. 2003. "Igniting the Next Broadband Revolution." Accenture. Outlook Journal no. 1, January. http://www.accenture.com/SiteCollectionDocuments/PDF/outlook_broadband.pdf.

- Whitacre, Brian E. 2005. "Bridging the Rural Urban Digital Divide in Residential Internet Access." PhD diss., Virginia Polytechnic Institute. http://scholar.lib.vt.edu/theses/available/etd09262005123744/unrestricted/Whitacre.pdf.
- Whitt, Richard S. 2009. Evolving Broadband Policy: Taking Adaptive Stances To Foster Optimal Internet Platforms.
- Williams, Kate, and Abdul Alkalimat. 2004 "A Census of Public Computing in Toledo, Ohio." In *Shaping the Network Society: The New Role of Civic Society in Cyberspace*, edited by Doug Schuler and Peter Day. Cambridge, Massachusetts: MIT Press. http://www.communitytechnology.org/toledo/toledo.pdf.
- Williams, Kate and Abdul Alkalimat. 2008. "Cyberpower." In *Liberating Voices! A Pattern Language for Communication Revolution*, edited by Douglas Schuler. Massachusetts: MIT Press. http://www.publicsphereproject.org/patterns/pattern.pl/public?pattern_id=821.
- Williams, Kate, and Joan C. Durrance. 2009. "Community Informatics." *Encyclopedia of Library and Information Science*, edited by Marcia Bates and Mary Miles Maack. http://people.lis.illinois.edu/~katewill/williams-durrance-encyclo-community-informatics.pdf.



- Williams, Kate, and Joan C. Durrance. 2008. "Social Networks and Social Capital: Rethinking Theory in Community Informatics." *Journal of Community Informatics*. http://people.lis.illinois.edu/~katewill/rethinking-theory-williams-durrance.pdf
- Williams, Kate. 2007. "Community Informatics Memory as Archive: Assembling and Using the Records of the Technology Opportunities Program (USA), 1994-2005." In *Constructing and Sharing Memory: Community Informatics, Identity and Empowerment*, edited by Graeme Johanson and Larry Stillman. Cambridge: Cambridge Scholars Press,. http://people.lis.illinois.edu/~katewill/williams-memoryarchives-top-2007.pdf.
- Williams, Kate. 2003. "Literacy and computer literacy: Analyzing the NRC's 'Being Fluent with Information Technology'." *Journal of Literacy and Technology* 3.
- Williams, Kate. 2001. "What is the Digital Divide?" Draft working paper for the D3 Workshop, Ann Arbor, Michigan, August.
- Williams, Kate. 2000. "Libraries as ISPs." Ohio Libraries, Spring.
- Winhuasen, John Jr. 2008. "A Blueprint for Big Broadband." Paper presented at Presentation to Internet2 Spring Meeting, April 23. http://www.state.hi.us/auditor/agendas/windhausen_slides.pdf.
- Wireline Competition Bureau. 2010. "Internet Access Services: Status as of June 30, 2009." Industry Analysis and Technology Division, Wireline Competition Bureau. Last modified September. http://www.fcc.gov/Daily_Releases/Daily_Business/2010/db0902/DOC-301294A1.pdf.
- Wireline Competition Bureau. 2008. "Section 706 Report and Broadband Data Gathering Order." Presentation, Open Meeting, United States Federal Communication Commission, March 19. http://www.fcc.gov/WCB_031908_open_meeting_slides.pdf.
- Wireline Competition Bureau. 2010. "High-Speed Services for Internet Access: Status as of December 31, 2008." Report, Industry Analysis and Technology Division, United States Federal Communication Commission. http://hraunfoss.fcc.gov/edocs_public/attachmatch/DOC-296239A1.pdf.
- Woodside, Simon. 2010. "Mobile Phones Access Agricultural Market Price Information in Developing Nations." Simon & Woodside Blog, January 9. http://simonwoodside.com/weblog/2010/1/9/mobile_phones_access_agricultural_market/.
- WorldatWork. 2009. "Telework Trendlines 2009." A summary brief, February. http://www.worldatwork.org/waw/adimLink?id=31115.
- Wynne, Maria E. 2007. "Microsoft Director Sees Blueprint Emerging for Digital Inclusion in Cities." *Government Technology's Digital Communities*, September 10. http://www.digitalcommunities.com/articles/Microsoft-Director-Sees-Blueprint-Emerging-for.html.
- Wynne, Maria E., Karen Archer Perry, and Lane F. Cooper. 2009. "Innovating for Inclusion: A Digital Inclusion Guide for Those Leading the Way." Microsoft Digital Inclusion, Microsoft Corporation.
- Xin, Gong. 2010. "3G Applications to Tap SMEs Market Potential." *China Economic Net*, February 5. http://en.ce.cn/Insight/201001/29/t20100129_20888385.shtml.

Zero Divide. 2009. Community WiFi Study.

